Surveillance, early diagnosis and intervention for children with cerebral palsy in low resource settings: Lessons from Bangladesh and Vietnam

Dr Tasneem Karim

The University of Sydney Faculty of Medicine and Health Discipline of Child and Adolescent Health

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy 2021

THESIS STATEMENT OF ORIGINALITY

This is to certify that to the best of my knowledge the content of this thesis is my own work. This thesis has not been submitted for any degree or other purposes.

I certify that the intellectual content of this thesis is the product of my own work and that all the assistance received in preparing this thesis and sources have been acknowledged.

Signature

Date: 30 Jun 2021

ABSTRACT

Cerebral palsy (CP), the most common physical disability of childhood, is estimated to be more prevalent in low and middle-income countries (LMICs). The pathways and risk factors for CP in LMICs are believed to differ in low resource settings. This in addition to a range of key factors including delayed diagnosis; lack of evidence-based early intervention programs; poor or no access to rehabilitation and assistive devices; often result in greater severity of CP and associated impairments compared to high income countries.

This doctoral research aims to: (i) better understand the clinical characteristics of children with CP in Bangladesh with a specific focus on hip dysplasia and epilepsy, (ii) explore pathways to improve age at diagnosis of CP in Bangladesh, (iii) enhance interventions for children with CP and their primary caregivers through a quasi-experimental study assessing the outcome of an existing program in Bangladesh and (iv) describe CP in another low resource setting, Vietnam, using an alternative epidemiological tool, hospital-based surveillance.

This thesis describes two robust methodologies of data collection on children with CP in LMICs and affirms substantial burden of CP and associated impairments i.e., hip dysplasia, epilepsy, and malnutrition among children with CP in Bangladesh and Vietnam. It maps the path towards the establishment and use of national CP registers in low resource settings. Strategies for early detection and a promising early intervention program for children with CP and their families in Bangladesh, a typical LMIC, have also been described. The findings from this doctoral research have significant bearing in informing practice policy and research in low resource settings globally.

AUTHORSHIP ATTRIBUTION

This thesis includes six papers in total: four published papers, one under review and one ready for submission.

Chapter 1 of this thesis includes the published paper,

Karim T, Al Imam MH, Golland P, Khan AI, Hossain I, Smithers-Sheedy H, Badawi N, Muhit M, Khandaker G. Hip dysplasia among children with spastic cerebral palsy in rural Bangladesh. BMC Musculoskeletal Disorders. 2019 Dec 1;20(1):494.

Author role: I was the lead and corresponding author on this paper. I conducted the data analysis and data interpretation of the study. I wrote the draft and incorporated edits to the manuscript. I responded to comments from reviewers during the publication process with input from supervisors.

Chapter 2 of this thesis includes the includes the publication under review,

Karim T, Das MC, Muhit M, Badawi N, Khandaker G, Mohammad SM. Improving epilepsy control among children with Cerebral Palsy in rural Bangladesh. [Submitted to BMJ Open].

Author role: I was the lead and corresponding author on this study. I supported and contributed to the study design, development of the study materials and overall conduct of the study. I completed data analysis, and interpretation of the data. I wrote the draft and incorporated edits to the manuscript. I responded to comments from reviewers during the publication process with input from supervisors.

Chapter 3 of this thesis includes the published paper,

Karim T, Scherzer A, Muhit M, Badawi N, Khandaker G. Use of a Developmental Milestone Chart (DMC) in Rural Bangladesh to Educate Health Workers and Stimulate Referral for Early Diagnosis and Intervention. Journal of Tropical Pediatrics. 2019 Oct;65(5):505-9.

Author role: I was the lead author on this paper. I coordinated the study activities and conducted data analysis and interpreted the data. I wrote the first draft and incorporated edits to the manuscript. I responded to comments from reviewers during the publication process with input from supervisors.

Chapter 4 of this thesis includes the publication nearly ready for submission,

Karim T, Muhit M, Jahan I, Galea C, Smithers-Sheedy H, Badawi N, Khandaker G. Outcome of community-based parents led early intervention for children with cerebral palsy in rural Bangladesh - a quasi-experimental study.

This will be submitted for publication on a special issue of Brain Sciences journal titled, Early Detection and Early Intervention Strategies for Cerebral Palsy in Low and High Resource Settings.

Author role: I was the lead and corresponding author on this paper. I co-designed and cosupervised this study with co-authors. I curated, analyzed, and interpreted the data. I wrote the draft and incorporated edits to the manuscript. I responded to comments from reviewers during the publication process with input from supervisors.

Chapter 5 of this thesis includes the two published papers,

Karim T, Dosseter R, Giang NT, Dung TQ, Son TV, Hoa NX, Van Anh NT, Chau CM, Van Bang, N, Badawi N, Khandaker G, Elliott E. Data on cerebral palsy in Vietnam will inform clinical practice and policy in Low and Middle-Income Countries. Disability and Rehabilitation. Nov 2020.

Author role: I was the lead and corresponding author on this paper. I cleaned, pooled, analyzed and interpreted the data. I wrote the draft and incorporated edits to the manuscript. I responded to comments from reviewers during the publication process with input from supervisors.

Karim T, Jahan I, Dossetor R, Giang NT, Van Anh NT, Dung TQ, Chau CM, Van Bang N, Badawi N, Khandaker G, Elliott E. Nutritional Status of Children with Cerebral Palsy—Findings from Prospective Hospital-Based Surveillance in Vietnam Indicate a Need for Action. Nutrients. 2019 Sep;11(9):2132.

Author role: I was the lead and corresponding author on this paper. I cleaned, pooled, analyzed and interpreted the data. I wrote the draft and incorporated edits to the manuscript. I responded to comments from reviewers during the publication process with input from supervisors.

Appendix A includes the publication,

Khandaker G, Muhit M, Karim T, Smithers-Sheedy H, Novak I, Jones C, Badawi N. Epidemiology of cerebral palsy in Bangladesh: a population-based surveillance study. Developmental Medicine & Child Neurology. 2018 Nov 5.

Author role: I supported ethics applications and reporting on this paper. I was responsible for recruitment and data collection in Bangladesh. I assisted in data entry and analysis. I drafted the initial and revised manuscript along with GK with input from all the co-authors.

For all publications, supervisors and co-authors were involved in assisting with data interpretation and critically revising the manuscripts for important intellectual content.

In addition to the statements above, in the case where I am not the corresponding author of a published item, permission to include the published material has been granted by the corresponding author.

Student name: Dr Tasneem Karim

Signature

Date: 30 Jun 2021

As supervisors for the candidature upon which this thesis is based, we can confirm that the authorship attribution statements above are correct.

Supervisor Name: Professor Gulam Khandaker Signature

Date: 30 Jun 2021

Auxiliary Supervisor Name: Professor Nadia Badawi Signature

Date: 30 Jun 2021

DEDICATION

Dedicated to *Maa*, my guiding light.

Forever engraved in fond memories, a true inspiration, and a quiet rebel.

আমি চির-বিদ্রোহী বীর – বিশ্ব ছাড়ায়ে উঠিয়াছি একা চির-উন্নত শির!

ACKNOWLEDGEMENT

The love, encouragement and support from my family, friends, mentors, supervisors, team members and supporters of this work made my doctoral journey possible. There are many I would like to acknowledge for their support during my candidature.

Firstly, my supervisors Gulam and Nadia, for their foresight, wisdom, and encouragement. I am grateful for the opportunities they supported me with, the memories and the lessons I learned from them, they will continue to inspire me beyond my doctoral work. I am privileged to have my research garnering with expert opinion from them across a range of disciplines - neonatology, pediatrics, epidemiology, public health - instrumental to cerebral palsy research. I would like to sincerely thank Professor Mohammad Muhit for his guidance and expansive work in the sphere of international childhood disability that has been foundational to my doctoral research. Special thanks to Professor Elizabeth Elliott, Professor Alfred Scherzer, Dr Shekeeb S Mohammad, Dr Sarah McIntyre, Dr Hayley Smithers-Sheedy and Dr Catherine Morgan for their expertise, guidance, and support with vital parts of this doctoral research.

Thanks to my fellow team members and colleagues at CSF Global, Cerebral Palsy Alliance and The University of Sydney. A heartfelt thanks to Israt, a dear friend and colleague. Our journey has been wonderfully bizarre, and I look forward to what the future holds for us. Thanks to Petra, Rosalie and Shona for their guidance and the necessary sanity checks throughout my candidature. Special thanks to Anna and her family for welcoming me into their lives.

Thanks to Chris, *Baba*, Ivan, Safwaan, Faiza and my dear friends for their love and unwavering faith in me.

A special thanks to Rob White and the Research Foundation of Cerebral Palsy Alliance for making this doctoral research possible. I would like to specially acknowledge Nadia and Alf Taylor from tha solution for supporting the work in Bangladesh. Thanks to the Rotary Club of Turramurra, Wheelchairs for Kids and Don Kidson for supporting the Bangladesh Cerebral Palsy Register team in enhancing wheelchair accessibility for children with cerebral palsy in Shahjadpur subdistrict of Bangladesh. Thanks to Dave and Kerry Rickards for generously supporting the children with cerebral palsy and their families in Bangladesh during the pandemic. Finally, my highest gratitude for the children and their families for generously sharing their stories with us to support work that will benefit future generations in low resource settings across the world. I am pleased to share some of their smiles in my thesis through photos I had the joy of capturing during my doctoral journey.¹



All human beings are born free and equal in dignity and rights. Article 1, United Nations Universal Declaration of Human Rights

¹ Informed written consent has been obtained for all the photos included in this thesis

TABLE OF CONTENTS

Thesis Statement of Originality	2
Abstract	3
Authorship Attribution	4
Dedication	8
Acknowledgement	9
Table of Contents	11
Preface	14
Publications included in this thesis	14
First author oral presentations related to this thesis	14
First author poster presentations related to this thesis	15
Awards and funding related to this thesis	15
Relevant training completed during candidature	16
Relevant part time employment during candidature	16
Co-authored publications during candidature	16
Co-authored presentations during candidature	17
Community and professional engagement during candidature	18
List of Figures	20
List of Abbreviations	20
Key Words	20
Key Words INTRODUCTION AND RESEARCH FOCUS	20 21
Key Words INTRODUCTION AND RESEARCH FOCUS Overview	20 21 21
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background	20 21 21 21
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy	20 21 21 21 22
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition	20 21 21 21 22 22
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy	20 21 21 21 22 22 22
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy	20 21 21 21 22 22 22 23
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy	20 21 21 21 22 22 23 23
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy Cerebral Palsy Registers	20 21 21 21 22 22 23 23 23
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy Cerebral Palsy Registers Research Context: Bangladesh	20 21 21 21 22 22 23 23 23 23
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy Cerebral Palsy Registers Research Context: Bangladesh Bangladesh Cerebral Palsy Register	20 21 21 21 22 22 22 23 23 23 26 27
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy Cerebral Palsy Registers Research Context: Bangladesh Bangladesh Cerebral Palsy Register Thesis Outline and Aims	20 21 21 21 22 22 22 23 23 23 23 23 23 23 23
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy Cerebral Palsy Registers Research Context: Bangladesh Bangladesh Cerebral Palsy Register Thesis Outline and Aims Thesis Concept Map	20 21 21 21 22 22 22 23
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy Cerebral Palsy Registers Research Context: Bangladesh Bangladesh Cerebral Palsy Register Thesis Outline and Aims Thesis Concept Map	20 21 21 21 22 22 22 23 23 23 23 23 23 30 33 34
Key Words INTRODUCTION AND RESEARCH FOCUS Overview Background Cerebral Palsy Definition Burden of cerebral palsy Classifications of cerebral palsy Risk factors for cerebral palsy Cerebral Palsy Registers Research Context: Bangladesh Bangladesh Cerebral Palsy Register Thesis Outline and Aims Thesis Concept Map CHAPTER 1. Hip dysplasia among children with spastic cerebral palsy in rural Bangladesh	20 21 21 21 22 22 22 23 23 23 23 23 23 30 33 34 35

1.3 Chapter synopsis	43
CHAPTER 2. Improving epilepsy control among children with cerebral palsy in rural Ban	gladesh .44
2.1 Introduction	45
2.2 Paper #2. Improving epilepsy control among children with Cerebral Palsy in rural Ba	ngladesh45
2.3 Chapter synopsis	78
Chapter 3. Early Detection of children with cerebral palsy in Bangladesh	79
3.1 Introduction	79
3.2 Paper #3. Use of developmental milestone chart (DMC) in rural Bangladesh to educ workers and stimulate referral for early diagnosis and intervention	ate health 80
3.3 Chapter synopsis	86
Chapter 4. Intervention for children with cerebral palsy in Bangladesh	
4.1 Introduction	88
4.2 Paper #4. Outcome of community-based parents led early intervention for	88
children with cerebral palsy in rural Bangladesh: A quasi-experimental study	88
4.3 Chapter synopsis	108
Chapter 5. Hospital-based surveillance of children with cerebral palsy in Vietnam	109
5.1 Introduction	110
5.2 Paper #5 Data on Cerebral Palsy in Vietnam: Findings from Prospective Hospital-Ba Surveillance in Vietnam	sed 110
5.3 Paper #6. Nutritional Status of Children with Cerebral Palsy: Findings from Prospect Hospital-Based Surveillance in Vietnam	tive 110
5.4 Chapter synopsis	134
DISCUSSION AND RECOMMENDATIONS FOR PRACTICE, POLICY AND RESEARCH	135
Thesis summary: Bridging the gaps	135
Cerebral palsy registers in low resource settings	136
Bangladesh Cererbal Palsy Register: Onwards and upwards	136
Hospital-based surveillance in Vietnam: Next steps	137
Associated impairments	138
Hip dysplasia	138
Epilepsy	
Malnutrition	139
Early detection	140
Early intervention	141
Limitations and strengths	141
Overall recommendations for practice, policy and future research	143
Conclusion	144

References	
Appendices	151
Appendix A. Paper: Epidemiology of Cerebral Palsy in Bangladesh: a population-based su study	rveillance
Appendix B: Bengali translation of the epilepsy management guideline	

PREFACE

PUBLICATIONS INCLUDED IN THIS THESIS

- Khandaker G, Muhit M, Karim T, Smithers-Sheedy H, Novak I, Jones C, Badawi N. Epidemiology of cerebral palsy in Bangladesh: a population-based surveillance study. Developmental Medicine and Child Neurology. 2018 Nov 5. [Appendix A]
- 2. Karim T, Scherzer A, Muhit M, Badawi N, Khandaker G. Use of a Developmental Milestone Chart (DMC) in Rural Bangladesh to Educate Health Workers and Stimulate Referral for Early Diagnosis and Intervention. Journal of Tropical Pediatrics. 2018 Dec 19.
- Karim T, Jahan I, Dossetor R, Giang NT, Van Anh NT, Dung TQ, Chau CM, Van Bang N, Badawi N, Khandaker G, Elliott E. Nutritional Status of Children with Cerebral Palsy— Findings from Prospective Hospital-Based Surveillance in Vietnam Indicate a Need for Action. Nutrients. 2019 Sep;11(9):2132.
- Karim T, Al Imam MH, Golland P, Khan AI, Hossain I, Smithers-Sheedy H, Badawi N, Muhit M, Khandaker G. Hip dysplasia among children with spastic cerebral palsy in rural Bangladesh. BMC Musculoskelet Disord. 2019 Oct 27;20(1):494.
- Karim T, Dosseter R, Giang NT, Dung TQ, Son TV, Hoa NX, Van Anh NT, Chau CM, Van Bang, N, Badawi N, Khandaker G, Elliott E. Data on cerebral palsy in Vietnam will inform clinical practice and policy in Low and Middle-Income Countries. Disability and Rehabilitation. Nov 2020.
- 6. Karim T, Das MC, Muhit M, Badawi N, Khandaker G, Mohammad SM. Improving epilepsy control among children with Cerebral Palsy in rural Bangladesh. [under review]
- 7. Karim T, Muhit M, Jahan I, Galea C, Smithers-Sheedy H, Badawi N, Khandaker G. Outcome of community-based parents led early intervention for children with cerebral palsy in rural Bangladesh a quasi-experimental study. [under review]

FIRST AUTHOR ORAL PRESENTATIONS RELATED TO THIS THESIS

- Challenges in Setting up a Data Registry in Low and Middle-income Countries: Bangladesh Experience. 1st Congress of Pediatric Alliance and Cerebral Palsy Alliance, 14 - 16 Sep, Hunan, China
- Karim T, Dossetor R, Giang NH, Anh NV, Dung TQ, Chau CM, Badawi N, Bang NV. Khandaker G, Elliott E. Novel data on cerebral palsy in Vietnam will inform clinical practice and policy. Sydney Vietnam Symposium, 21 Sep, The University of Sydney Business School, Sydney, Australia
- 3. Karim T, Dossetor R, Giang NH, Anh NV, Dung TQ, Chau CM, Badawi N, Bang NV. Khandaker G, Elliott E. Novel data on cerebral palsy in Vietnam will inform clinical practice and policy. Sydney Vietnam Research Innovation Showcase: 13 Dec 2018, Hanoi, Vietnam
- 4. Karim T, Das MC, Muhit M, Badawi N, Khandaker G, Mohammad SM. Improving epilepsy control among children with Cerebral Palsy in rural Bangladesh. 31st Annual meeting of the European Academy of Childhood Disability (EACD), 23 25 May 2019, Paris, France
- 5. What Works for Early Intervention Implementation in Diverse Low Resource Settings? Experiences from Bangladesh, India And Uganda. International Society on Early Intervention Conference (ISEI), 25 28 Jun 2019, Sydney, Australia [Workshop]

- Karim T, Jahan I, Imam MHA, Das MC, Muhit M, Badawi N, Khandaker G. Cerebral palsy and health inequity in Bangladesh: Findings from the Bangladesh Cerebral Palsy Register. 74th Annual Meeting of the American Academy for Cerebral Palsy and Developmental Medicine (AACPDM), 23-26 September 2020, Virtual Meeting
- Karim T, Jahan I, Imam MHA, Das MC, Muhit M, Badawi N, Khandaker G. Cerebral palsy and health inequity in Bangladesh: Findings from the Bangladesh Cerebral Palsy Register.
 32nd Annual meeting of the European Academy of Childhood Disability (EACD), May – June 2021, Virtual Meeting

FIRST AUTHOR POSTER PRESENTATIONS RELATED TO THIS THESIS

- Dossetor R, Karim T, Giang NH, Anh NV, Dung TQ, Chau CM, Badawi N, Bang NV. Khandaker G, Elliott E. Novel data on cerebral palsy in Vietnam will inform clinical practice and policy. 72nd Annual Meeting of the American Academy for Cerebral Palsy and Developmental Medicine (AACPDM), 9-13 October 2018, Cincinnati, Ohio, USA
- Karim T, Muhit M, Jahan I, Galea C, Smithers-Sheedy H, Badawi N, Khandaker G. Outcome of community-based parents led early intervention for children with cerebral palsy in rural Bangladesh - a quasi-experimental study. 72nd Annual Meeting of the American Academy for Cerebral Palsy and Developmental Medicine (AACPDM), 9-13 October 2018, Cincinnati, Ohio, USA
- Karim T, Jahan I, Dossetor, Giang N, Anh NV, Dung T, Chau C, Thuong NV, Badawi N, Bang NV, Khandaker G, Elliott E. Nutritional status of children with cerebral palsy: findings from hospital-based surveillance in Vietnam. 73rd Annual Meeting of AACPDM, 18-21 September 2019, Anaheim, California, USA
- 4. Karim T, Das MC, Muhit M, Badawi N, Khandaker G, Mohammad S. Improving epilepsy control among children with Cerebral Palsy in rural Bangladesh. 73rd Annual Meeting of the AACPDM, 18-21 September 2019, Anaheim, California, USA

AWARDS AND FUNDING RELATED TO THIS THESIS

- 1. First place at the <u>2016 Life Shots</u> photo contest at the AACPDM 70th Annual Meeting in Hollywood, Florida.
- 2. American Academy for Cerebral Palsy and Developmental Medicine (AACPDM) 2017, International Travel Scholarship to attend and present at the 71st Annual Meeting of AACPDM, Montreal, Canada.
- 3. Cerebral Palsy Alliance Research Foundation Career Development Grant 2017 for project titled 'Cerebral palsy (CP) in Bangladesh: towards developing a national CP register, surveillance, early diagnosis and interventions for children with CP in low resource setting'.
- 4. Travel scholarship from the Sydney Vietnam Academic Leadership Group, University of Sydney to present at the Sydney Vietnam Research Showcase in Dec 2018, Hanoi, Vietnam.
- 5. Bursary to attend and present at the 31st Annual meeting of the European Academy of Childhood Disability (EACD) held on 23 25 May 2019 in Paris, France.
- 6. Cerebral Palsy Alliance Research Foundation PhD Research Grant 2019 for project titled 'Cerebral palsy (CP) in Bangladesh: towards developing a national CP register,

surveillance, early diagnosis and interventions for children with CP in low resource setting'.

- 7. Scholarship to present at the Australasian Academy of Cerebral Palsy and Developmental Medicine 10th Biennial Conference 2020 held on 11 14 March 2020 in Perth, Australia.
- 8. Recipient of scholarship from the University of Sydney Postgraduate Research Support Scheme 2020 awarded in July 2020.
- 9. Cerebral Palsy Alliance Research Foundation PhD Research Grant 2020 for project titled 'Cerebral palsy (CP) in Bangladesh: towards developing a national CP register, surveillance, early diagnosis and interventions for children with CP in low resource setting'.
- 10. Cerebral Palsy Alliance Research Foundation Project Research Grant 2021 for project titled Evidence based approach to surveillance, early diagnosis, and intervention of children with cerebral palsy in Vietnam (PRG08419, AU\$200,230.00 over 3 years) Principal Investigator.

RELEVANT TRAINING COMPLETED DURING CANDIDATURE

- 1. Prechtl's Method of the Qualitative Assessment of General Movements Advanced Course, 1 - 4 November 2018, Sydney, Australia.
- 2. Hammersmith Infant Neurological Examination (HINE) training, 4 November 2018, Sydney, Australia.

RELEVANT PART TIME EMPLOYMENT DURING CANDIDATURE

- Research Assistant [Jul 2018 July 2019] The University of Sydney Westmead Clinical School, Faculty of Medicine and Health Sydney, Australia.
- Research Officer [Jul 2019 Sep 2020] Cerebral Palsy Alliance Research Institute, The University of Sydney, Sydney, Australia.

CO-AUTHORED PUBLICATIONS DURING CANDIDATURE

- Chan J, Wu Y, Wood J, Muhit M, Mahmood M, Karim T, Moushumi F, Jones C, Snelling T, Khandaker G. Burden of Congenital Rubella Syndrome (CRS) in Bangladesh: Systematic review of existing literature and transmission modelling of sero– prevalence studies. Infectious Disorders - Drug Targets. Oct 2018.
- 2. Khan A, Ashher F, Karim T, Nawar A, Jahan I, Muhit M, Dey A, Beard F, Khandaker G, immunization of mothers of children with cerebral palsy in rural Bangladesh. Infectious Disorders Drug Targets. Oct 2018.
- Jahan I, Karim T, Das MC, Muhit M, Mcintyre S, Smithers-Sheedy H, Badawi N, Khandaker G. Mortality in children with cerebral palsy in rural Bangladesh: a population-based surveillance study. Developmental Medicine and Child Neurology. 2019 May 12.
- 4. Power R, Muhit M, Heanoy E, Karim T, Badawi N, Akhter R, Khandaker G. Healthrelated quality of life and mental health of adolescents with cerebral palsy in rural Bangladesh. PloS one. 2019 Jun 11;14(6):e0217675.

- 5. Power R, Galea C, Muhit M, Heanoy E, Karim T, Badawi N, Khandaker G. Determinants of health-related quality of life among adolescents with cerebral palsy in Bangladesh. Research Square; 2019 Sep. DOI: 10.21203/rs.2.13888/v1.
- 6. Power R, Galea C, Muhit M, Heanoy E, Karim T, Badawi N, Khandaker G. What predicts the health-related quality of life of adolescents with cerebral palsy in Bangladesh?. Research Square; 2019 Nov. DOI: 10.21203/rs.2.13888/v2.
- Power R, Akhter R, Muhit M, Wadud S, Heanoy E, Karim T, Badawi N, Khandaker G. A quality of life questionnaire for adolescents with cerebral palsy: psychometric properties of the Bengali CPQoL-teens. Health and quality of life outcomes. 2019 Dec 1;17(1):135.
- 8. Power R, Akhter R, Muhit M, Wadud S, Heanoy E, Karim T, Badawi N, Khandaker G. Cross-cultural validation of the Bengali version KIDSCREEN-27 quality of life questionnaire. BMC pediatrics. 2019 Dec;19(1):19.
- 9. Jahan I, Muhit M, Hardianto D, Karim T, Al Imam MH, Das MC, Smithers-Sheedy H, Badawi N, Khandaker G. Nutritional status of children with cerebral palsy in remote Sumba Island of Indonesia: a community-based key informants study. Disability and Rehabilitation. 2019 Oct 25:1-0.
- 10. Jahan I, Karim T, Al Imam MH, Das MC, Ali KM, Muhit M, Khandaker G. Childhood Disability and Nutrition: Findings from a Population-Based Case Control Study in Rural Bangladesh. Nutrients. 2019 Nov;11(11):2728.
- 11. Power R, Muhit M, Heanoy E, Karim T, Galea C, Badawi N, Khandaker G. Depression, anxiety and stress among caregivers of adolescents with cerebral palsy in rural Bangladesh. Disability and Rehabilitation. 2019 Dec 3:1-8.
- 12. Power R, Galea C, Muhit M, Heanoy E, Karim T, Badawi N, Khandaker G. What predicts the proxy-reported health-related quality of life of adolescents with cerebral palsy in Bangladesh? BMC public health. 2020 Dec 1;20(1):18.
- Jahan I, Al Imam MH, Karim T, Muhit M, Hardianto D, Das MC, Smithers-Sheedy H, Badawi N, Khandaker G. Epidemiology of cerebral palsy in Sumba Island, Indonesia. Developmental Medicine and Child Neurology. 2020 Jul 20.
- 14. Power R, Wiley K, Muhit M, Heanoy E, Karim T, Badawi N, Khandaker G. 'Flower of the body': menstrual experiences and needs of young adolescent women with cerebral palsy in Bangladesh, and their mothers providing menstrual support. BMC Women's Health. 2020 Dec;20(1):1-9.

CO-AUTHORED PRESENTATIONS DURING CANDIDATURE

- Jahan I, Al Iman MH, Karim T, Muhit M, Hardianto D, Das MC, Smithers-Sheedy H, Badawi N, Khandaker G. Epidemiology of cerebral palsy in remote Sumba Island of Indonesia: Findings from a community-based Key Informant Method (KIM) survey. 31st Annual meeting of the European Academy of Childhood Disability (EACD), 23 – 25 May 2019, Paris, France [Poster].
- Jahan I, Karim T, Das MC, Muhit M, Smithers-Sheedy H, Badawi N, Khandaker G. Mortality among Children with Cerebral Palsy (CP) in rural Bangladesh: Results from the Bangladesh CP Register (BCPR). 31st Annual meeting of the European Academy of Childhood Disability (EACD), 23 – 25 May 2019, Paris, France [Poster].

- Dossetor R, Karim T, Giang NH, Anh NV, Dung TQ, Chau CM, Badawi N, Bang NV. Khandaker G, Elliott E. Novel data on cerebral palsy in Vietnam will inform clinical practice and policy. 31st Annual meeting of the European Academy of Childhood Disability (EACD), 23 – 25 May 2019, Paris, France [Poster].
- Karim T, Muhit M, Jahan I, Galea C, Smithers-Sheedy H, Badawi N, Khandaker G. Outcome of community-based parents led early intervention for children with cerebral palsy in rural Bangladesh - a quasi-experimental study. 31st Annual meeting of the European Academy of Childhood Disability (EACD), 23 – 25 May 2019, Paris, France [Poster].
- 5. Muhit M, Karim T, Jahan I, Imam MHA, Das MC, Khandaker G. Epidemiology of eye diseases among children with disability in rural Bangladesh: findings from the Shahjadpur Children's Cohort. 73rd Annual Meeting of AACPDM, 18-21 September 2019, Anaheim, California, USA [Poster].
- Jahan I, Muhit M, Hardianto D, Karim T, Imam MHA, Das MC, Smithers-Sheedy H, Badawi N, Khandaker G. Nutritional status of children with cerebral palsy in remote Sumba Island of Indonesia: a community-based key informants study. 73rd AACPDM 18-21 September 2019, Anaheim, California, USA [Poster].
- Muhit M, Jahan I, Imam MHA, Karim T, Ghose R, Das MC, Chhetri AB, Khandaker G. Epidemiology of cerebral palsy in Gorkha district of Nepal: findings from a communitybased Key Informant Survey. 73rd Annual Meeting of the AACPDM, 18-21 September 2019, Anaheim, California, USA [Poster].
- Jahan I, Al Iman MH, Karim T, Muhit M, Hardianto D, Das MC, Smithers-Sheedy H, Badawi N, Khandaker G. Community based Key Informants Method (KIM) survey of children with cerebral palsy in rural Sumba Island of Indonesia. 72nd Annual Meeting of the American Academy for Cerebral Palsy and Developmental Medicine (AACPDM), 9-13 October 2018, Cincinnati, Ohio, USA [Oral Presentation].
- Jahan I, Karim T, Das MC, Muhit M, Smithers-Sheedy H, Badawi N, Khandaker G. Mortality among Children with Cerebral Palsy (CP) in rural Bangladesh: Results from the Bangladesh CP Register (BCPR). 72nd Annual Meeting of the American Academy for Cerebral Palsy and Developmental Medicine (AACPDM), 9-13 October 2018, Cincinnati, Ohio, USA [Oral Presentation].
- 10. Jahan I, Al Iman MH, Karim T, Muhit M, Hardianto D, Das MC, Smithers-Sheedy H, Badawi N, Khandaker G. Nutritional Status of Children with Cerebral Palsy (CP) in Remote Sumba Island of Indonesia. 31st Annual meeting of the European Academy of Childhood Disability (EACD), 23 – 25 May 2019, Paris, France [Oral Presentation].

COMMUNITY AND PROFESSIONAL ENGAGEMENT DURING CANDIDATURE

- Senior Research Program Manager [May 2017 present] and Research Program Manager [Apr 2016 - Apr 2017], CSF Global | Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka, Bangladesh (https://www.disabilityasia.org/)
- Research Affiliate [Jul 2019 present] The University of Sydney Westmead Clinical School, Faculty of Medicine and Health Sydney, Australia.

- Student Theme Leader, Disability and Chronic Disease theme of the Sydney Global Child Health Network, a joint initiative by the Discipline of Child and Adolescent Health and the Sydney School of Public Health at the University of Sydney.
- Member, Cerebral Palsy Alliance Research Committee [Jun Sep 2020].
- Member, Scientific Review Panel of the Australian Paediatric Surveillance Unit.
- Member, Australasian Academy of Cerebral Palsy and Developmental Medicine.
- Member, American Academy for Cerebral Palsy and Developmental Medicine.
- Member, European Academy of Childhood Disability.
- Member, International Society on Early Intervention.
- Member, Ethics Committee of the Asian Institute of Disability and Development.
- Academic Member, Sydney Vietnam Initiative, Sydney Southeast Asia Centre, University of Sydney.

LIST OF FIGURES

Figure 1. Map of Bangladesh	26
Figure 2. Thesis Concept Map	33

LIST OF ABBREVIATIONS

ACPR	Australian cerebral palsy register				
BCPR	Bangladesh cerebral palsy register				
СР	Cerebral palsy				
GLM-CPR	Global low-and middle-income country cerebral palsy registe				
GMA	General movement assessment				
GMFCS	Gross motor functional classification system				
HIC	High income country				
HINE	Hammersmith infant neurological examination				
LMIC	Low and middle-income country				
PAEDS	Paediatric Active Enhanced Disease Surveillance				
SCPE	Surveillance of Cerebral Palsy in Europe				

KEY WORDS

Cerebral palsy

Population-based surveillance

Hospital-based surveillance

Bangladesh

Vietnam

Low resource settings

Low and middle-income country

INTRODUCTION AND RESEARCH FOCUS

OVERVIEW

This thesis constitutes a series of papers with supplementary text to describe the body of work completed as part of the doctoral program. The epidemiology, clinical characteristics, and associated impairments of cerebral palsy (CP) are described, with concurrent exploration of measures for early identification and intervention for children with CP in low resource settings. This chapter includes the background to this doctoral research, the outline of the chapters and the concept map for the thesis.

BACKGROUND

Following the completion of my Bachelor of Medicine and Bachelor of Surgery and Master of Public Health in Bangladesh, I was eager to contribute to the preventive sphere. Contributing to the wellbeing of marginalized populations, a sentiment I fostered from my parents' example in their work in the medical field, has been central to my career aspirations. I was drawn to opportunities that would enable me to work with children, particularly in roles and organizations supporting children with disability and their families. During my search for employment in the new yet familiar world of public health, I was intrigued by an available position at CSF Global (www.csf-global.org).

CSF Global, an independent not-for-profit organisation, is committed to the establishment of a rights-based inclusive society for children with disability in developing countries. The core activities of the organisation constitute multidisciplinary research and generation of highquality scientific evidence to inform practice, policy, and service delivery. CSF Global was recruiting for the implementation of the Health-Related Quality of Life of Children with Cerebral Palsy in Rural Bangladesh study (Bangladesh CPQoL study). Thus, began my journey in CP research as the Research Program Manager of a highly productive multidisciplinary team at CSF Global in early 2016.

My initial role was to support the Bangladesh CPQoL study, Dr Rosalie Power's doctoral research describing the health-related quality of life, mental health, and sexual and

reproductive health of adolescents with CP in rural Bangladesh, and mental health of their primary caregivers. Alongside this role, I had the opportunity to contribute to a range of ongoing research projects and related service programs at CSF Global. This role enabled me to gain an in depth understanding of the various challenges of one of the most marginalized populations in the world, children with disability and their families in low-and middle-income countries (LMICs) such as Bangladesh with a special focus on CP.

CEREBRAL PALSY

Definition

CP is the most common cause of physical disability in childhood. It is a heterogenous condition, both in causation and clinical characteristics. The definition of CP underwent several changes as the understanding of CP advanced over the years. The most recent and widely used definition of CP defines it as "a group of permanent disorders of the development of movement and posture, causing activity limitation, that are attributed to non-progressive disturbances that occurred in the developing fetal or infant brain. The motor disorders of cerebral palsy are often accompanied by disturbances of sensation, perception, cognition, communication, and behaviour; by epilepsy, and by secondary musculoskeletal problems".(1) Due to the multidimensional nature of the condition and the advent of knowledge, there is continuing debate on the definition of CP.(2)

Burden of cerebral palsy

According to global estimates there are an astounding 50 million people with CP worldwide.(3) This estimate however, predominantly relies on data from high income countries (HICs) due to the overall dearth of evidence on the epidemiology of CP from LMICs where the burden of CP is disproportionately higher. Available research from LMICs report a range of rates of CP.(4, 5) These differences are in part due to variation in case definition which is unsurprising due to the heterogeneity of the condition. Methodological differences (i.e. population-based versus institution/hospital-based data) also contribute to these differences.(6, 7) Generation of population representative data using standard/harmonised definition of CP is therefore imperative for true estimates of the burden of CP from LMICs.

Classifications of cerebral palsy

The motor impairments of CP are classified into four main subtypes: spastic, dyskinetic, ataxic and hypotonic. Spastic CP, the most common type of CP, is further classified into monoplegia, diplegia, triplegia and quadriplegia based on the distribution of motor impairment.

The severity of motor impairment of CP is widely classified into five subgroups using the Gross Motor Functional Classification System (GMFCS); level I: walks without limitations, level II: walks with limitations, level III: walks using a hand held mobility device, level IV: self-mobility with limitations, may use powered mobility, and level V: transported in a manual wheelchair.(8)

CP is classified into two groups based on the timing of the injury responsible for causing CP: prenatal/perinatal and postnatal. However, understanding the complex causal pathway of CP is challenging.(9)

Risk factors for cerebral palsy

A variety of risk factors of CP spanning from preconception to post neonatal period have been identified. Yet they are often difficult to dissect for the development of appropriate preventive strategies.(10, 11) Recent studies from LMICs show that some risk factors/antecedents of CP (i.e. hypoxic ischaemic encephalopathy, severe neonatal jaundice/kernicterus, low birth weight, intrauterine growth retardation and infections) are more common in low resource settings.(4, 12-14) Public health measures and perinatal care to prevent CP with concurrent temporal data collection on the incidence of CP through long standing CP registers have proven to be instrumental in gaining a better understanding of causal pathways of CP and monitoring trends in HICs.(15, 16) With the emerging CP registers such initiatives are now possible in LMICs.(4) Furthermore, the role of socioeconomic factors on the risk factors of CP will additionally have significant bearing on prevention, causation and clinical outcomes of CP in low resource settings.(17-20)

CEREBRAL PALSY REGISTERS

Long standing CP registers in high income settings, such as Australia and Europe, enable reporting of robust data on the incidence, prevalence, aetiologies, risk factors and temporal

trends over decades.(21, 22) Such data is indispensable to the design and success of preventive strategies, interventions, and they inform policy and service delivery models. Furthermore, CP registers continue to serve as the sampling frame for further research to advance the knowledge of CP across the world [10]. The merits of CP registers are apparent from the declining birth prevalence of CP on HICs. According to the most recent international reports, there has been an astounding 35.0% decline over a 15-year period owing to multifaceted factors including a variety of strategies (i.e. improved perinatal care, antenatal magnesium sulphate, therapeutic hypothermia, the increased uptake of early detection and consequent early intervention) informed by evidence from CP registers.(15, 23) It is now anticipated that with such continued success, there will be further decline in the rates of CP in HICs and result in a shift in focus to the preventive sphere in LMICs.(15)

Better understanding of the risk factors and aetiopathogenesis of CP, in addition to the ensuring implementation of preventive strategies and improved perinatal care, has resulted in reported declines in prevalence in HICs and the changing epidemiology of CP.(15, 23) Working towards the replication of these successes in LMICs is imperative. Lessons from HICs such as Australia where reduced birth prevalence and the lowest rates of CP are observed can accelerate breakthroughs in prevention, treatment and cure for CP globally.(22) It is encouraging that the exponential increase in childhood CP research in the recent decades is extending into the LMICs where majority of the burden is observed.

In the absence of such rigorous studies reporting population-based findings from LMICs, data from HICs and extrapolation of institution-based findings were often used to estimate the burden and the diverse health care needs of children with CP. Prior to the commencement of this doctoral program in 2018, there was an overall lack of scientific evidence on CP in Bangladesh. In particular, there were gaps in the knowledge of CP in the areas of epidemiological research, effective intervention, and service utilization. There is an urgent need to gain a better understanding of the etiology of CP in these settings. Contemporary data on the severity and associated impairments of CP, rehabilitation status is also invaluable for the identification and implementation of appropriate strategies for children living with CP. The majority of the known risk factors of CP continue to prevail in LMICs. In the era of increasingly available evidence and the successes observed in other parts of the world,

addressing the known preventable risk factors for CP in LMICs is warranted. The majority of the known risk factors of CP continue to prevail in LMICs.

Population-based CP registers are a powerful tool for epidemiological research(24) and generate the most accurate data for of incidence, prevalence, risk factors and temporal trends.(25) There are over 40 CP registers across the world, predominantly in HICs. Long standing CP registers have yielded robust data over decades and significantly enhanced our knowledge on the epidemiology of CP and enabled monitoring of trends with advances in medicine, public health initiatives, preventive strategies, interventions, and disability inclusive policies. These registers are powerful tools for advocacy, in the strategy to describe aetiology, prevent CP and find evidence-based treatments.(26) There is a dire need for representative population data to support researchers, clinicians, service providers and policy makers for an evidence-based approach to prevention, early detection, and intervention for CP, particularly in LMICs.

We therefore focused our efforts on developing and utilizing a population-based surveillance of children with CP, the Bangladesh Cerebral Palsy Register (BCPR), to generate vital data. The BCPR study is a collaboration between Cerebral Palsy Alliance Research, the University of Sydney, Australia and CSF Global, Bangladesh and funded by the Cerebral Palsy Alliance Research Foundation. This study and further research inspired by the findings from the BCPR forms the core part of the research and related services for children with CP that I contributed to as part of my roles at CSF Global, the University of Sydney and Cerebral Palsy Alliance Research Institute prior to and during my doctoral candidature. This experience fostered my interest in registers and the tremendous potential they hold in generation of robust data to facilitate evidence-based approaches to interventions and future projects to enhance the wellbeing of children with CP and their families in low resource settings.

RESEARCH CONTEXT: BANGLADESH

Bangladesh, officially the People's Republic of Bangladesh, has been an independent nation since the liberation war in 1971. It is a south east Asian riverine country on the Bay of Bengal, bordered predominantly by India, and a short border with Myanmar (Figure 1).²

Although a small country spread across 148,460 square kilometres, it is the 8th most populous country in the world with a population density of ~1161 people per square kilometre, with highest density observed in and around the capital of Dhaka. ~166 million people reside in



Figure 1. Map of Bangladesh

Bangladesh; the median age of the population is 26.3 years. Bengali is the mother tongue of the entire population of Bangladesh apart from the indigenous minority groups.

Bangladesh has eight divisions and 64 districts, these however, have a limited role in public policy. For the purposes of local government, the country is divided into 577 sub-districts (upazillas) including the metropolitan thanas.(27)

Once the second poorest country in the world, Bangladesh has since shown remarkable strides in poverty reduction and reached the lower-middle-income status in 2015.(28) Bangladesh is one of the fastest growing economies in the world in the past decade.(28) Yet a mere 3.0% of the gross domestic product accounted for the total health expenditure per capita.(29) As a result, a significant proportion of health expenditure is out of pocket expense within the pluralistic health system of Bangladesh.(30, 31) This further contributes to the

² Image source : <u>https://www.britannica.com/place/Bangladesh</u>

challenges of meeting the complex health care needs of people with CP and associated impairments.

BANGLADESH CEREBRAL PALSY REGISTER

In 2015, recruitment into the BCPR commenced in a rural sub-district of Bangladesh (i.e. Shahjadpur) and the BCPR study protocol was published for wider dissemination.(32) This was soon followed by the expansion of the CP register to other parts of Bangladesh, and the first population-based data from an LMIC CP register defining the epidemiology was published which I co-authored during my doctoral candidature.(33) The published manuscript is included in *Appendix I*. The BCPR is an ongoing surveillance program generating robust data on prevalence, sociodemographic and clinical characteristics, risk factors, nutrition, education, and rehabilitation status of children with CP in Bangladesh.

BCPR Study Site

The BCPR covers a defined geographical area, the Shahjadpur subdistrict of Sirajganj located in the northern part of Bangladesh. The known denominator population in the study area has enabled population-based estimations using the BCPR data.

The study site, Shahjadpur, has 296 villages with an estimated 70,998 households. It has a total population of 561,076 (child population ~ 226,114) and 12,117 live births per annum.(34)

BCPR Case Definition

CP case definition was based on the Surveillance of CP in Europe and Australian CP Register.(35, 36) This enables the use of any definition for CP including the following five key elements common to the published definitions.

Cerebral palsy:

- i. Is an umbrella term for a group of disorders,
- ii. Is a condition that is permanent but not unchanging,
- iii. Involves a disorder of movement and/or posture and of motor function,
- iv. Is due to a non-progressive interference, lesion, or abnormality, and
- v. The interference, lesion, or abnormality originates in the immature brain.

BCPR Case Ascertainment

Children with suspected CP were identified using the Key Informant's Method (KIM).(32, 37) This method involves the engagement and training of local volunteers who reside in the same community as the potential study participants i.e., children with CP, thus enabling recruitment from the communities. The Key Informants received structured training on the identification of children with suspected CP. The Key Informants then identified and listed potential cases from their communities and informed the families of the date and venue of the upcoming medical assessment camps within their vicinity.

BCPR Assessment and Data Collection

The medical assessment camps were conducted on a regular basis in the study area. The children identified by the Key Informant underwent clinical assessment for confirmation of the diagnosis of CP by the medical assessment team prior to registration into the BCPR. A multidisciplinary medical assessment team including a paediatrician, a physiotherapist, and a counsellor conducted a detailed medical assessment for data collection. The BCPR registration form, a modified version of the Australian CP Register record form, was used to collect information on perinatal history, clinical characteristics and risk factors for CP. Neurological types of CP, along with Gross Motor Function Classification System (GMFCS)(38) and Manual Ability Classification System (MACS) levels,(39) were assessed and documented.

Anthropometric measurements were taken using standard measuring instruments and guidelines for the assessment of nutritional status. Detailed methodology and relevant study findings on the nutritional status of the BCPR cohort has been reported in a separate study not included in this thesis.(40)

Data on the presence and severity of associated impairments including epilepsy, visual, hearing, speech, and intellectual impairments were documented based on review of limited available medical records, report by the parents or primary caregivers of the children with CP, and clinical assessment by the medical assessment team. Information on sociodemographic characteristics, education and rehabilitation status were collected to identify the diverse needs of the children with CP and their families in Bangladesh.

28

BCPR Study Findings

The BCPR has enabled comprehensive epidemiological exploration and generated populationbased data on children with CP in Bangladesh.(33) Between January 2015 and December 2016, 726 children with CP were identified and registered in the BCPR. The mean age was 7.6 years (standard deviation [SD] 4.5 years) and 38.2% were female. The observed prevalence was 3.4 per 1000 children (95% confidence interval [CI]: 3.2–3.7), resulting in an estimated 233,514 children (95% CI: 219 778–254 118) with CP in Bangladesh. The mean age at the time of diagnosis of CP was 5.2 years (SD 3.8 years), which is substantially delayed compared to HICs and well beyond the age of optimal neuroplasticity of the brain.(41)

The type of CP was predominantly (79.6%) spastic CP of whom 25.6% (n=186) had quadriplegia. In total, 68.2% (n=495) were described as Gross Motor Functional Classification System (GMFCS) level III to V. 79.6% had at least one associated impairment; speech 67.6%, intellectual 39.0%, epilepsy 23.7%, visual 10.2%, and hearing 10.2%. The majority (78.2%) had never received any form of rehabilitation. Causes of CP in Bangladesh comprised potentially modifiable aetiologies. Perinatal causes of CP included intrapartum-related neonatal respiratory depression, neonatal encephalopathy, and infections; while post-neonatal causes included infection, drowning, and trauma.(33)

The BCPR is an ongoing surveillance which continues to generate robust data and facilitate further research and services for children with CP and their families in Bangladesh. This expansive amount of work has led to international collaborations for the initiation of work on CP registers and surveillance in low resource settings across the world.(4) Different study methodologies have been adopted for the collection of different but equally valuable data from several countries. While population-based surveillance such as the BCPR is the best source of epidemiological data,(24, 33) institution-based studies also yield valuable insights. They are an alternative method of data collected with different aims. One such study, a hospital-based surveillance of children with CP in Vietnam, have been described in Chapter 5 of this thesis.(42-44)

THESIS OUTLINE AND AIMS

The BCPR which was modelled on the Australian CP Register is considered the prototype of CP registers in LMICS and has enabled estimates of prevalence, facilitated clinical surveillance, and continues to inform future research and promote translational research to improve the lives of children with CP and their families in Bangladesh. This thesis addresses some of the gaps identified through the initial BCPR study.(33) Key findings and the ensuing research stimulated by the BCPR study constitute this thesis. These have been summarized in the following thesis outline and thesis concept map.

The *Introduction and Research Focus* described the background to this thesis. It introduces the state of the evidence on CP in Bangladesh with the specific focus on the BCPR study and demonstrates that CP registers are a useful tool for framing disability research even in LMIC settings.

Through the BCPR study, we identified that a substantial burden of CP among children in Bangladesh with a prevalence of 3.4 per 1000 children resulting in an estimates 233,514 children with CP in the country. Furthermore, we identified the following important areas that require further exploration:

- A substantial burden of CP among children in Bangladesh with a prevalence of 3.4 per 1000 children resulting in an estimates 233,514 children with CP in the country
- Greater severity of motor impairments compared to HICs; 25.6% had quadriplegia and 68.2% were described as GMFCS level III to V
- Substantial burden of associated impairments including epilepsy reported among 23.7% of the children
- Delayed diagnosis at 5.2 years of age
- Limited opportunities for interventions for CP with a mere 21.8% of the children receiving any form of rehabilitation

This doctoral research focuses on addressing some of these gaps. The overall aim of this thesis is to describe the clinical characteristics of CP, with concurrent exploration of measures for early identification and intervention for children with CP in low resource settings.

The aims of the thesis are to:

i. Better understand the clinical characteristics of children with CP in Bangladesh with a specific focus on hip dysplasia (*Chapter 1*) and epilepsy (*Chapter 2*)

ii. Explore pathways to improve age at diagnosis of CP in Bangladesh (Chapter 3)

iii. Enhance interventions for children with CP and their primary caregivers in Bangladesh through a quasi-experimental study assessing the outcome of an existing program in Bangladesh (*Chapter 4*)

iv. Describe CP in another low resource setting using an alternative epidemiological tool through a hospital-based surveillance in Vietnam (*Chapter 5*)

Gaps	Aims	Thesis Chapter and Manuscript	Key Findings
Limited population representative data on CP from LMICs such as Bangladesh	To examine the prevalence, clinical features, and risk factors for CP in Bangladesh	Introduction and Research Focus	 Prevalence: 3.4/1000 children, 233,514 children with CP in Bangladesh Majority had preventable risk factors Greater severity of impairments Diagnosis is delayed Limited opportunities for early intervention
Burden and severity of CP in low resource settings	To describe the burden of hip dysplasia from a population-based study in Bangladesh	Chapter 1. <i>Hip dysplasia among children</i> <i>with spastic cerebral palsy in</i> <i>rural Bangladesh</i>	 40 children with spastic CP (80 hips) had pelvic X-Rays 62.5% GMFCS level III-V 20% had hip subluxation
Burden of associated impairments including epilepsy	To define the prevalence and seizure subtypes among children with CP in rural Bangladesh and explore barriers to optimum epilepsy control	Chapter 2. Improving epilepsy control among children with Cerebral Palsy in rural Bangladesh	 62.0% had ongoing epileptic seizures; none with prolonged seizures had an emergency seizure management plan. Non-epileptic events were being pharmacologically treated as seizures in 18.1% Financial constraints were the main reason for non-compliance
Diagnosis of CP in low resource settings	To explore the feasibility of the use of the DMC to assess childhood neurodevelopment as part of regular medical check ups	Chapter 3. Use of developmental milestone chart (DMC) in rural Bangladesh to educate health workers and stimulate referral for early diagnosis and intervention	 Diagnosis of CP is delayed in low resource settings DMC can be used for developmental screening as part of regular health check- ups in low resource settings
Ways for early intervention and rehabilitation of children with CP in low resource settings	To evaluate the outcome of an existing program for children with CP and their primary caregivers in a rural subdistrict of Bangladesh.	Chapter 4. Outcome of community-based parents led early intervention for children with cerebral palsy in rural Bangladesh: A quasi- experimental study	• A community-based early intervention and rehabilitation program in low-resource setting is effective in improving motor outcomes and speech of children with CP as well as helping with the depressing, anxiety and stress of their primary caregivers
Lack of data on epidemiology and clinical features of CP from other LMICs	To identify children with CP and describe their clinical characteristics in Vietnam To assess the burden of malnutrition and the underlying risk factors among children with CP in Vietnam	Chapter 5. Data on Cerebral Palsy in Vietnam: Findings from Prospective Hospital-Based Surveillance in Vietnam Nutritional Status of Children with Cerebral Palsy: Findings from Prospective Hospital- Based Surveillance in Vietnam	 Children predominantly had spastic CP (95.2%) Most (69.6%) were quadriplegic 60.3% were GMFCS level III-V 28.9% underweight and 29.0% stunted Significantly greater odds of underweight among children aged >5 years Underweight and/or stunting high among children with quadriplegia (81%, 84.5%) and GMFCS level IV-V





KEY FINDINGS



 62% had ongoing seizures; none with prolonged seizures had an emergency plan
 18.1% had non-epileptic events treated as seizures
 Financial constraints caused non-compliance

• DMC can be used for developmental screening as part of regular health check-ups

→ To evaluate the feasibility of best practice tools for early diagnosis of CP in Bangladesh as part of a funded ongoing study

 this intervention is effective in improving,
 motor outcomes and speech of children with CP
 mental wellbeing of their primary caregivers

95.2% had spastic CP:
69.6% quadriplegic &
60.3% GMFCS levels III-V
28.9% underweight &
29.0% stunted

→ To maintain the existing hospital-based surveillance & promote best practices for CP in Vietnam; funded study for three years commencing July 2021

To describe the burden of hip dysplasia in Bangladesh To define the prevalence High burden & seizure subtypes 2 & severity of among children with CP CP in in rural Bangladesh & Bangladesh High burden explore barriers to of epilepsy optimum epilepsy control among children with СР 3 To explore the feasibility Delayed of DMC for diagnosis of neurodevelopmental CP at 5.3 years of age screening as part of in Bangladesh regular health check-ups BANGLADESH 4 To evaluate the outcome of an existing program for Limited children with CP & their opportunities primary caregivers in for early Bangladesh intervention in Bangladesh To identify children Lack of such with CP & describe their vital data from other epidemiology in LMICs hospital-based settings • To assess the burden of malnutrition & its risk factors among children with CP

CHAPTER 1. HIP DYSPLASIA AMONG CHILDREN WITH SPASTIC CEREBRAL PALSY IN RURAL BANGLADESH



1.1 INTRODUCTION

Children with CP are at risk of developing progressive hip displacement.(45) It is the second most common musculoskeletal deformity among children with CP, more commonly observed among children with more severe functional limitations.(45) Several studies have confirmed a strong relationship between the risk of hip displacement and functional ability of children with CP as classified by the GMFCS and those with spastic quadriplegia.

Findings from the BCPR study showed that majority of the children with CP in Bangladesh had severe functional impairments (23.7% GMFCS IV and 22.9% GMFCS V) and the majority (79.6%) have spastic CP of whom a substantial proportion (25.6%) of the children were reported to have spastic quadriplegia.(33) Thus, a significant proportion of these children were at high risk of hip displacement. However, population-based data on hip dysplasia in children with CP from Bangladesh were not available.

In this study we aimed to describe the burden of hip dysplasia among children with CP in Bangladesh using a population-based CP register (i.e. BCPR) as the sampling frame.

1.2 PAPER #1 HIP DYSPLASIA AMONG CHILDREN WITH SPASTIC CEREBRAL PALSY IN RURAL BANGLADESH

RESEARCH ARTICLE

Hip dysplasia among children with spastic cerebral palsy in rural Bangladesh

Tasneem Karim^{1,2,3,4*}, Mahmudul Hassan Al Imam^{1,2}, Prue Golland⁴, Aynul Islam Khan^{1,5}, Iqbal Hossain⁶, Hayley Smithers-Sheedy⁴, Nadia Badawi^{3,4}, Mohammad Muhit^{1,2} and Gulam Khandaker^{1,2,3,7}

Abstract

Background: Hip dysplasia is common among children with cerebral palsy (CP), particularly in spastic CP. It can result in pain, reduced function and quality of life. However, the burden of hip dysplasia among children with CP in llow-and middle-income countries (LMICs) like Bangladesh is unknown. We aimed to define the burden of hip dysplasia among children with spastic CP in Bangladesh.

Methods: This study includes a subset of the Bangladesh CP Register (BCPR) study cohort who were registered between January and March 2015. The BCPR is a population-based surveillance of children with CP (aged < 18 years) operating in a northern sub-district (Shahjadpur; child population ~ 226,114) of Bangladesh. Community-based key informant's method (KIM) survey conducted to identify children with CP in the surveillance area. A diagnosis of CP was made based on clinical history and examination by the study physicians and physiotherapist. Study participants had an antero-posterior (AP) X-ray of their pelvis. The degree of subluxation was assessed by calculating the migration percentage (MP).

Results: During the study period, 196 children with CP were registered, 144 had spastic CP. 40 children with spastic CP (80 hips) had pelvic X-Rays (mean age 9.4 years, range 4.0-18.0 years) and 32.5% were female. Gross Motor Function Classification System (GMFCS) showed 37.5% (n = 15) with GMFCS level I-II and 62.5% (n = 25) with GMFCS level III-V. Twenty percent (n = 8) of the children had hip subluxation (MP: 33-80%). Osteopenic changes were found in 42.5% (n = 17) children.

Conclusions: To the best of our knowledge this is one of the first studies exploring hip dysplasia among children with spastic CP in Bangladesh. Our findings reflect that hip dysplasia is common among children with spastic CP. Introduction of hip surveillance programmes is imperative for prevention of secondary complications, reduced function and poor quality of life among these children.

Keywords: Hip dysplasia, Cerebral palsy, Children, Bangladesh, Surveillance

Background

Cerebral palsy (CP) is a heterogenous group of conditions that affects the developing brain, resulting in a permanent non-progressive dysfunction of the central nervous system manifested by disorders of motor function, movement, and posture [1]. It is the leading cause of childhood physical disability globally affecting two to three children per thousand live births [2, 3]. However, the burden of CP is even higher in low-and middle-

* Correspondence: tasneem.karim.tk@gmail.com

¹CSF Global, Dhaka, Bangladesh

© The Author(s). 2019 **Open Access** This article is distributed under the terms of the Creative Commons Attribution 4.0 International License (http://creativecommons.org/licenses/by/4.0/), which permits unrestricted use, distribution, and reproduction in any medium, provided you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons license, and indicate if changes were made. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated.

income countries (LMICs). Recent population-based studies from Bangladesh and Uganda reported population prevalence of CP as 3.4 and 2.9 per 1000 children respectively [4, 5].

The motor disorders of CP are often associated with musculoskeletal anomalies [6], of which hip displacement is the second most common abnormality preceded only by the abnormalities of the foot and ankle. It affects nearly one-third of the children with CP and the incidence varies from 1.0% in children with spastic hemiplegia, up to 75.0% in those with spastic quadriple-gia [7, 8]. Progressive hip displacement leading to dislocation can be catastrophic and result in extreme





Open Access

 $^{^2\}mbox{Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka, Bangladesh$

Full list of author information is available at the end of the article
pain, reduced function and quality of life [9, 10]. Other complications include severe contracture, pelvic obliquity and scoliosis, resulting in poor sitting, standing and walking abilities [9, 11, 12]. Most of the studies on hip dysplasia among children with CP are reported from high-income countries, and mostly from areas with an existing hip surveillance programme. However, there is limited data on hip dysplasia from LMICs such as Bangladesh where the burden of CP is substantially high [4].

Studies recommend that radiological screening of children with spastic quadriplegia [13] and GMFCS level IV and V [14] should commence from less than 18 months of age [13]. It is also recommended that children with CP with a migration percentage (MP) of 33 to 40 should undergo radiological examination at 6 months interval prior to surgery [13]. Early detection and management of hip dysplasia can maintain flexible, properly located and pain-free hips with availability of symmetrical range of movement [15]. Introduction of formal surveillance programmes has demonstrated a significant reduction in the incidence of hip dislocation in populations of children with CP in high-income countries [14]. Hip dysplasia and dislocation are preventable through early identification and intervention [8]. However, the burden of hip dysplasia among children with CP in LMICs such as Bangladesh is unknown.

We aimed to describe the burden of hip dysplasia among a population-based cohort of children with CP in rural Bangladesh using the Bangladesh CP Register (BCPR) to inform clinicians and public health professionals [16].

Methods

Study design, setting and participants

This study is a part of the BCPR project. BCPR is one of the first population-based CP registers in a LMIC setting and has been ongoing since January 2015 in a northern part of Bangladesh i.e. Shahjadpur (child population ~ 226,114) [17]. Children with CP registered with the BCPR were identified using a novel method, the Key Informant Method (KIM) which involves training of local volunteers to serve as key informants (KIs) and list children with CP from their community. The children then undergo detailed clinical assessment by a multidisciplinary team for confirmation of diagnosis and registration into the BCPR. The details of the BCPR study has been described previously [4, 16].

Between January and March 2015, a sub-group of the BCPR registrants were included in this study. Due to limited available resources, every third child from the BCPR cohort with spastic CP was selected for radiological assessment. In the event that any prospective study participant was unwilling the next child was approached for participation in the study.

Clinical assessment

Children with CP were identified from the community using the KIM, [18] and a detailed neurological examination was performed by a trained paediatrician and physiotherapist during the clinical assessment and BCPR registration process. Sociodemographic and clinical data were collected using BCPR record forms. Motor type and motor severity including Gross Motor Functional Classification System (GMFCS) levels were determined following standard guideline [19]. The predominant motor type of CP included spastic, dyskinetic or ataxic movement motor types based on clinical assessment by a paediatrician. Spastic topography was further classified as mono/hemiplegic, diplegic, triplegic, and quadriplegic.

Radiological assessment

Children with spastic CP had an anteroposterior (AP) pelvic X-ray. Hip displacement was measured from an AP radiograph of the hips using existing protocol [20]. The degree of subluxation (dysplasia) was assessed by calculating the MP i.e. the percentage of the bony width of the femoral capital epiphysis which falls lateral to a line drawn vertically from the bony lateral margin of the acetabulum, the Perkins line. This technique has been found useful to determine hip dysplasia among children [21]. Studies of validity have shown that a change greater than 8.3% in MP represents a real change in displacement of the femoral head with 95% confidence [20]. The MP found among the subjects were classified as normal hip (MP: <32%), and subluxated hip (MP: 33 – 80%) [22]. Radiological diagnosis of osteopenia was made by the radiologist after qualitative assessment of the X rays. Qualitative features of osteopenic bone included increased radiolucency, vertical striation, end plate thinning and accentuation of cortical margins of any vertebra including changes in shape.

Ethical consideration

The study was approved in Australia by the Cerebral Palsy Alliance NHMRC Human Research Ethics Committee (Ref no. 2015–03–02), and in Bangladesh by the Asian Institute of Disability and Development (AIDD) Human Research Ethics Committee (southasia-irb-2014-l-01) and Bangladesh Medical Research Council (BMRC) HREC (BMRC/NREC/ 2013–2016/1267). Informed written consent was obtained from the primary caregiver of the children prior to recruitment into the study.

Statistical analysis

The data were processed using SPSS (Statistical Package for Social Science, version 22.0 for windows; SPSS Inc., Chicago, IL, USA). The measures of central tendency (mean) and dispersion (SD) were used. Descriptive analyses were done to present the proportion of dysplastic hips according to the background factors (age, sex, parental education), and degree of motor impairment of the children with CP.

Results

Out of the 196 BCPR registrants during the study period, 73.4% (n = 144) had spastic CP of which 40 children underwent radiological assessment. The mean age was 9.4 ± 4.0 years (range 4.0–18.0 years) and 32.5% (n = 13) of the study cohort were female.

The majority of the mothers had completed primary education (42.5%, n = 17) whereas more than one third of the fathers had not have any education (37.5%, n = 15). The median age of diagnosis of CP was 3 years. The majority (37.5%, n = 15) of the children had diplegia. 37.5% (n = 15) were described as GMFCS level I-II and 62.5% (n = 25) as GMFCS level III-V. Large numbers of children with CP (60%, n = 24) had not ever received rehabilitation services. Twenty percent (n = 8) of the children had hip subluxation of which five children were between 5 to 9 years of age (Table 1).

Hip subluxation and age CP diagnosis, type of CP, GMFCS and rehabilitation status

The majority (75.0%, n = 6) of the children who had subluxation of hips were diagnosed with CP between 24 to 47 months of age. Children with subluxated hips had spastic diplegia (50.0%, n = 4), quadriplegia (37.5%, n = 3) and hemiplegia/monoplegia (12.5%, n = 1). Children with GMFCS levels III -V had the highest proportion of subluxated hips; 37.5% (n = 3). Among the children with hip subluxation 62.5% (n = 5) never accessed any rehabilitation services (Table 2).

Qualitative findings

On qualitative assessment of the x-rays by a radiologist we found that 42.5% (n = 17) of the children had signs of osteopenic changes of which 94.1% (n = 16) had mild osteopenia and 5.8% (n = 1) had moderate osteopenic changes.

Discussion

The aim of the study was to investigate the burden of hip dysplasia among children with spastic CP from a population-based CP register in rural Bangladesh. We found that a fifth of our cohort had subluxation of hips which has yielded valuable insights into the high burden of hip dysplasia among children with spastic CP in Bangladesh.

Subluxation of hips was more commonly observed among children over 5 years of age and with more severe functional motor limitations (i.e. GMFCS levels III - V). This is consistent with other studies from high income countries which showed that the more severe outcomes of hip dysplasia are observed among older children and

Table 1 Socio-demographic characteristics, age of CP diagnosis,
neurological subtype of Spastic CP, gross motor function
classifications and rehabilitation status of children with CP

Characteristics	n (%)
Gender	
Male	27 (67.5)
Female	13 (32.5)
Age (years)	
< 5	6 (15.0)
5–9	18 (45.0)
10–14	12 (30.0)
15–18	4 (10.0)
Maternal education	
Illiterate	11 (27.5)
Primary	17 (42.5)
Secondary	9 (22.5)
More than Secondary	3 (7.5)
Paternal education	
Illiterate	15 (37.5)
Primary	10 (25.0)
Secondary	10 (25.0)
More than Secondary	5 (12.5)
Age of diagnosis of CP (months)	
0–23	1 (2.5)
24–47	21 (52.5)
48–84	11 (27.5)
85 & above	7 (17.5)
Spastic subtype	
Hemi/monoplegia	11 (27.5)
Diplegia	15 (37.5)
Triplegia	4 (10.0)
Quadriplegia	10 (25.0)
GMFCS levels	
Level I	5 (12.5)
Level II	10 (25.0)
Level III	12 (30.0)
Level IV	4 (10.0)
Level V	9 (22.5)
Received rehabilitation services	
Yes	16 (40.0)
No	24 (60.0)

among those with higher GMFCS levels [7, 14, 23]. Furthermore, all the children with subluxation of hips in our cohort were diagnosed with CP after 24 months of age and nearly two thirds of them had not ever received any rehabilitation services.

	Right		Left		Total	
Age	Normal	Subluxated	Normal	Subluxated	Normal	Subluxated
0–4 years ($n = 6$)	5 (14.3)	1 (20.0)	5 (13.5)	1 (33.3)	10 (13.9)	2 (25.0)
5–9 years ($n = 18$)	14 (40.0)	4 (80.0)	17 (45.9)	1 (33.3)	31 (43.1)	5 (62.5)
10-14 years (n = 12)	12 (34.3)	0 (0.0)	11 (29.7)	1 (33.3)	23 (31.9)	1 (12.5)
15 and above $(n = 4)$	4 (11.4)	0 (0.0)	4 (10.8)	0 (0.0)	8 (11.1)	0 (0.0)
Maternal Education						
Illiterate ($n = 11$)	11 (31.4)	0 (0.0)	11 (29.7)	0 (0.0)	22 (30.6)	0 (0.0)
Primary ($n = 16$)	15 (42.9)	2 (40.0)	15 (40.5)	2 (66.7)	30 (41.7)	4 (50.0)
Secondary ($n = 9$)	7 (20.0)	2 (40.0)	8 (21.6)	1 (33.3)	15 (20.8)	3 (37.5)
More than Secondary $(n = 3)$	2 (5.7)	1 (20.0)	3 (8.1)	0 (0.0)	5 (6.9)	1 (12.5)
Paternal Education						
Illiterate ($n = 15$)	14 (40.0)	1 (20.0)	15 (40.5)	0 (0.0)	29 (40.3)	1 (12.5)
Primary ($n = 10$)	9 (25.7)	1 (20.0)	9 (24.3)	1 (33.3)	18 (25.0)	2 (25.0)
Secondary ($n = 10$)	7 (20.0)	3 (33.3)	9 (24.3)	1 (33.3)	16 (16.7)	4 (50.0)
More than Secondary $(n = 5)$	5 (14.3)	0 (0.0)	4 (10.8)	1 (33.3)	9 (12.5)	1 (12.5)
Age of diagnosis of CP (months)						
0–23 (<i>n</i> = 1)	1 (2.9)	0 (0.0)	1 (2.7)	0 (0.0)	2 (2.8)	0 (0.0)
24–47 (n = 21)	17 (48.6)	4 (80.0)	19 (51.4)	2 (66.7)	36 (50.0)	6 (75.0)
48–84 (<i>n</i> = 10)	10 (28.6)	1 (20.0)	10 (27.0)	1 (33.3)	20 (27.8)	2 (25.0)
85 & above (<i>n</i> = 7)	7 (20.0)	0 (0.0)	7 (18.9)	0 (0.0)	14 (19.4)	0 (0.0)
Spastic subtype						
Hemiplegia/ monoplegia (n = 11)	11 (31.4)	0 (0.0)	10 (27.0)	1 (33.3)	21 (29.2)	1 (12.5)
Diplegia (n = 14)	12 (34.3)	3 (60.0)	14 (37.8)	1 (33.3)	26 (36.1)	4 (50.0)
Triplegia ($n = 4$)	4 (11.4)	0 (0.0)	4 (10.8)	0 (0.0)	8 (11.1)	0 (0.0)
Quadriplegia ($n = 10$)	8 (22.9)	2 (40.0)	9 (24.3)	1 (33.3)	17 (23.6)	3 (37.5)
GMFCS						
- (n = 27)	25 (71.5)	2 (40.0)	25 (67.5)	2 (66.6)	50 (69.4)	4 (50.0)
IV-V (<i>n</i> = 13)	10 (28.6)	3 (60.0)	12 (32.4)	1 (33.3)	22 (30.5)	4 (50.0)
Received rehabilitation services						
Yes (n = 24)	23 (65.7)	1 (20.0)	22 (59.5)	2 (66.7)	45 (62.5)	3 (37.5)
No (<i>n</i> = 16)	12 (34.3)	4 (80.0)	15 (40.5)	1 (33.3)	2 (37.5)	5 (62.5)

 Table 2 Distribution of hip displacement according to socio-demographic factors, spastic CP sub-type, GMFCS and rehabilitation status

Findings from our study have a bearing on the importance of hip surveillance among children with CP in LMIC settings. Hip dysplasia can be detected early by monitoring through regular hip X-rays to facilitate timely treatment and can thereby avert complications which are detrimental to the quality of life of children with CP and their families [9, 10]. In contrast to high income countries, where a range of treatment options are available for the management of of children with CP, including botulinum toxin injection, bracing, soft tissue release surgery for spasticity, reconstructive surgeries and salvage procedure [24], the prognosis of children with CP is remarkably different in poorer countries where there is limited access to such services. In addition to the lack of services these children and their families face additional challenges due to socioeconomic factors and rampant stigma around disabilities such as CP [25].

The position of femoral head within the acetabulum up to the age of 5 years is important to ensure stability of the hip and acetabular development [26, 27]. Therefore, early detection of progressively displaced hips and referral for orthopedic management is imperative to prevent further displacement and to keep the femoral head in position within the age of 5 years [28]. However, early diagnosis of hip dysplasia is challenging particularly through physical examination alone. Regular clinical inspection for the range of abduction of the hip, and repeated radiological examination of the hips is necessary to identify hip dysplasia in time [8, 29, 30]. The aim of hip surveillance programmes that are based on the Consensus Statement on Hip Surveillance for Children with Cerebral Palsy is to ensure that progressive hip displacement is detected early enough to enable timely referral for orthopedic assessment and management [31]. Decrease in incidence of hip dislocation has been previously documented with the introduction of hip surveillance programme [13]. Incorporation of hip surveillance into routine care has been recommended for all children with CP. This includes radiological assessment between 12 to 24 months of age with subsequent management varying with gross motor severity i.e. GMFCS levels [14]. Studies also suggested that non-ambulatory children and those who have an annual increment of the MP by more than 7% require more vigilant monitoring and consideration for orthopedic referral [1, 32].

A study conducted in Norway and Southern Sweden compared the burden of hip dislocation among children with CP in regions with and without hip surveillance programmes [20]. The sample size at the two sites (n = 119 in Norway)and n = 136 in Southern Sweden) were comparable to our study cohort of 196 children with CP. The proportion of children at GMFCS levels III-V among both the study populations were substantially lower than our cohort; 34.0% in Norway and 38.0% in Southern Sweden population compared to 62.5% in our study cohort in Bangladesh [20]. It was observed that in Southern Sweden where hip surveillance services are provided the prevalence of hip dislocation was lower, fewer children required surgical intervention and the children underwent hip operations at an earlier age [20]. 20.0% of the children from our cohort had hip subluxation which is comparable to the findings from Norway (15.1%). Whereas Southern Sweden had a significantly lower proportion (0.7%, p < 0.001) of children with hip subluxation in presence of a surveillance programme [20].

Early identification of hip dysplasia is limited by the absence of hip surveillance programmes, thereby resulting in a high burden [31, 32]. A recent study conducted in India among 118 children with CP aged 2 to 12 years with GMFCS levels III-V showed that 12.7% had developmental dysplasia of spastic hip [21]. An even higher burden of hip dysplasia (54.7%) was reported in another study conducted in Malaysia [33].

Findings from these studies collectively illustrate the apparent benefits of hip surveillance programmes in the reduction of the burden of hip dysplasia and the aversion of associated complications which drastically impair the wellbeing of children with CP [20, 21, 33]. There is a dire need for prospective hip surveillance programmes for children with CP in LMICs. Our study has direct clinical implication; the established programmes in high

income countries and the method described in this study can be incorporated for the management of children with CP in LMICs. As radiological services are becoming increasingly available in semi-rural areas of LMICs such as Bangladesh, it is now possible to develop and implement hip surveillance programmes with a simplified protocol for these settings. Ongoing surveillance of children with CP such as the BCPR can serve as the basis for the implementation of recommended guidelines for hip dysplasia among children with CP in low resource settings. This will ensure early detection of hip dysplasia and subluxation which can thereby facilitate early intervention to avert and limit the adverse outcomes of hip dysplasia.

Study limitations

Despite our best efforts there are several limitations in this study which are inherent to observational studies. The cross-sectional design limited the establishment of causal relations between the independent variables and hip dysplasia. Radiography of hips were undertaken in a position as neutral as possible which was often difficult to achieve owing to spasm or contractures. We had to restrict or study only among children with spastic CP due to budget constraints. However, studies have shown that children with spastic CP are the most affected by hip dysplasia. Advanced statistical analyses were also limited due to the small number of children in subgroups (i.e. eight children with subluxation). Furthermore, caution is recommended in the interpretation of comparisons between studies due to methodological differences.

Conclusion

To the best of our knowledge this is one of the first studies exploring hip dysplasia among children with spastic CP in Bangladesh. Our findings from this selected sample suggest hip subluxation is common among children with spastic CP in rural Bangladesh. Further studies are needed to address the true extent of this complex complication of CP to improve the quality of life and ensure better prognosis among children with CP and their families. Introduction of hip surveillance programmes is imperative for the prevention of secondary complications, reduced function and poor quality of life among these children.

Abbreviations

AP: Antero-posterior; BCPR: Bangladesh CP Register; CP: Cerebral Palsy; GMFCS: Gross Motor Function Classification System; KIM: Key Informant Method; KIs: Key Informants; LMICs: Low and Middle-Income Countries; MP: Migration Percentage

Acknowledgements

We would like to acknowledge the CSF Global team in Bangladesh for their support in the implementation of the project and for supporting the children with CP and their families in access to services through a strong referral system.

Authors' contributions

GK and MM conceptualized, designed and implemented the study. GK was responsible for collection of the data. AIK and IH reviewed the X-rays and provided specialist input. TK wrote the first draft of the manuscript with input from all co-authors. MHAI and GK contributed to data analysis and interpretation. MM, HSS, PG and NB provided specialist advice in the study. All authors read and approved the final manuscript and were involved in the decision to submit the manuscript.

Funding

This study has been conducted as part of the BCPR study. BCPR is funded by Cerebral Palsy Alliance Research Foundation (PG4314) and through internal funding from CSF Global, Bangladesh. The funding bodies played no role in the design of the study; collection, analysis, and interpretation of data and in the preparation of the manuscript. TK is supported by Cerebral Palsy Alliance Research Foundation Career Development Grant a (CDG 04617). HSS is supported through an NHMRC ECF 1144566 and the Australasian Cerebral Palsy Clinical Trials Network.

Availability of data and materials

The datasets generated and/or analysed during the current study are not publicly available due to confidentiality but are available from the corresponding author on reasonable request.

Ethics approval and consent to participate

The study was approved in Australia by the Cerebral Palsy Alliance Human Research Ethics Committee (Ref no. 2015–03–02), and in Bangladesh by the Asian Institute of Disability and Development (AIDD) Human Research Ethics Committee (southasia-irb-2014-I-01) and Bangladesh Medical Research Council (BMRC) HREC (BMRC/NREC/2013–2016/1267). Written informed written consent was obtained from the primary caregiver of the children prior to recruitment into the study.

Consent for publication

Not applicable

Competing interests

The authors declare that there is no competing interest.

Author details

¹CSF Global, Dhaka, Bangladesh. ²Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka, Bangladesh. ³Discipline of Child and Adolescent Health, Sydney Medical School, The University of Sydney, Sydney, Australia. ⁴Cerebral Palsy Alliance Research Institute, The University of Sydney, Sydney, Australia. ⁵Singair Upazilla Health Complex, Singair, Manikganj, Bangladesh. ⁶Radiology and Imaging Department, Bangabandhu Sheikh Mujib Medical University (BSMMU), Dhaka, Bangladesh. ⁷Central Queensland Public Health Unit, Central Queensland Hospital and Health Service, Rockhampton, Queensland, Australia.

Received: 8 April 2019 Accepted: 30 August 2019 Published online: 27 October 2019

References

- Huser A, Mo M, Hosseinzadeh P. Hip surveillance in children with cerebral palsy. Orthop Clin North Am. 2018;49(2):181–90.
- Ozmen M, Caliskan M, Apak S, Gokcay G. 8-year clinical experience in cerebral palsy. J Trop Pediatr. 1993;39:52–4.
- Howard J, Soo B, Graham HK, Boyd RN, Reid S, Lanigan A, Wolfe R, Reddihough DS. Cerebral palsy in Victoria: motor types, topography and gross motor function. J Paediatr Child Health. 2005;41:479–83.
- Khandaker G, Muhit M, Karim T, Smithers-Sheedy H, Novak I, Jones C, Badawi N. Epidemiology of cerebral palsy in Bangladesh: a populationbased surveillance study. Developmental Medicine & Child Neurology. 2019; 61(5):601–9.
- Kakooza-Mwesige A, Andrews C, Peterson S, Wabwire Mangen F, Eliasson AC, Forssberg H. Prevalence of cerebral palsy in Uganda: a populationbased study. Lancet Glob Health. 2017;5:e1275–82.

- Rosenbaum P, Paneth N, Leviton A, Goldstein M, Bax M, Damiano D, Dan B, Jacobsson B. A report: the definition and classification of cerebral palsy April 2006. Dev Med Child Neurol Suppl. 2007 Feb 1;109(suppl 109):8–14.
- Soo B, Howard J, Boyd R, Reid S, Lanigan A, Wolfe R, et al. Hip displacement in cerebral palsy. J Bone Joint Surg Am. 2006;88(1):121.
- Miller F, Dabney KW, Rang M. Complications in cerebral palsy treatment. In: Epps Jr CH, Bowen JR, editors. Complications in pediatric orthopaedic surgery. Philadelphia: Lippincott Company; 1995. p. 477–544.
- Bagg MR, Farber J, Miller F. Long-term follow-up of hip subluxation in cerebral palsy patients. J Pediatr Orthop. 1993;13:32–6.
- Jung NH, Pereira B, Nehring I, Brix O, Bernius P, Schroeder SA, Kluger GJ, Koehler T, Beyerlein A, Weir S, von Kries R, Narayanan UG, Berweck S, Mall V. Does hip displacement influence health-related quality of life in children with cerebral palsy? Dev Neuro rehabil. 2014;17(6):420–5.
- Muthusamy K, Chu H, Friesen R, Chou P, Eilert R, Chang F. Femoral head resection as a salvage procedure for the severely dysplastic hip in nonambulatory children with cerebral palsy. J Pediatr Orthop. 2008;28(8):884–9.
- Moreau M, Drummond DS, Rogala E, Ashworth A, Porter T. Natural history of the dislocated hip in spastic cerebral palsy. Dev Med Child Neurol. 1979;21:749–53.
- Hägglund G, Andersson S, Düppe H, Lauge-Pedersen H, Nordmark E, Westbom L. Prevention of dislocation of the hip in children with cerebral palsy. J Bone Joint Surg Br. 2005;87-B(1):95–101.
- Wynter M, Gibson N, Willoughby KL, Love S, Kentish M, Thomason P, et al. Australian hip surveillance guidelines for children with cerebral palsy: 5-year review. Dev Med Child Neurol. 2015;57(9):808–20.
- Dobson F, Boyd R, Parrott J, Nattrass G, Graham H. Hip surveillance in children with cerebral palsy. J Bone Joint Surg Br. 2002;84-B(5):720–6.
- Khandaker G, Smithers-Sheedy H, Islam J, Alam M, Jung J, Novak I, et al. Bangladesh cerebral palsy register (BCPR): a pilot study to develop a national cerebral palsy (CP) register with surveillance of children for CP. BMC Neurol. 2015;15:173.
- Bangladesh Bureau of Statistics. Population and housing census 2011: national volume 2- union statistics. Bangladesh: statistics and informatics division, Ministry of Planning. 2014. Available from: http://203.112.218.65: 8008/WebTestApplication/userfiles/Image/National%20Reports/ Union%20Statistics.pdf.
- Mackey S, Murthy GV, Muhit MA, Islam JJ, Foster A. Validation of the key informant method to identify children with disabilities: methods and results from a pilot study in Bangladesh. J Trop Pediatr. 2012;58(4):269–74.
- Rosenbaum PL, Palisano RJ, Bartlett DJ, Galuppi BE, Russell DJ. Development of the gross motor function classification system for cerebral palsy. Dev Med Child Neurol. 2008;50(4):249–53.
- Elkamil AI, Andersen GL, Hägglund G, Lamvik T, Skranes J, Vik T. Prevalence of hip dislocation among children with cerebral palsy in regions with and without a surveillance programme: a cross sectional study in Sweden and Norway. BMC Musculoskelet Disord. 2011;12(1):284.
- Vykuntaraju KN, Manohar V, Lakskman RR, Ramaswamy P. Developmental dysplasia of spastic hip in children with cerebral palsy in southern India. Indian Pediatr. 2016;53(3):259–60.
- 22. Pountney T, Green E. Hip dislocation in cerebral palsy. BMJ. 2006; 332(7544):772–5.
- Terjesen T. The natural history of hip development in cerebral palsy. Dev Med Child Neurol. 2012;54(10):951–7.
- 24. Shore B, Spence D, Graham H. The role for hip surveillance in children with cerebral palsy. Curr Rev Musculoskelet Med. 2012;5:126–34.
- Benfer KA, Jordan R, Bandaranayake S, Finn C, Ware RS, Boyd RN. Motor severity in children with cerebral palsy studied in a high-resource and lowresource country. Pediatr. 2014;134(6):e1594–602.
- Harris NH, Lloyd-Roberts GC, Gallien R. Acetabular development in congenital dislocation of the hip: with special reference to the indications for acetabuloplasty and pelvic or femoral realignment osteotomy. J Bone Joint Surg Br. 1975;57-B:46–52.
- 27. Kalen V, Bleck EE. Prevention of spastic paralytic dislocation of the hip. Dev Med Child Neurol. 1985;27:17–24.
- 28. Scrutton D. The early management of hips in cerebral palsy. Dev Med Child Neurol. 1989;31:108–16.
- Rang M. Cerebral palsy. In: Morrissy RT, editor. Lovell and Winter's pediatric orthopaedics. 3rd ed. Philadelphia: Lippincott Company; 1990. p. 465–506.
- 30. Scrutton D, Baird G. Surveillance measures of the hips of children with bilateral cerebral palsy. Arch Child. 1997;76:381–4.

- Wynter M, Gibson N, Kentish M, Love S, Thomason PK, Kerr Graham H. The consensus statement on hip surveillance for children with cerebral palsy: Australian standards of care. J Pediatr Rehabil Med. 2011;4(3):183–95.
- Gordon G, Simkiss D. A systematic review of the evidence for hip surveillance in children with cerebral palsy. J Bone and Joint Surg Br. 2006; 88-B(11):1492–6.
- Ching BH, Khoo TB. Prevalence and predictive factors of hip displacement in children with cerebral palsy at Paediatric institute, Kuala Lumpur hospital. Neurol Asia. 2017;1(3):22.

Publisher's Note

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions



1.3 CHAPTER SYNOPSIS

In this chapter, the burden of hip dysplasia among children with CP in Bangladesh has been described. The findings show that 20.0% of the children had hip subluxation and 42.5% had osteopenic changes. These finding affirm the importance of a hip surveillance program to reduce functional limitations and avert secondary complications through early detection and intervention for hip dysplasia among children with CP.

Chapter 2 will describe another common associated impairment, epilepsy, among children with CP in Bangladesh.

CHAPTER 2. IMPROVING EPILEPSY CONTROL AMONG CHILDREN WITH CEREBRAL PALSY IN RURAL BANGLADESH





2.1 INTRODUCTION

The original BCPR study findings showed that 23.4% of children with CP had epilepsy.(33) Given the frequency of the epilepsy in this population, in this chapter we sought to further explore this common comorbidity. We identified the prevalence and seizure subtypes in children with CP in Bangladesh. Furthermore, we identified the barriers to optimal epilepsy control in this group of children. This study was a joint collaborative effort between CSF Global, the University of Sydney and Cerebral Palsy Alliance.

2.2 PAPER #2. IMPROVING EPILEPSY CONTROL AMONG CHILDREN WITH CEREBRAL PALSY IN RURAL BANGLADESH

BMJ Open

BMJ Open

Improving epilepsy control among children with cerebral palsy in rural Bangladesh: A prospective cohort-based study

Journal:	BMJ Open
Manuscript ID	Draft
Article Type:	Original research
Date Submitted by the Author:	n/a
Complete List of Authors:	Karim, Tasneem; The University of Sydney, Discipline of Child and Adolescent Health, Children's Hospital at Westmead Clinical School; CSF Global Das, Manik Chandra; CSF Global; University of South Asia, Asian Institute of Disability and Development Muhit, Mohammad; CSF Global; University of South Asia, Asian Institute of Disability and Development Badawi, Nadia; The University of Sydney, Discipline of Child and Adolescent Health, Children's Hospital at Westmead Clinical School; The University of Sydney, Cerebral Palsy Alliance Research Institute, Specialty of Child & Adolescent Health, Sydney Medical School, Faculty of Medicine & Health Khandaker, Gulam; The University of Sydney, Discipline of Child and Adolescent Health, Children's Hospital at Westmead Clinical School; Central Queensland Hospital and Health Service Mohammad, Shekeeb S; The University of Sydney, Discipline of Child and Adolescent Health, Children's Hospital at Westmead Clinical School
Keywords:	Epilepsy < NEUROLOGY, Developmental neurology & neurodisability < PAEDIATRICS, PUBLIC HEALTH
	•

SCHOLARONE[™] Manuscripts



I, the Submitting Author has the right to grant and does grant on behalf of all authors of the Work (as defined in the below author licence), an exclusive licence and/or a non-exclusive licence for contributions from authors who are: i) UK Crown employees; ii) where BMJ has agreed a CC-BY licence shall apply, and/or iii) in accordance with the terms applicable for US Federal Government officers or employees acting as part of their official duties; on a worldwide, perpetual, irrevocable, royalty-free basis to BMJ Publishing Group Ltd ("BMJ") its licensees and where the relevant Journal is co-owned by BMJ to the co-owners of the Journal, to publish the Work in this journal and any other BMJ products and to exploit all rights, as set out in our <u>licence</u>.

The Submitting Author accepts and understands that any supply made under these terms is made by BMJ to the Submitting Author unless you are acting as an employee on behalf of your employer or a postgraduate student of an affiliated institution which is paying any applicable article publishing charge ("APC") for Open Access articles. Where the Submitting Author wishes to make the Work available on an Open Access basis (and intends to pay the relevant APC), the terms of reuse of such Open Access shall be governed by a Creative Commons licence – details of these licences and which <u>Creative Commons</u> licence will apply to this Work are set out in our licence referred to above.

Other than as permitted in any relevant BMJ Author's Self Archiving Policies, I confirm this Work has not been accepted for publication elsewhere, is not being considered for publication elsewhere and does not duplicate material already published. I confirm all authors consent to publication of this Work and authorise the granting of this licence.

reliez oni

For peer review only - http://bmjopen.bmj.com/site/about/guidelines.xhtml

Improving epilepsy control among children with cerebral palsy in rural Bangladesh: A prospective cohort-based study

Tasneem KARIM^{1,2,3*}, Manik Chandra DAS^{1,2}, Mohammad MUHIT^{1,2}, Nadia BADAWI^{3,4}, Gulam KHANDAKER^{1,2,3,5}, Shekeeb S MOHAMMAD³

¹CSF Global, Dhaka, Bangladesh

²Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka, Bangladesh ³Discipline of Child and Adolescent Health, Sydney Medical School, University of Sydney, Sydney, Australia

⁴Cerebral Palsy Alliance Research Institute, University of Sydney, Sydney, Australia

⁵Central Queensland Public Health Unit, Central Queensland Hospital and Health Service, Rockhampton, Queensland, Australia

Word count: 2672, 3 figures, 1 table, 24 references

Corresponding Author

Dr Shekeeb S Mohammad The University of Sydney Children's Hospital Westmead Clinical School, Locked Bag 4001, NSW 2145 Australia Tel: 00 61 298 450 182 Email: <u>shekeeb.mohammad@sydney.edu.au</u>

ABSTRACT

Objective: To define the prevalence and seizure subtypes among children with cerebral palsy (CP) in rural Bangladesh and explore barriers to optimum epilepsy control.

Design: Prospective cohort study

Setting: The study was conducted in Shahjadpur, a rural subdistrict of Bangladesh.

Participants: Children (<18 years) with CP and epilepsy identified using the Bangladesh CP Register in the study site.

Methods: Assessments were conducted in three focused epilepsy clinics overseen by a pediatric neurologist between December 2016 - January 2018, with intervening phone and video-conference follow-ups. Details of event type, frequency and medication compliance were collected. Antiepileptic drugs (AED) were prescribed based on seizure type, family income, comorbidity and medication availability.

Results: 23.4% (170/726) of the BCPR cohort had a clinical diagnosis of epilepsy of whom 166 were assessed. Following the focused epilepsy clinics, 62.0% (103/166) children were clinically determined to have ongoing epileptic seizures. 62.1% (64/103) had generalized tonic clonic seizures, 27.2% (28/103) had focal seizures with altered awareness and 10.7% (11/103) had other seizure types. None of the children with prolonged seizures (31/103) had an emergency seizure management plan. Non-epileptic events were being pharmacologically treated as seizures in 18.1% (30/166) children. Financial constraints were the main reason for non-compliance on follow up.

Conclusions: Gaps in optimum epilepsy management in rural Bangladesh are amenable to improvement anchored with local health care workers. Training and clinical care focused on recognition of common seizure types, seizure mimics and rationalizing use of available AEDs can be facilitated by better referral pathways and telehealth support.

Key Words: Epilepsy, cerebral palsy, Bangladesh, CP

Strengths and limitations of this study

- Children with CP and epilepsy identified through an ongoing population-based surveillance.
- Specialist clinical assessments were conducted overseen by a pediatric neurologist.
- Phone and video-conference follow-ups were conducted.
- The study provided opportunity for continuing local capacity building.
- The clinical diagnoses relied on clinical impression and were not corroborated by investigations.

for oper teries only

INTRODUCTION

Cerebral palsy (CP) is a term that defines a heterogenous group of early-onset, non-progressive, neurodevelopmental disorders secondary to injury to the developing brain [1]. Studies show that epilepsy is associated with greater impairment of cognitive function, poorer motor outcomes, more profound behavioral and psychological problems, and poorer quality of life among children with CP, all of which collectively contribute to a greater burden of disability and care [2]. Children with CP and epilepsy tend to have early onset of seizures which can often be difficult to control [3].

Recent estimates from a population-based study in Bangladesh showed a high burden of CP with an estimated prevalence of 3.4 per 1000 children [4]. Bangladesh is one of the most densely populated and under resourced countries in the world [5]. The World Health Organization (WHO) classified Bangladesh as one of the countries with severe shortages of health workers. There is inequity in the skill mix and distribution of health workers between urban and rural Bangladesh [6]. One of the four axes of the value-based framework for global health delivery highlights the need for alignment of care delivery to the local context [7].

Resources for the diagnosis and management of neurologic disorders such as epilepsy are often limited in low and middle-income countries (LMICs) such as Bangladesh [8]. Several aspects of epilepsy management that may be considered routine in tertiary or specialist settings are not applicable to community-based settings [9]. There is a substantial epilepsy treatment gap in low resource settings owing to a wide spectrum of factors including shortage of doctors particularly in the rural areas,[6] lack of available investigation and inpatient treatment facilities as well as decreased service utilization due to the stigma around a disability diagnosis.[10]

We aimed to define the prevalence, clinical phenotypes and barriers to optimum epilepsy control among children with CP in a community-based setting in Bangladesh.

METHODS AND ANALYSIS

Cohort compilation

We used the Bangladesh CP Register (BCPR); a prospective population-based surveillance of children with CP in Shahjadpur a northern subdistrict of Rajshahi division in Bangladesh for identification of children with CP and epilepsy. Detailed account of the BCPR study protocol and findings have been described in previous publications [4]. During previous BCPR camps, a diagnosis of epilepsy had been

based on history of one or more unprovoked seizures in the previous 3 months recorded by medical practitioners and review of any available medical records [4].

Clinical assessment of epilepsy

Children with CP and epilepsy identified via the BCPR were clinically reviewed in three focused epilepsy clinics held for three days each time at three different locations within the BCPR study site. Specialist clinical assessments at the clinics were overseen by a pediatric neurologist from Australia who travelled to Bangladesh for the focused epilepsy clinics during the study period. Diagnoses of epilepsy and seizure like events were reviewed during assessment in the clinics. Details of seizure/event type, frequency, medication use and compliance were collected according to a predesigned standard proforma (Appendix A). Workflow during the clinic is outlined in Figure 1.

Local capacity building

Two physicians were trained by the pediatric neurologist in classifying seizure types according to the 2017 International League Against Epilepsy (ILAE) guidelines [11], demonstration of clinical signs during the epilepsy clinics, and discussions around seizure mimics and drug choice (Appendix B). One community worker based in the study area was also trained to conduct phone follow ups of the children on antiepileptic drugs (AED).

Selection of antiepileptic drugs

Before the clinics, the community worker collected information on availability and cost of AED in local pharmacies within the study area between December 2016 and January 2018. A dose equivalence table was drawn up for easy prescription in the clinic along with notes on important side effects and interactions. During the clinics, AED were prescribed based on seizure type, medication availability and family income. The approach undertaken for shared decision making in AED prescription is outlined in Figure 2.

Telehealth supported follow up and clinics

Phone follow up

Targeted phone follow ups of the children on AED were conducted by the trained community health worker every three months during the study period, following the initial specialist assessment at the focused epilepsy clinics. The phone follow ups were semi-structured. The design, conduct and the outcome measures for the follow ups were additionally informed by the study team's experience

BMJ Open

and input from the primary caregivers (Appendix C). Seizure control was documented during phone follow ups. In our study seizure control was defined as no reported seizure since the last follow up.

Telemedicine clinics

Telemedicine clinics were initiated in May 2018 and held on a monthly basis using Skype as part of ongoing capacity building to improve epilepsy control among the study cohort. The local trained physician saw the patients face to face in the study site and used a handheld, internet connected tablet to videoconference with the pediatric neurologist in Australia. Patient interview for new and follow up patients followed a set format (Appendix C). New patient data from the telemedicine clinics are not included in this paper.

Patient and Public Involvement

This work was informed by the priorities, experience and preferences of the primary caregivers of the children with CP and epilepsy who participated in the study. The design and implementation of the follow ups, including outcome measures important to the study participants, relied on feedback from families of children with CP and epilepsy. Baseline information was communicated to the primary caregivers by the study team. This informed shared decision making related to the treatment and follow up for their children. Furthermore, the follow ups and telemedicine clinics were conducted by a local community worker and local physician, which enhanced community involvement during and beyond the study.

Statistical analysis

Descriptive analyses were carried out. All statistical analysis was conducted using SPSS version 24 (IBM Armonk, NY, USA).

Ethics

This study was conducted as part of the Bangladesh Cerebral Palsy Register Study which has been approved by the Bangladesh Medical Research Council (BMRC) Human Research Ethics Committee (Ref no. BMRC/NREC/2013–2016/1267) in Bangladesh, and by the Cerebral Palsy Alliance NHRMC Human Research Ethics Committee (Ref no.2015–03-02) in Australia. Written informed consent was taken from the primary caregiver/parents/guardian of the children with CP.

RESULTS

Prevalence and basic demographic characteristics

726 children with CP were registered into the BCPR between January 2015 and December 2016, 23.4% (170/726) of whom had a clinical diagnosis of epilepsy. 166 of these children attended the three focused epilepsy clinics between December 2016 and January 2018 and form the study cohort. 55 (33.1%) were female. The mean age of the children was 6 years 10 months (SD: 4 years 5 months) years.

After the focused epilepsy clinics, 62.0% (103/166) children were clinically determined to have ongoing epileptic seizures based on review of their history, existing medical records and specialist clinical evaluation (Figure 3). Therefore, the revised prevalence of epilepsy among the BCPR cohort during the study period was 14.3%.

Seizure subtypes

62.1% (64/103) had generalized tonic clonic seizures (GTCS), 27.2% (28/103) had focal seizures with altered awareness and 5.8% (6/103) had other seizure types (focal seizures with preserved awareness, epileptic spasms, myoclonic seizures and tonic seizures). Data on seizure type was unclear on history for 4.9% (5/103). At the time of first assessment, seizures were already controlled with AED in 5.8% (6/103) children. 30.1% (31/103) of children had a history of prolonged seizures (>30 minutes) and none of these patients had an emergency seizure plan. Their caregivers tended to wait at home till the seizures settled and did not seek emergency medical assistance due to geographical or financial constraints.

Barriers to optimum epilepsy control

Non-epileptic events among children with CP

Non-epileptic events were determined to have been mislabeled as seizures in 18.1% (30/166) children which included extremity clonus (n=7), dystonic postures (n=6), spasticity related spasms (n=4), breath holding spells (n=3), mannerisms (n=3), sleep related myoclonic jerks (n=2), startles (n=2), stereotypies (n=2) and rhythmic movement disorders in sleep (n=1). 23.3% (7/30) of these children were being treated with AED.

Epilepsy control

Of the 103 children with seizures, 62 were already on AEDs at the time of our clinical review. Polypharmacy with more than two concurrent AED was commonly observed and AED changes were made for the majority of them. Advised AED changes consisted of dose alteration in 54.8% (34/62) and medication change in 17.7% (11/62). 27.4% (17/62) were advised to continue treatment already

 initiated by various providers. We initiated treatment for 39/41 children not previously on AED who were clinically determined to still be having epileptic seizures; 2/41 only had short seizures once or twice a year and were not put on AED.

Telehealth supported follow up and clinics

Phone follow up

We were able to review 75.8% (78/90) children with epileptic seizures on follow up during the study period. On follow-up (median 6.0 months), 69.2% (54/78) were taking prescribed medications as advised. Among them 75.9% (41/54) showed improvement in seizure control (>50% seizure reduction), including 14 children who became seizure free. 30.8% (24/78) families had discontinued the advised treatment due to affordability (8/24, 33.3%), excessive drowsiness (7/24, 29.2%), development of a rash (4/24, 16.7%), no perceived benefit with medication and lack of understanding behind the use of regular medications (2/24, 8.3%). Three (3/24, 12.5%) children who discontinued medications were reported to be seizure free. None of the families reported any adverse effects that led to reported cardiorespiratory compromise, hospital presentation or death.

Two children from our cohort died during the follow up period, one due to meningitis and the other due to a lower respiratory tract infection. Their cause of death was determined by verbal autopsy conducted as part of a separate study [12].

Telemedicine

Five telemedicine clinics undertaken in 2018 contributed to patient follow up and clinical capacity building. During these clinics 47 patients were seen by a local medical practitioner with internetbased videoconference support from the pediatric neurologist in Australia. Each clinic was of three hours duration during which patient interview was undertaken in the same manner as in the focused epilepsy clinics. Thirty minutes were marked during each clinic for discussion regarding clinical signs, history taking and AED choice. Clinical details for new patients reviewed during telemedicine clinics were not included in this cohort.

DISCUSSION

Epilepsy is a significant comorbidity in some individuals with CP. Previous studies have described a prevalence of 15-90% epilepsy in CP cohorts [13,14]. Overall, epilepsy contributes more significantly to the global burden of disease in resource poor settings as evident from the 2015 Global burden of disease studies. We found an initial prevalence of epilepsy of 23.4% in our cohort. Interestingly,

BMJ Open

 following reassessment in our clinics, as described, this was revised to 14.3%. Previous studies have also noted such discrepancy between determination of a clinical diagnosis of epilepsy between specialist and community-based settings with a misdiagnosis of epilepsy being made in as many as 25% of cases [15]. This has flow on impacts as we noted in terms of incorrect, often excessive use of medications. Epilepsy poses substantial economic burden on families [16]. When families devote a significant proportion of their finances, attention, time or all of these towards one aspect of their child's management, other aspects of care such as physical therapy, nutrition, pain and musculoskeletal management are likely to be neglected, more so in resource poor settings [17].

As demonstrated by recent innovative projects in neighbouring Nepal, education of community level workers and general medical practitioners can lead to more consistent clinical diagnosis of epilepsy [18]. In our experience, rationalisation or cessation of medications after focused clinical assessments led to changes in family finances diverted towards medication use. We envision that the development of simplified print and multimedia based educational resources for health care workers and medical practitioners hold the potential to improve epilepsy diagnosis in resource scarce settings such as our study site.

Polypharmacy with more than two concurrent AEDs is unlikely to contribute significantly to seizure control [19]. In countries like Bangladesh with a mismatch of clinical care practices between urban and rural areas, the use of less conventional or alternative medications is very likely to be encountered.

AED availability is very limited in rural Bangladesh [20]. Medications need to be purchased by families and hence, cost per month for AEDs is a significant consideration when choosing medications for chronic use to ensure good compliance. The cheapest and most readily available AEDs are phenobarbitone, clobazam and sodium valproate. If a diagnosis of CP is very likely based on clinical evaluation and history, earlier use of sodium valproate or clobazam in this setting is a viable option for transitioning from phenobarbitone which is most commonly prescribed in infancy. As outlined in our methods and Figure 2, AED choice can be rationalized based not only on the seizure type but also existing comorbidity as some AEDs can help improve comorbid psychiatric symptoms or sleep disturbance.

Our experience highlighted a gap in the recognition and management of prolonged seizures in settings like ours compared to conventional management in urban and resource rich settings.

BMJ Open

Benzodiazepines are the mainstay of out of hospital, particularly health worker or parent led management of prolonged seizures. However, midazolam or lorazepam are not available at all in rural Bangladesh. Diazepam is only available in glass ampoules through restricted prescriptions in some pharmacy outlets. In our and wider reported experience caregivers are often reluctant to use glass ampoules or follow several steps in medication administration to a child at home [21]. Other readily available benzodiazepines are cheap (clobazam: 0.042 USD per 10 mg tablet and clonazepam: 0.048 USD per 1 mg tablet; prices mid 2018) but there is very little evidence regarding their use in the setting of prolonged seizures [22,23]. Status epilepticus can significantly add to the burden of cumulative brain injury and therefore warrants a solution [24]. This may be in the form of a perrectal, oral or, alternative routes for delivery of well-established medications for status epilepticus such as phenobarbitone, valproate or midazolam. Alternatively, the use of medications such as clonazepam drops via open label trials requires urgent exploration for such settings.

Our model has demonstrated that immediate positive impact on epilepsy management and reduction in burden of care on families can be achieved through structured assessments by medical and allied personnel who are trained to assess children for epilepsy and use available medications according to a structured framework. This can be achieved for a population base such as in our study area with limited personnel and without additional investigation or formalized health care facilities, though these would be desirable to further improve patient outcomes. We piloted the use of videoconference-based telemedicine clinics after initial face to face clinics. With some prior training in the use of a structured clinical approach, this method can be very time-efficient in reviewing patients led by a non-specialist medical practitioner/community worker and supported by a specialist. In our experience, this not only provided continuity of clinical support with existing personnel but also provided an opportunity for continuing professional development and capacity building. We hope that in the post-COVID era, implementation and incorporation of telemedicine should be easier and more acceptable to providers, policymakers and the community.

We summarize the key barriers identified and proposed or already implemented solutions in Table 1. Development of multimedia or mobile application-based resources that may simply illustrate clinical assessment of children with epilepsy, examples of non-epileptic events and emergency seizure management will provide convenient means for translation of our findings to the wider population in Bangladesh and, with language translation, to similar resource poor settings across the world. We have engaged with tertiary paediatric neurology centres in Bangladesh to support some families with requisite investigations or more frequent specialist review. However, this will always be limited to financial and logistic constraints of rural families.

Study limitations

We did not systematically collect baseline investigation information for this cohort as a small proportion had any previous tests such as electroencephalography (EEG) or neuroimaging. The clinical diagnosis of seizures and non-epileptic events were not corroborated by investigations as they were unavailable in this resource limited setting. We had to rely on the clinical impression of a limited number of observers. Although we utilized standard criteria to assess seizure reduction, the collection of the follow up data was based on reporting by the primary caregiver which may have been a source of potential bias.

CONCLUSION

Epilepsy is prevalent among children with CP in rural Bangladesh and the various gaps in optimum epilepsy management are lack regular follow-up, recognition of common seizure types and nonepileptic seizure mimics, familiarization with commonly available, affordable AED and availability of guidelines for prolonged seizure management. These gaps are amenable to proposed low cost, educational interventions. Health care workers can improve epilepsy management with regular follow-up, education on common seizure types, seizure mimics, use of commonly available, affordable AED and guidelines for prolonged seizure management.

AUTHORS' CONTRIBUTION

All listed authors meet the appropriate authorship criteria, and nobody who qualifies for authorship has been omitted. GK and SM conceptualized and established this research study. They also contributed to study design, development of the study materials and overall conduct of the study supported by TK. SM, TK and MCD were responsible for assessment of study participants and data collection. SM, GK and MM provided specialist advice in this study. TK, SM and GK completed data analysis, interpretation of the data and drafted the initial and revised manuscript with input from all the co-authors. All authors have read and approved the final manuscript.

FUNDING STATEMENT

This study has been conducted as part of the BCPR study funded by the Cerebral Palsy Alliance Research Foundation (PG4314) and through internal funding from CSF Global, Bangladesh. TK is

supported by the Cerebral Palsy Alliance Research Foundation (CDG04617, PHD02119). SM is supported by Cerebral Palsy Alliance Research Foundation Career Development Grant (CDG7916). The study funders played no role in the design of the study and collection, analysis, interpretation of data and in the preparation of the manuscript, and in the decision to submit the paper for publication.

ACKNOWLEDGEMENTS

We would like to acknowledge the CSF Global team in Bangladesh for their cordial support in implementing this project and supporting the families of children with CP in referrals and access to services. We also want to acknowledge the primary caregivers of the children with CP and epilepsy who participated in the study. Their input was invaluable to the design and conduct of this study.

CONFLICT OF INTEREST DISCLOSURES

The authors declare no competing interests. erezien onz

DATA AVAILABILITY STATEMENT

No additional data are available.

REFERENCES

 [1] Bax M, Goldstein M, Rosenbaum P, Leviton A, Paneth N, Dan B, et al. Proposed definition and classification of cerebral palsy, April 2005. Developmental Medicine & Child Neurology. 2005;47(8):571-6.

[2] Mesraoua B, Ali M, Deleu D, Al Hail H, Melikyan G, Haddad N, Alalamy O, Athanasios C, Asadi-Pooya AA. Epilepsy and Cerebral Palsy. Neurodevelopment and Neurodevelopmental Disorder 2019 May 8. IntechOpen.

[3] Singhi P, Jagirdar S, Khandelwal N, Malhi P. Epilepsy in children with cerebral palsy. Journal of Child Neurology. 2003;18(3):174-9.

[4] Khandaker G, Muhit M, Karim T, Smithers-Sheedy H, Novak I, Jones C, et al. Epidemiology of cerebral palsy in Bangladesh: a population-based surveillance study. Developmental Medicine & Child Neurology. 2019 May;61(5):601-9.

[5] National Institute of Population Research and Training (NIPORT), ICF International. Bangladesh Demographic and Health Survey. Dhaka, Bangladesh; 2014.

[6] Ahmed SM, Hossain MA, RajaChowdhury AM, Bhuiya AU. The health workforce crisis in Bangladesh: shortage, inappropriate skill-mix and inequitable distribution. Human Resources for Health. 2011;9(1):3.

[7] Kim JY, Farmer P, Porter ME. Redefining global health-care delivery. The Lancet. 2013;382(9897):1060-9.

[8] Bearden DR. Pediatric Neurology in Resource-Limited Settings: a Systematic Review. Current Pediatrics Reports. 2018;6(1):34-9.

[9] Kim JY FP, Porter ME. A global health delivery framework approach to epilepsy care in resource-limited settings. Journal of the Neurological Sciences. 2015;358(1-2):263-5.

[10] Patel P, Baier J, Baranov E, Khurana E, Gambrah-Sampaney C, Johnson A, et al. Health beliefs regarding pediatric cerebral palsy among caregivers in Botswana: A qualitative study. Child: Care, Health and Development. 2017;43(6):861-8.

[11] Fisher RS, Cross JH, French JA, et al. Operational classification of seizure types by the International League Against Epilepsy: Position Paper of the ILAE Commission for Classification and Terminology. Epilepsia 2017;58:522–530

[12] Jahan I, Karim T, Das MC, Muhit M, Mcintyre S, Smithers-Sheedy H, Badawi N, Khandaker G.
 Mortality in children with cerebral palsy in rural Bangladesh: a population-based surveillance study.
 Developmental Medicine & Child Neurology. 2019 Nov;61(11):1336-43.

[13] Aicardi J. Epilepsy in Brain-Injured Children. Developmental Medicine & Child Neurology. 2010;32(3):191-202.

BMJ Open

[14] Aksu F. Nature and Prognosis of Seizures In Patients With Cerebral Palsy. Developmental Medicine & Child Neurology. 2008;32(8):661-8.

[15] National Clinical Guideline Centre. The epilepsies: the diagnosis and management of the epilepsies in adults and children in primary and secondary care.

http://www.nice.org.uk/nicemedia/live/13635/57784/57784.pdf (accessed

March 2021)

[16] Allers K, Essue BM, Hackett ML, Muhunthan J, Anderson CS, Pickles K, Scheibe F, Jan S. The economic impact of epilepsy: a systematic review. BMC neurology. 2015 Dec 1;15(1):245.

[17] Nuri RP, Aldersey HM, Ghahari S. Needs of families of children with cerebral palsy in Bangladesh: A qualitative study. Child: Care, Health and Development. 2018;45(1):36-44.

[18] Rajbhandari H, Joshi S, Malakar S, Paudel P, Jain P, Uppadaya K, et al. Epilepsy field workers, a smartphone application and telephone telemedicine: Safe and effective epilepsy care in rural Nepal. Seizure. 2019;64:54-8.

[19] Patsalos PN, Perucca E. Clinically important drug interactions in epilepsy: general features and interactions between antiepileptic drugs. The Lancet Neurology. 2003;2(6):347-56.

[20] Hussain ME. Burden of Epilepsy in Bangladesh: Current Approach. Journal of Current and Advance Medical Research. 2018;4(1):1-2.

[21] Chiang LM, Wang HS, Shen HH, Deng ST, Tseng CH, Chen YI, Chou ML, Hung PC, Lin KL. Rectal diazepam solution is as good as rectal administration of intravenous diazepam in the first-aid cessation of seizures in children with intractable epilepsy. Pediatrics & Neonatology. 2011 Feb 1;52(1):30-3.

[22] Shangguan Y, Liao H, Wang X. Clonazepam in the treatment of status epilepticus. Expert Review of Neurotherapeutics. 2015;15(7):733-40.

[23] Treiman DM, Meyers PD, Walton NY, Collins JF, Colling C, Rowan AJ, et al. A Comparison of Four Treatments for Generalized Convulsive Status Epilepticus. New England Journal of Medicine. 1998;339(12):792-8.

[24] Kwong KL, Wong SN, So KT. Epilepsy in children with cerebral palsy. Pediatric Neurology. 1998;19(1):31-6.

Table 1: Barriers to epilepsy control and suggested interventions

Barriers	Suggested interventions
AED availability	Selection of locally available medications for management through a structured guideline
Lack of skilled personnel for	Capacity building and engagement of local medical practitioners and
epilepsy management and	community health workers
follow up locally	Development of multimedia or mobile application-based resources Telemedicine
Affordability	Rationalization of drugs
Poor treatment compliance	Rationalization of drugs
	Training and engagement of health workers for follow up
	Caregiver education
Prolonged seizure management	Development of guideline and resources for management of prolonged
	seizure for training of local health workers
Misidentification of non-	Development of video resources describing seizures and non-epileptic
epileptic episodes as seizures	events
Lack of parental understanding	Parent education on epilepsy treatment
regarding epilepsy treatment	

BMJ Open

Figure 1: Clinical assessment of epilepsy in children with CP in Shahjadpur

Bangladesh Cerebral Palsy Register

Population-based surveillance of children with cerebral palsy in Shahjadpur

Dates for epilepsy clinics decided

Children with CP having epilepsy identified from the BCPR cohort

Families of the children informed of the clinic date and location by phone by community worker

Epilepsy clinics held at 3 sites in Shahjadpur

Structured proforma populated and anthropometric measurement taken by community workers

10 minutes per patient

- Clinical review by local medical practitioners overseen by pediatric neurologist from Australia and review of relevant medical records
 15 minutes per patient with interpretation
- Medications explained by community worker 5 – 10 minutes per patient

For peer review only - http://bmjopen.bmj.com/site/about/guidelines.xhtml

BMJ Open







ASSESSMENT DETAILS			
ID NUMBER:	SER: SERIAL NUMBER:		JMBER:
ASSESSMENT LOCATION:		ASSESSME	ENT DATE: DD / MM / YYYY
CHILD'S DETAILS		l	
NAME:	GENDER:	M 🗆 F	DOB: DD / MM / YYYY
HEAD CIRCUMFERENCE (cm):	WEIGHT (kg): LENGTH/HEIGHT (cm		LENGTH/HEIGHT (cm):
FATHER'S DETAILS			
NAME:	DOB: DD / MI	VI / YYYY	OCCUPATION:
EDUCATION			I
□ Illiterate □ Primary □ Secondary □ Higher seconda	iry 🛛 Graduatio	n 🗌 Post-gra	duation 🛛 Diploma/other trade qualification
MOTHER'S DETAILS			
NAME:	DOB: DD / MI	M / YYYY	OCCUPATION:
EDUCATION Illiterate Primary Secondary Higher secondary qualification	ary 🛛 Graduatio	n 🗌 Post-gra	aduation 🛛 Diploma/other trade
CONTACT DETAILS			
DISTRICT:	SUB-D	ISTRICT:	
UNION:	VILLAGE:		
POST CODE:	PHON	E NO.:	
TYPE OF CASE (select one): New Follow	v-Up		
SEIZURE CONTROL: Same Better Worse	SEIZURE FF	REE ON TREA	ATMENT: 🗆 Yes 🛛 No
COMPLIANT: 🗆 Yes 🗆 🛛 IF NO, REASON FOR N	NON-COMPLIA	ANCE:	
No			
REASON FOR POOR SEIZURE CONTROL:			
MAIN CONCERN:		0	
HISTORY AND EXAMINATION FINDINGS			
BIRTH HISTORY:			
SEIZURE			
FIRST:	LAST:		

MAIN CONCERN:	
HISTORY AND EXAMINATION FINDINGS	
BIRTH HISTORY:	
SEIZURE	
FIRST:	LAST:
HISTORY OF PROLONGED SEIZURE (> 5 mins):	es 🗆 No
CURRENT FREQUENCY: times per 🗆 day	🗆 week 🛛 month 🖓 year
ТҮРЕ:	
DURATION OF SEIZURE:	DESCRIPTION:
PREVIOUS MEDICATION & INVESTIGATION	

1		
2		
3	Appendix B: Guideline for drug choice at clinics	
4		
5	Table of Contents	
0 7		
7 8		
9	Phonobarbitana	л
10	Prenobarbitone	•4 1
11	Dose range	.4
12	<u>Seizure types useful for</u>	.4
13	Adverse effects	.4
14	Comment	.4
15	Sodium Valproate	.4
16	Dose Range	.4
17	Seizure types useful for	.4
18	Adverse effects	. 4
19	Comment	.4
20	<u>Clobazam</u>	. 5
21	Dose Range	. 5
22	Seizure types useful for	. 5
23	Adverse effects	. 5
25	Comment	. 5
26	Nitrazepam.	.5
27	Dose Bange	5
28	Seizure types useful for	5
29	Adverse effects	5
30	Commont	د. د
31	Carbamazonina	۰۰. د
32	Carbamazepine	.0
33	Dose Range	. 0
34	Seizure types useful for	. 6
35	Adverse effects	. 6
30 27	<u>Comment</u>	. 6
20	MEDICATIONS FOR DYSTONIA/SPASTICITY MANAGEMENT	
30	Tribexinhenidyl/Benzbexol	. 6
40	Dose Range	6
41	Symptoms useful for	.0 6
42	Adverse effects	.0 6
43	Adverse effects	.0
44	Comment	. /
45	Bacioten	./
46	Dose Range	. /
47	Start Symptoms useful for	.7
48	Adverse effects	.7
49	Comment	.7
50	Levodopa/Carbidopa	.7
51	Dose Range	.7
52	Symptoms useful for	.7
55 57	Adverse effects	. 8
54 55	Comment	. 8
56		
57	WIEDICATIONS FOR BEHAVIOUR MANAGEMENT	_
58	Clonidine	. 8
59	Dose Range	. 8
60	Start Symptoms useful for	. 8

Adverse effects	
Comment	
Risperidone	
Dose Range	
Symptoms useful for	
Adverse effects	
Comment	
MEDICATIONS FOR EMERGENCY MANAGEMENT	9
Diazepam	
Dose Range	
Symptoms useful for	
Adverse effects	
<u>Comment</u>	
REFERENCES	9

8
9
10
11
12
13
14
15
16
17
18
19
20
21
22
23
24
25
26
27
28
29
30
31
32
33
34
35
36
37
38
39
40
41
42
43
44

MEDICATIONS FOR SEIZURE MANAGEMENT

PHENOBARBITONE

Dose range

• 1–6 mg/kg daily in 1 or 2 doses; start at the lower end of range and increase slowly if required.

Seizure types useful for

- For generalized seizures and focal seizures.
- Recurrent febrile seizures
- Neonatal seizures
- Less likely to help blank staring "absence" seizures

Adverse effects

- Allergic rash, rarely can cause extreme hypersensitivity reaction
- Hyperactivity
- Sedation

Comment

- Generally, a good medication to continue unless it is not working or the side effects are not tolerable
- Do not stop suddenly and wean over several weeks if patient has been on it for more than 3 months

SODIUM VALPROATE

Dose Range

• 20–40 mg/kg daily in 2 divided doses; start at 5-10 mg/kg/day and increase slowly to 20 mg/kg/day. Then increase further if required.

Seizure types useful for

• For all seizure types

Adverse effects

- Hyperactivity
- Liver dysfunction
- Caution with use in children with developmental delay less than two years or older who are likely to not have true CP. Valproate can cause fatal liver dysfunction in those with a history of mitochondrial disorders. This is suggested by a history of developmental regression eg. A child who was able to sit before can no longer do so.

Comment

- Valproate is likely to work for most seizure types and cause less sedation than phenobarbitone
- Valproate works synergistically with clobazam and lamotrigine.
- Lamotrigine should be introduced very cautiously if someone is already on valproate

CLOBAZAM

Dose Range

- Start at 1-2 mg/dose once a day. Can increase as required to
 - 2.5-5 mg BD in children <2 years
 - o 5-10 mg BD in children 2-10 years
 - o 10 mg TDS in older children
 - o maintenance 0.3-1 mg/kg/day in 2 divided doses
- The above doses are a guide and higher doses can be used in younger children if tolerated and if thought to be beneficial for seizure control

Seizure types useful for

- For all seizure types
- Also helpful for dystonia management in some cases

Adverse effects

- Sedation
- Hyperactivity
- Hallucinations
- Drooling

Comment

- Useful monotherapy or add on medication
- Start at small doses like 1 mg BD and grade up
- Can crush tablet and suspend in water to make up small doses if liquid not available
- If difficult for family to understand use quarter tablet / half tablet instead of dissolving
- Wean very slowly similar to phenobarbitone if patient has been taking Clobazam for more than 3 months
- Children can sometimes get used to benzodiazepines. If seizures break through after a few months of good control, then consider swapping over to another benzodiazepine like Nitrazepam.

NITRAZEPAM

Dose Range

- Start at 1 month 2 years: 0.25mg/kg twice daily, up to 0.5mg/kg twice daily
- The above doses are a guide and higher doses can be used in younger children if tolerated and if thought to be beneficial for seizure control

Seizure types useful for

- For all seizure types
- Infantile spasms

Adverse effects

- Sedation
 - Drooling

- Useful monotherapy or add on medication
- Start at small doses like and grade up
- Wean very slowly similar to phenobarbitone if patient has been taking Nitrazepam for more than 3 months
- Children can sometimes get used to benzodiazepines. If seizures break through after a few months of good control, then consider swapping over to another benzodiazepine like Clobazam.

CARBAMAZEPINE

Dose Range

 10–20 mg/kg daily in 2-3 divided doses; start at 2.5-5 mg/kg/day and increase slowly to 10 mg/kg/day. Then increase further if required. Some patients can respond to low doses 5-10 mg/kg/day and can be maintained on these doses without further increasing unless required

Seizure types useful for

- Focal seizures only
- Avoid for generalized, absence and febrile seizures

Adverse effects

- Hyperactivity
- Liver dysfunction

Comment

Useful drug for focal seizures, e.g. With hemiplegic CP

MEDICATIONS FOR DYSTONIA/SPASTICITY MANAGEMENT

TRIHEXIPHENIDYL/BENZHEXOL

Dose Range

- Start at 0.25 mg once a day and gradually increase to 0.25 mg tds.
- If tolerated, can trial up to 2 mg 4 mg tds
- In older children, higher doses can be used if benefit is noted

Symptoms useful for

Mainly for dystonia management. Can also help drooling due to its anticholinergic properties

Adverse effects

- Dry mouth, eyes
- Constipation
- Confusion
- Double vision

Irritability

Comment

- About 25-50% efficacy for dystonia management if side effects are not a problem. Some children can benefit remarkably more
- Avoid other anticholinergic medications or if any of the adverse effects are already a clinical problem
- If not benefit after maximum doses for 3-4 weeks, then discontinue as delayed benefit is unlikely to occur.

BACLOFEN

Dose Range

- **2-7 y** -10-40 mg/day divided in three to four doses/day. Start: 2.5-5 mg twice a day, may increase by 5-15 mg/day every 3-4 days, Max: 40 mg/day.
- 8-11 y Dose: 10-60 mg/day divided in three to four doses/day. Start: 2.5-5 mg twice a day, may increase by 5-15 mg/day every 3-4 days; Max: 60 mg/day.
- **12 y and older** Dose: 20-80 mg/day PO divided in three to four doses/day. Start: 5 mg twice a day, may increase by 15 mg/day every 2-3 days; Max: 80 mg/day.

Start Symptoms useful for

Mainly for spasticity management.

Adverse effects

- Hypotonia
- Drooling
- Sedation

Comment

- Baclofen is a good medication for high tone which is due to spasticity
- It is not so good when there is dominant or mixed dystonia
- Relatively high doses may be needed in some patients making side effects intolerable, these have to be balanced with dose
- Other sedative medications will add to sedative effects and drooling benzodiazepines, phenobarbitone
- Taper and stop slowly over few weeks if patient has been taking Baclofen for more than 3 months.

LEVODOPA/CARBIDOPA

Dose Range

- 1-4 mg/kg/day (levodopa component). Start slowly at 1 mg/kg/day divided in 2 doses and increase to target 4 mg/kg/day in 3 divided doses.
- Can increase further if focal but clear benefit.

Symptoms useful for

- Dystonia management
- Can be very helpful when dystonia shows a trend of worsening as the day progresses or is exercise induced
Adverse effects

Nausea

Comment

- About 25% efficacy for dystonia management in cerebral palsy but a safe drug to try
- Very useful in genetic dopamine responsive dystonia which can mimic CP but is rare.
- Some preparations are available as Levodopa/Benserazide. Dose guide is same for levodopa component

MEDICATIONS FOR BEHAVIOUR MANAGEMENT

CLONIDINE

Dose Range

- Start at 25 micrograms at night for sleep management
- Can increase to 25-100 microgram three times a day for behavior management

Start Symptoms useful for

- Management of hyperactive or aggressive behavior. e.g. biting, inattentive in school, disturbs other children, fidgety, can't sit still (these symptoms have to be sufficiently severe to be disruptive to daily home or school life to be considered for treatment)
- Also helpful for episodic management of severe dystonia in patients who get periodic worsening. Doses up to 100 micrograms 4-6 times per day can be helpful for short bursts of 3-4 days. Then wean back to baseline doses or stop

Adverse effects

- Sedation
- Sometimes postural dizziness due to postural hypotension more likely at lower doses

Comment

- Wean slowly over a week if patient has been on clonidine for more than 3 months
- Average efficacy for ADHD, stimulants are better

RISPERIDONE

Dose Range

• 0.25mg – 5 mg/day in children. Try to manage on least efficacious dose

Symptoms useful for

• Management of hyperactive or aggressive behaviour.

Adverse effects

- Increased appetite
- Weight gain
- Metabolic disturbance hyperlipidemia after years of use
- Extrapyramidal effects like rigidity

Comment

- Try clonidine first
- Can be quite useful if behavioral issues are really disruptive for daily life.

MEDICATIONS FOR EMERGENCY MANAGEMENT

DIAZEPAM

- Dose Range
 - 0.5 mg/kg <6 y/o; 0.3mg/kg 6-11 y/o; 0.2mg/kg >11y/o

Symptoms useful for

• Management of prolonged seizures >5 min at home.

Adverse effects

- Sedation
- Respiratory depression and arrest
- Local injury

Comment

- Only prescribe if family have received education on use and understand the administration process
- If a child is having a seizure in which he/she is convulsing or is unconscious, it is important to follow simple first aid measures ie. protection from injury, positioning on their side to assist breathing.
- Materials needed
 - a 25ml bottle of diazepam mixed with a stabilizing solution, containing 1mg of diazepam in each 1ml (or alternative concentration)
 - a reusable 10ml syringe
 - a reusable soft plastic tube to attach to the syringe for drawing up and injecting the diazepam
 - o a sachet of lubricant jelly

REFERENCES

- Australian Medicines Handbook
- MIMS Australia
- Epocrates
- www.rch.org.au
- <u>www.dhs.winsconsin.gov</u>

Appendix C: Follow up questionnaire

Child's Name:				Gender:	\Box Male \Box Female
Weight in kg:	Phone			DOB:	DD/MM/YYYY
	No:				
Mother's Name:		Father's Nan	ne:		
Assessment Location:			Assess	sment Date:	DD/MM/YYYY

Check before phone call/follow up clinic:

Medications and doses the child is on

Was there a change made in the last clinic

Phone Call:

- 1. Is your child taking the prescribed medication regularly?
 - \Box No, is your child taking any other medication? \Box Yes \rightarrow *Fill up table below* \Box No \rightarrow *Go to* 3

Name of Medication	Formulatio n	Dose	Daily Dose Frequency
□ Valproic acid [Valex/Epilim/Epilim/Valpro]	□ Syrup □ Tablet	□ ml/spoon □ mg, tabs	
Phenobarbitone [Barbit/Berdinal/Emer/Epinal/Pheno/Phenoba/Phenoson]	□ Syrup □ Tablet	□ ml/spoon □ mg, tabs	
Epinephrine [Adrinor/Adrenaline/Adrin]	□ Injection □ IV	□ ml □ mg	
□ Benzodiazepine [Clonazepam/Clobazam/Alsium/Clob/Clobam/Epson/ Frisium /Epiclon/Epnil/Leptic/Myotril/Rivotril/Rivo]	□ Syrup □ Tablet	□ ml/spoon □ mg, tabs	
Clonidine [Catapres 0.1/Clonipres 0.1]	□ Syrup □ Tablet	□ ml/spoon □ mg, tab	
Piracetam [Neurolep/Neuratam/Piratam/Juvain/Piramax]	Syrup Tablet	□ ml/spoon □ mg, tabs	
□ Baclofen [Flexifen/Bacofen/Mylofen/Axant/Beclovan]	□ Syrup □ Tablet	□ ml/spoon □ mg, tabs	
Trihexyphenidyl [Hexinor/Trihexy]	□ Syrup □ Tablet	□ ml/spoon □ mg, tabs	
□ Other(<i>write</i>)	□ Syrup □ Tablet	\square ml/spoon \square mg, tabs	

2. Is your child seizure free at current dose?

 \Box Yes \rightarrow End

□ No, Frequency of seizure: _____ per day / week / month / year

Average duration of seizure _____ minutes/hours

For those **not taking any medicine now**:

3. Has the medicine caused any problem? \Box **Yes** \rightarrow *Fill up table below and end.* \Box **No** \rightarrow *Go to* 4.

Problem		Immediate action on the phone				
Extensive rash devel	oped on medication	Stop the medication and need to review urgently				
Child too drowsy to	feed safely on medication	Reduce to older dose/previous medication				
Other Problem:						

4. Specify any other reason for not taking prescribed medication regularly:

STROBE Statement-checklist of items that should be included in reports of observational studies

	Item No	Recommendation	Page No
Title and abstract	1	(a) Indicate the study's design with a commonly used term in the title or	1
		the abstract	
		(b) Provide in the abstract an informative and balanced summary of what	2
		was done and what was found	
Introduction			
Background/rationale	2	Explain the scientific background and rationale for the investigation being reported	4
Objectives	3	State specific objectives, including any prespecified hypotheses	4
Methods			•
Study design	4	Present key elements of study design early in the paper	4-6
Setting	5	Describe the setting, locations, and relevant dates, including periods of	4-6
26		recruitment, exposure, follow-up, and data collection	
Participants	6	(a) Cohort study—Give the eligibility criteria, and the sources and	4-5
1 uniterpunto	0	methods of selection of participants Describe methods of follow-up	
		<i>Case-control study</i> —Give the eligibility criteria and the sources and	
		methods of case ascertainment and control selection. Give the rationale	
		for the choice of cases and controls	
		Cross-sectional study—Give the eligibility criteria and the sources and	
		methods of selection of participants	
		(b) Cohort study. For matched studies, give matching criteria and	NA
		(b) Conort study—1 of matched studies, give matching effectia and	
		Case control study. For matched studies, give matching criteria and the	
		number of controls per case	
Variables	7	Clearly define all outcomes exposures predictors potential confounders	1.6
variables	/	and effect modifiers. Give diagnostic criteria, if applicable	4-0
Data sources/	8*	For each variable of interest, give sources of data and details of methods	4-6
measurement	0	of assessment (measurement). Describe comparability of assessment	4-0
measurement		methode if there is more then one group	
Piec	0	Describe any efforts to address notential sources of bias	1.5
Study size	<u> </u>	Even lain how the study size was arrived at	4-5
Study Size	10	Explain now the study size was arrived at	4-3
Quantitative variables	11	explain now quantitative variables were handled in the analyses. If	0
<u>Q</u> (-)(-)(-)(-)(-)(-)(-)(-)(-)(-)(-)(-)(-)(12	() Describe all statistical matheda including these used to control for	(
Statistical methods	12	(a) Describe all statistical methods, including those used to control for	6
		(b) Describe any methods used to examine subgroups and interactions	NA
		(c) Explain how missing data were addressed	NA
		(d) Cohort study—If applicable, explain how loss to follow-up was	5
		addressed	
		Case-control study—If applicable, explain how matching of cases and	
		controls was addressed	
		Cross-sectional study-If applicable, describe analytical methods taking	
		account of sampling strategy	
		(<u>e</u>) Describe any sensitivity analyses	NA

Continued on next page

Results			
Participants	13*	(a) Report numbers of individuals at each stage of study—eg numbers potentially eligible, examined for eligibility, confirmed eligible, included in the study,	4-
		completing follow-up, and analysed	
		(b) Give reasons for non-participation at each stage	Fi
			3
		(c) Consider use of a flow diagram	Fi
			1
Descriptive	14*	(a) Give characteristics of study participants (eg demographic, clinical, social)	6
data		and information on exposures and potential confounders	
		(b) Indicate number of participants with missing data for each variable of interest	Fi
		(c) <i>Cohort study</i> —Summarise follow-up time (eg. average and total amount)	
Outcome data	15*	<i>Cohort study</i> —Report numbers of outcome events or summary measures over	6-
		time	
		Case-control study—Report numbers in each exposure category, or summary	N
		measures of exposure	
		Cross-sectional study—Report numbers of outcome events or summary measures	N
Main results	16	(a) Give unadjusted estimates and, if applicable, confounder-adjusted estimates	N
		and their precision (eg, 95% confidence interval). Make clear which confounders	
		were adjusted for and why they were included	
		(b) Report category boundaries when continuous variables were categorized	N
		(c) If relevant, consider translating estimates of relative risk into absolute risk for	N
		a meaningful time period	
Other analyses	17	Report other analyses done—eg analyses of subgroups and interactions, and	N
		sensitivity analyses	
Discussion			
Key results	18	Summarise key results with reference to study objectives	8-
Limitations	19	Discuss limitations of the study, taking into account sources of potential bias or	11
		imprecision. Discuss both direction and magnitude of any potential bias	
Interpretation	20	Give a cautious overall interpretation of results considering objectives,	8-
		limitations, multiplicity of analyses, results from similar studies, and other	
		relevant evidence	
Generalisability	21	Discuss the generalisability (external validity) of the study results	8-
Other informati	on		
Funding	22	Give the source of funding and the role of the funders for the present study and, if	11
		and include the entries of the decay which the account entitle is been d	

*Give information separately for cases and controls in case-control studies and, if applicable, for exposed and unexposed groups in cohort and cross-sectional studies.

Note: An Explanation and Elaboration article discusses each checklist item and gives methodological background and published examples of transparent reporting. The STROBE checklist is best used in conjunction with this article (freely available on the Web sites of PLoS Medicine at http://www.plosmedicine.org/, Annals of Internal Medicine at http://www.annals.org/, and Epidemiology at http://www.epidem.com/). Information on the STROBE Initiative is available at www.strobe-statement.org.

2.3 CHAPTER SYNOPSIS

In this chapter, the prevalence and phenotypes of epilepsy among children with CP in Bangladesh has been described with concurrent exploration of the barriers to optimal epilepsy control.

Following focused epilepsy clinics, 62.0% of the children with CP and epilepsy were clinically determined to have ongoing epileptic seizures. 62.0% had generalized tonic clonic seizures, 27.2% had focal seizures with altered awareness and 10.7% had other seizure types. Non-epileptic events being treated pharmacologically as seizures were found in 18.1% of children. None of these children had an emergency seizure management plan.

The findings highlight the need for such strategies for epilepsy control among children with CP. This work ensured development of local resources and capacity building towards a sustainable model for epilepsy control. This approach to epilepsy control among children with CP in Bangladesh can potentially be scaled up in other low resource settings.

As one of most common associated impairments among children with CP, measures for optimal epilepsy control should be integrated into early intervention and rehabilitation programs for children with CP, particularly in low resource settings. However, early intervention is often limited by delayed diagnosis in LMICs.

Chapter 3 explores strategies for early detection of children with CP in Bangladesh.



CHAPTER 3. EARLY DETECTION OF CHILDREN WITH CEREBRAL PALSY IN BANGLADESH



3.1 INTRODUCTION

The diagnosis of CP in Bangladesh is substantially delayed beyond the critical period of optimizing neuroplasticity of the developing brain.(3, 33) Studies from HICs has shown that it is now possible to identify babies at high risk of CP as early as three months of age. This is well before any other signs, such as activity limitations, are evident. However, in the absence of the use of the best practice tools for the early detection of CP i.e., General Movement Assessment, HINE, and neuroimaging, (46) activity limitation associated with delayed developmental milestones, particularly classical motor milestones, are often the first signs of CP or early onset childhood disability in Bangladesh.

The current best practice guideline recommends a shift towards referral for intervention as soon as an infant is identified to be at risk for CP,(47) as an alternative to delaying it until a formal CP description is confirmed later. This will ensure early intervention when the infant brain has the most neuroplastic potential. Although developmental milestones are not predictive of CP, they may serve as a simple screening tool enabling early referral for intervention to timely nurture the developing brain of infants at risk for CP in low resource settings.

In this study the feasibility of a simple tool, the Developmental Milestone Chart, for neurodevelopmental screening as part of regular health check-ups by medical practitioners in rural Bangladesh is explored.

3.2 PAPER #3. USE OF DEVELOPMENTAL MILESTONE CHART (DMC) IN RURAL BANGLADESH TO EDUCATE HEALTH WORKERS AND STIMULATE REFERRAL FOR EARLY DIAGNOSIS AND INTERVENTION

BRIEF REPORT

Use of a Developmental Milestone Chart (DMC) in Rural Bangladesh to Educate Health Workers and Stimulate Referral for Early Diagnosis and Intervention

Tasneem Karim, MBBS, MPH,^{1,2,3} Alfred Scherzer, MD, EdD, FAAP,⁴ Mohammad Muhit, MBBS, MSc, MSc (Ophth), PhD,^{2,3} Nadia Badawi, MBBCh, DCH, MSc, PhD, FRCPI, FRACP,^{1,5} and Gulam Khandaker, MBBS, DCH, MPH, PhD, FAFPHM,^{1,2,3}

¹Discipline of Child and Adolescent Health, Sydney Medical School, University of Sydney, Sydney 2145, Australia

²Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka 1213, Bangladesh

³CSF Global, Dhaka 1213, Bangladesh

⁴ Weill Cornell Medical College, New York 10065, USA

⁵Cerebral Palsy Alliance Research Institute, University of Sydney, Sydney 2050, Australia

Correspondence: Dr Gulam Khandaker, The Children's Hospital at Westmead (Clinical School), Cnr Hawkesbury Rd and Hainsworth St, 15

Locked Bag 4001, Westmead, NSW 2145, Australia. Tel: +61 0423 229 073, Fax: +612-9845 1418.

E-mail <gulam.khandaker@health.nsw.gov.au>.

ABSTRACT

We aimed to determine the feasibility of using a simplified Developmental Milestone Chart (DMC) for assessment of neurodevelopmental status of children of age ≥ 1 month and ≤ 8 years. Participants were assessed by medical practitioners using DMC as part of regular health checkups in three sub-districts in Bangladesh between January and May 2017. Total 256 children were recruited (41.0% girls, mean age 1.3 ± 1.6 years). Total 107 children (41.8%) failed at least one and 3 (1.2%) failed all four developmental milestones. Majority missed motor milestones (75.6%, n = 81). Four medical practitioners trained on the use of DMC deemed it to be an appropriate tool for developmental screening as part of regular health checkups/care in terms of acceptability, practicality and implementation. In countries with limited facilities, a simplified instrument such as the DMC can be administered by medical practitioners in rural settings. However, further studies are required to establish the validity of DMC before it could be adopted into routine clinical practices.

KEYWORDS: developmental screening; early diagnosis; Bangladesh; low- and middle-income country (LMIC); developmental milestones

© The Author(s) [2018]. Published by Oxford University Press. All rights reserved. For permissions, please email: journals.permissions@oup.com

BACKGROUND

Advances in perinatal care have led to a marked increase in neonatal survival in low- and middleincome countries (LMICs) globally [1]. This is met by concerns of increased adverse neurodevelopmental outcomes [2], which is an emerging concern in countries such as Bangladesh where marked reduction in under-five mortality has been observed in the past decades [3]. Diagnosis of developmental disorders is delayed in part due to lack of awareness among health practitioners in LMICs [4].

Developmental milestones are useful predictors for neurodevelopment, which can enable early identification of disability [5, 6]. Several tools are being used to identify developmental delays. However, they are often extensive, restricted to certain age groups and have not been incorporated into routine clinical practice [7, 8].

We explored the feasibility of the use of the DMC to assess the childhood neurodevelopment as part of regular health checkups by medical practitioners and assessed the neurodevelopmental status of children in rural Bangladesh.

METHODS

This is a feasibility study on the application of a neurodevelopmental screening tool (i.e. DMC) by medical practitioners in rural settings. We followed Fischer *et al.* in a review describing the feasibility of the use of developmental screening tools at primary health-care level in low-and middle-income setting. In consultation with a group of international experts, the following characteristics were selected to assess feasibility: cost of the tool, access to application, training required, time to administer the tool, validity, reliability, results useful to guide action and results understood by caregivers and workers [9].

Children of age ≥ 1 month to ≤ 8 years were assessed by medical practitioners using the DMC in three subdistricts (Shahjadpur, Sirajganj and Manikganj Sadar) located in central and northern parts of Bangladesh between January 2017 and May 2017.

Medical practitioners underwent an hour-long training by the study investigators on the use of the DMC. The tool was administered in outpatient clinics as part of routine checkups. Extensive field notes and comments were collected from the medical practitioners on their experience in administering the DMC. Qualitative data were thematically analysed to determine whether the use of DMC as part of routine medical checkups is appropriate in terms of acceptability, practicality and implementation.

Acceptability reflected the extent to which DMC was considered appropriate [10]. Practicality implied the degree to which the DMC can be used with existing resources and implementation reflected the degree to which it can be used within existing contexts [11].

DMC summarizes age-specific representative milestones (see Supplementary Item 1 and Supplementary Item 2). In total, 50 representative milestones have been selected, modified and included in the DMC from the existing milestone charts and from field experience of health-care providers during previous trials in Cambodia [12–16]. The milestones are grouped into four domains: gross motor, fine motor, language and speech and social development. A translated Bengali version of the tool has been used for this study (see Supplementary Item 2).

Descriptive analysis (mean, median and proportion) was done by SPSS Statistics software version 23 (IBM Corporation, Chicago, Illinois, USA). Feedback from medical practitioners were documented as field notes and reflections, which were subsequently transcribed for further analysis.

This study has been approved by Bangladesh Medical Research Council Human Research Ethics Committee (HREC No. 08914122017) and the Asian Institute of Disability and Development Human Research Ethics Committee (southasia-irb-2017-l-03). Informed written consent was obtained from caregivers.

RESULTS

Neurodevelopmental status of the children

DMC was administered to 256 children as part of routine checkup by four medical practitioners. In total, 41.0% (n = 105) of the study participants were girls, and the mean age was 1.3 ± 1.6 years (median 9 months).

In total, 107 (41.8%) children failed at least one milestone, and 64 (25.0%) children failed at least two milestones. In total, 149 (58.0%) children achieved all milestones and 3 children (1.2%) failed all four milestones. Majority (61.7%, n = 66) of the

Number of milestones failed	n (%)			
None	149 (58.2)			
At least 1	107 (41.8)			
At least 2	64 (25.0)			
At least 3	11 (4.3)			
All 4	3 (1.2)			
Age range	Study	Children who	missed milestone	
	population	Girl (%)	Boy (%)	Total (%)
1–3 months	25	9 (22.0)	8 (12.1)	17 (15.9)
4–6 months	57	9 (22.0)	19 (28.8)	28 (26.2)
7–9 months	51	5 (12.2)	16 (24.2)	21 (19.6)
10-12 months	26	2 (4.9)	4 (6.1)	6 (5.6)
13-15 months	22	4 (9.8)	6 (9.1)	10 (9.3)
16-18 months	20	5 (12.2)	4 (6.1)	9 (8.40
19–23 months	6	0 (0.0)	3 (4.5)	3 (2.8)
2–3 years	20	4 (9.8)	3 (4.5)	7 (6.5)
3–4 years	7	0 (0.0)	0 (0.0)	0 (0.0)
4–5 years	9	1 (2.4)	2 (3.0)	3 (2.8)
5–6 years	0	0 (0.0)	0 (0.0)	0 (0.0)
6–8 years	12	2 (4.9)	1 (1.5)	3 (2.8)
Total	256	41 (100.0)	66 (100.0)	107 (100.0)

Table 1. Number of milestones failed and the number of children failed one or more milestones according to their age

children who failed milestones were between 1 and a 9 months of age (Table 1).

In total, 81 (75.6%) children missed motor milestones. In total, 37 (34.6%) children missed milestones in the language and speech domain, and 35 (32.7%) children missed milestones in the social development domain.

Acceptability

The medical practitioners reported that administration of the tool provided information covering multiple areas of child development that was useful to them for referrals of early intervention and for anticipatory guidance. The tool was easy to interpret and communicate to the study participants, and there were no reported barriers to administration from them. Examples of their responses include:

'The tool is easy to administer and did not take long'

and,

'I am happy to use this routinely during outpatient consultation'

Administration required minimal time (<15 min); however, exact duration was not recorded.

Practicality

The DMC yields valuable information through a brief non-intrusive assessment. All four medical practitioners managed to integrate DMC into routine practice. Moreover, it is a single page tool for up to 8 years of age. Therefore, maintaining the DMC as part of routine medical records of children as they grow is not resource-intensive. However, there was comments made to "add pictures of age appropriate milestones in the form".

Implementation

One trained medical practitioner is required to administer the DMC, and there is no additional cost for administration of the tool as part of routine checkup. The medical practitioners deemed the DMC a helpful tool for timely referral of children for specialized care. It was additionally reported to be useful for training of health workers for establishment of suitable referral system, thereby feasible within the existing context.

DISCUSSION

We primarily aimed to see how DMC will be accepted by health practitioners and what information it may yield if implemented in a rural setting. There are few available tools that have been developed and validated for assessment of neurodevelopmental status of children by low skilled personnel in LMICs like Bangladesh [7]. Two studies conducted in Bangladesh assessed and reported neurodevelopmental status among children in urban and rural settings [7, 8]. However, these tools are extensive and have long administration time (\sim 30–45 min).

The DMC in contrast is a comprehensive tool for developmental screening which is brief, easy to administer, and available in Bengali. We confirmed acceptability of the tool by medical practitioners and the felt need for such a tool in assisting them. Although further studies are required before the DMC could be adopted into routine clinical practices. It is also important to assess community practitioners' views on the use of the tool in subsequent studies to explore whether the DMC can serve as an educational resource for informing health-care workers regarding childhood development to facilitate early referral.

DMC shows some promise as a simple developmental screening tool for frontline health workers in LMICs. For example, it was previously used in Cambodia where it was deemed as a useful tool for continuous training in child development by encouraging health personnel to familiarize with normal milestones and alert them of deviances from the norm [12, 13].

Study limitations

One major limitation of the study is that this is an observational study, and no further follow up was conducted for the children who missed milestones. Further validation studies are needed for estimation of the sensitivity and specificity of the DMC in predicting neurodevelopmental outcomes. Moreover, we had to rely on opportunistic sampling which is not ideal, and the study tool has not been previously validated for local use. Data on time of administration were not collected as part of the study. Advanced statistical analysis was also limited by small sample size. Despite these limitations, we found that DMC is a useful tool for developmental screening. This study provides valuable insights for further exploration.

CONCLUSION

In countries with limited facilities, a simplified instrument such as the DMC can be administered by medical practitioners in rural settings. However, further studies are required to establish the validity of DMC before it could be adopted into routine clinical practices.

FUNDING

This work was supported by CSF Global. TK is supported by the Cerebral Palsy Alliance Research Foundation Career Development Grant [CDG04617].

ACKNOWLEDGEMENTS

The authors would like to thank the children who participated in the study, their caregivers and the study physicians who assessed the study participants using the DMC for their cooperation in the data collection process. The authors would also like to acknowledge the CSF Global team for their support in the implementation of the study.

REFERENCES

- Rajaratnam JK, Marcus JR, Flaxman AD, et al. Neonatal, postneonatal, childhood, and under-5 mortality for 187 countries, 1970–2010: a systematic analysis of progress towards Millennium Development Goal 4. Lancet 2010; 375:1988–2008.
- Lawn JE, Blencowe H, Oza S, et al. Every newborn: progress, priorities, and potential beyond survival. Lancet 2014;384:189–205.
- UNICEF. World prematurity day: saving "every newborn" born too soon. UNICEF. https://blogs.unicef.org/ blog/world-prematurity-day-saving-every-newborn-borntoo-soon/ (14 May 2017, date last accessed)
- Scherzer AL, Chhagan M, Kauchali S, et al. Global perspective on early diagnosis and intervention for children with developmental delays and disabilities. Dev Med Child Neurol 2012;54:1079–84.

- Engle PL, Fernald LC, Alderman H, et al. Strategies for reducing inequalities and improving developmental outcomes for young children in low-income and middleincome countries. Lancet 2011;378:1339–53.
- Brothers KB, Glascoe FP, Robertshaw NS. PEDS: developmental milestones-an accurate brief tool for surveillance and screening. Clin Pediatr 2008;47:271–9.
- Khan NZ, Muslima H, Shilpi AB, et al. Validation of rapid neurodevelopmental assessment for 2- to 5-Year-Old children in Bangladesh. Pediatrics 2013;131:e486–94.
- Khan NZ, Muslima H, Begum D, et al. Validation of rapid neurodevelopmental assessment instrument for undertwo-year-old children in Bangladesh. Pediatrics 2010;125: e755–62.
- Fischer VJ, Morris J, Martines J. Developmental screening tools: feasibility of use at primary healthcare level in lowand middle-income settings. J Health Popul Nutr 2014; 32:314–26.
- Sekhon M, Cartwright M, Francis JJ. Acceptability of healthcare interventions: an overview of reviews and development of a theoretical framework. BMC Health Serv Res 2017;17:88.

- 11. Bowen D, Kreuter M, Spring B, et al. How we design feasibility studies. Am J Prev Med 2009;36: 452–7.
- Scherzer AL. Experience in Cambodia with the use of a culturally relevant developmental milestone chart for children in low-and middle-income countries. J Policy Pract Intellect Disabil 2009;6:287–92.
- 13. Champheaktra N. Developmental milestones in Cambodian children-creation of the Angkor hospital for children milestone assessment tool (AHC DMAT) on siem reap population. Unpublished thesis work. University Health Sciences, Faculty of Medicine: Phnom Penh, Cambodia, 2008.
- 14. Boyle CA, Cordero JF, Trevathan E. The national center on birth defects and developmental disabilities: past, present, and future. Am J Prev Med 2012;43:655–8.
- Frankenburg WK, Dodds J, Archer P, et al. The Denver II: a major revision and restandardization of the Denver developmental screening test. Pediatrics 1992;89: 91–7.
- Frankenburg WK, Dodds JB. The Denver developmental screening test. J Pediatr 1967; 71:181–91.

3.3 CHAPTER SYNOPSIS

In this chapter, strategies for early detection of children with CP in low resouce settings have been explored. The use of the best practice tools for early detection to ensure early intervention is often lacking in low resource settings. The DMC has been identified as a suitable tool for developmental screening that could be incorporated into regular health checkups in low resource settings. It can serve as a valuable screening tool to ensure early referral for comprehensive assessment by a qualified clinician for an interim high risk of CP clinical diagnosis based on risk factors and clinical presentation and thereby facilitate early intervention within the period of optimal neuroplasticity.

Given the known importance of early intervention, we then explored the feasibility and outcomes of an early intervention program in a rural subdistrict of Bangladesh. The results of this trial are reported in the following chapter, Chapter 4.

CHAPTER 4. INTERVENTION FOR CHILDREN WITH CEREBRAL PALSY IN BANGLADESH





4.1 INTRODUCTION

Delayed diagnosis of CP limits early intervention within the period of optimum neuroplasticity of the developing brain. Although the benefits of early intervention for children with CP are apparent and evidence-based interventions for CP are available, (46) there is an overall dearth of evidence on the effect of interventions for children with CP in LMICs, such as Bangladesh. Caregiver led community-based rehabilitation programs are instrumental in an LMICs such a Bangladesh, where there is an overall lack of trained allied health workers.

In this chapter, the outcome of a community-based parent led early intervention and rehabilitation program for children with CP in rural Bangladesh is described.

4.2 PAPER #4. OUTCOME OF COMMUNITY-BASED PARENTS LED EARLY INTERVENTION FOR CHILDREN WITH CEREBRAL PALSY IN RURAL BANGLADESH: A QUASI-EXPERIMENTAL STUDY



Article

Outcome of community-based early intervention and rehabilitation for children with cerebral palsy in rural Bangladesh: A quasi-experimental study

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

Tasneem Karim^{1,2,3*}, Mohammad Muhit^{1,2}, Israt Jahan^{1,2,5}, Claire Galea⁴, Catherine Morgan^{3,4}, Hayley Smithers-Sheedy^{3,4}, Nadia Badawi^{3,4}, Gulam Khandaker^{3,5,6}

¹ CSF Global, Dhaka, Bangladesh

- ² Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka, Bangladesh
- ³ Specialty of Child and Adolescent Health, Sydney Medical School, University of Sydney, Sydney, Australia
 - Cerebral Palsy Alliance Research Institute, University of Sydney, Sydney, Australia
- ⁵ School of Health, Medical and Applied Sciences, Central Queensland University, Rockhampton, Queensland, Australia
- ⁶ Central Queensland Public Health Unit, Central Queensland Hospital and Health Service, Queensland Government, Rockhampton, Australia
- Correspondence: Tasneem.karim.tk@gmail.com
- Abstract: We aimed to evaluate the outcome of a community-based early intervention and 17 rehabilitation for children with cerebral palsy (CP) in Bangladesh. Children registered into the 18 Bangladesh CP Register (BCPR) were recruited in two groups for this study: Group A received a 19 comprehensive six months long community-based caregiver-led intervention and rehabilitation 20 program at the CSF "Shishu Shorgo" Rehabilitation and Early Intervention Centres developed to 21 support participants from the BCPR. Group B received standard care. A quasi-experimental study 22 was conducted. Data were obtained at baseline, end of program (i.e. 6 months), 12-month follow-23 up. Outcome measures included gross motor functional measure (GMFM-66), Communication 24 Function Classification System (CFCS), Viking Speech Scale (VSS) of children and depression, 25 anxiety, and stress scale (DASS 21) of primary caregivers. Between October-2016 and March-2017, 26 156 children with CP were recruited (77 in Group A and 79 in Group B). the total score of GMFM-27 66, CFCS level and VSS level significantly improved in Group A (p<0.05 for all), and deteriorated in 28 Group B (p<0.001, p=0.095, p=0.232). The intervention showed promising outcomes particularly for 29 children with CP under five years of age. There is a need for caregiver led community-based 30 rehabilitation programs for children with CP in LMICs. 31

Keywords: Cerebral palsy; children; early intervention; community-based; low-and middle-income32country; Bangladesh33

34

35

1. Introduction

Cerebral palsy (CP) is the most common physical disability of childhood. The most 36 recent and widely used definition of CP is "a group of permanent disorders of the 37 development of movement and posture, causing activity limitation, that are attributed to 38 non-progressive disturbances that occurred in the developing fetal or infant brain. The 39 motor disorders of cerebral palsy are often accompanied by disturbances of sensation, 40 perception, cognition, communication, and behaviour; by epilepsy, and by secondary 41 musculoskeletal problems" [1]. While the knowledge on the antecedents and preventive 42

Citation: Lastname, F.; Lastname, F.; Lastname, F. Title. *Brain Sci.* **2021**, *11*, x. https://doi.org/10.3390/xxxx

Academic Editor: Firstname Lastname

Received: date Accepted: date Published: date

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Submitted for possible open access publication under the terms and conditions of the Creative Commons Attribution (CC BY) license (https://creativecommons.org/licenses/ by/4.0/). strategies for CP has advanced considerably, our understanding of the prognosis of CP in terms of type and severity of everyday functioning, particularly in low resource settings, remains poorly understood.[2-4] Severity of brain lesions, age at the time of diagnosis and first rehabilitation, and access to evidence-based interventions are some of the important predictors of functional outcomes for children with CP.[5] 43

Recent studies from low and middle-income countries (LMICs) report a greater 49 burden and severity of CP and associated impairments.[6-8] A population-based study in 50 Bangladesh additionally reported delayed diagnosis at 5.2 years of age [6] beyond the 51 period of optimum neuroplasticity of the developing brain.[9] Children with CP reach 52 90% of their gross motor potential within the first five years of life, and even earlier for 53 those with severe CP.[10-12] Early intervention is therefore crucial for optimizing both 54 motor and functional outcomes of children with CP. [13] 55

In addition to maximizing neuroplasticity and minimizing the detrimental modifications to muscle and bone growth, nutrition and development, outcomes of interventions for children with CP are strongly linked to caregiver wellbeing.[9,14] This has been evidenced in studies from LMICs including Bangladesh and Ghana.[14,15] Findings from a study in Zimbabwe also reported that caregivers of children with CP had poor health-related quality of life, high levels of depression, anxiety and stress, and felt overwhelmed by the economic burden and their caregiving role.[16]

Due to the heterogeneity of CP in terms of etiology, brain injury, severity of 65 impairments and co-occurring conditions, children with CP have diverse needs which 66 warrant a comprehensive rehabilitation program. These programs also need to address a 67 range of medical, social and cultural barriers prevalent in rural and remote communities 68 in LMICs.[17] A recent systematic review highlights the lack of evidence on efficacy for 69 the majority of the interventions in use and the dire need for further research to address 70 the existing research-practice gaps.[18] This is particularly true in LMICs such as 71 Bangladesh where children with CP and their families are faced with numerous barriers 72 in addition to those still prevailing in high income settings.[19] 73

Barriers to intervention for children with CP include a) delayed diagnosis of CP 75 beyond critical window of neuroplasticity, b) poor or no access to evidence-based early 76 intervention, c) inadequate state funded support and initiatives further propagating the 77 disability and poverty cycle and widening the prevailing inequities, d) inability to afford 78 services, e) poor accessibility due to lack of disability inclusive infrastructure and public 79 transport systems, f) health workforce crisis particularly of allied health workers required 80 to support children with CP and g) prevailing misconceptions, stigma and social 81 exclusion. It is imperative that these factors are considered in the development of 82 interventions with enhanced relevance for children with CP and their families in low 83 resource settings.[6,17,19-22] 84

CSF Global (www.csf-global.org), an independent not-for-profit organization, committed to the establishment of a rights-based inclusive society for children with disability in LMICs, developed a comprehensive community-based caregiver-led intervention program for children with CP provided at the CSF "Shishu Shorgo" (Bengali title which translates to 'Children's Heaven') Rehabilitation and Early Intervention Centres in Bangladesh. This program was developed to support participants from the Bangladesh CP Register (BCPR). In this study, we aimed to evaluate the outcome of this 92

48

56

57

58

59

60

61

62

63 64

74

95

96

104

110

2. Materials and Methods

of Bangladesh.

Study design, setting and participants

This was a pragmatic design quasi-experimental study conducted in a Northern subdistrict of Bangladesh. The study participants included a subset of the BCPR study 98 population. The BCPR is an ongoing surveillance of children (0–18 years) which includes 99 children with CP from rural communities using a novel method of case ascertainment, the 100 key informant method (KIM), followed by clinical confirmation of diagnosis of CP and 101 data collection by a multidisciplinary team including assessment by a physician. The 102 detailed study protocol and findings have been described in previous publications.[6,23] 103

existing program for children with CP and their primary caregivers in a rural subdistrict

The rehabilitation needs of the children are assessed and documented as part of the 105 BCPR study. The CSF "Shishu Shorgo" rehabilitation centre's services were offered to all 106 children with CP who had an identified need for therapy between October 2016 and 107 March 2017. However, not all families were able to access these services due to personal 108 or family circumstances. 109

The CSF "Shishu Shorgo" rehabilitation centres offers a six months program for 111 children with CP with two intakes per year. Children enrolled at February and August 112 2016 intake (to be known throughout this manuscript as Group A) were recruited for this 113 study. Additionally, data were collected on another group (to be known throughout this 114 manuscript as Group B) of BCPR registrants who were not able to participate in the 115 program. This group received standard care only, which consisted of basic advice 116 provided by the BCPR team including physician and physiotherapist at the time of 117 registration into the BCPR. Primary caregivers of all the children were also interviewed 118 for assessment of their mental wellbeing. All assessments were conducted as part of the 119 services provided at the CSF "Shishu Shorgo" Rehabilitation Centres for Group A. 120 Assessments for Group B were conducted during home visits by an assistant 121 physiotherapist trained in study procedures. 122

We collected outcome data on both groups during three time points; at baseline (prior 124 to enrolment into the program), at the end of the program (i.e. 6 months) and follow-up 125 at 12 months.

Intervention received by Group A

The program at CSF "Shishu Shorgo" serves the local community as the backbone 128 for intervention for children with CP. It is a family centered program largely run by 129 primary caregivers along with two community therapists (CT) in each center. The CTs are 130 local community members who receive 15 days of structured training from a qualified 131 physiotherapist on the provision of family centered rehabilitation services for children 132 with disability and at least three months of supervised practice in a community 133 rehabilitation centre before they lead a group under supervision of the training 134 physiotherapist. 135

Twenty children were enrolled in each intake per centre, a morning group and 137 afternoon group each with ten children. The children at CSF "Shishu Shorgo" undergo 138 assessment and individual goal setting by the primary caregivers and the CT at 139 enrollment. They attend three hourly sessions a day, five days a week at CSF "Shishu 140

127

136

Shorgo" centers. A manual (https://www.disabilityasia.org/parent-information) has been 141 developed to assist the CTs to run the early intervention program aimed at optimising 142 neurocognitive outcomes among children with CP in rural communities. It also promotes 143 participation of the children with CP in their family, school, and the community. This 144 "Shishu Shorgo" early intervention program is comprised of the following key 145 components, 146

1. Group therapy: The group therapy focuses on the development of the following 148 skills: activities of daily living (toileting, dressing, eating), language and communication, 149 movement, cognition, and social skills. 150

2. Community follow-up: Throughout the six months long program, the CT provide 151 community follow up for each child and family. The goals of this community follow-up 152 are i) provision of strategies and assistive devices to assist the child at home with activities 153 of daily living (washing, toileting, eating, dressing), ii) supporting the child's local school 154 to enable their admission to school and participation in school activities, iii) increasing 155 awareness about disability and the child's abilities and rights to facilitate the child's family 156 and community develop support networks and increase opportunities for the child to 157 participate in the community, and iv) supporting the child to develop a meaningful 158 vocation in their family and community, particularly for those children unable to attend 159 school. 160

3. Primary caregiver training, peer support and education: The children attend the 161 sessions with at least one primary caregiver who is engaged in all elements of the program 162 to develop skills on day-to-day care of the children with CP using the CSF "Shishu 163 Shorgo" program manual. Parent support and education is provided through involving 164 the child's primary caregivers in all elements of the program. The CT involve the parent 165 in group therapy and community follow-up by providing family-centred care, keeping 166 the primary caregiver informed about the child's progress, and providing 167 recommendations and empowering primary caregivers to advocate for and facilitate their 168 child's participation in their home, community, school and vocation. Through this 169 program, the primary caregivers form support networks/peer group with other caregivers 170 of children with disabilities. 171

Intervention received by Group B

Group B did not opt for the rehabilitation at the CSF "Shishu Shorgo" Rehabilitation 173 Centres and received standard care i.e., no rehabilitation service aside from advice 174 provided by the multidisciplinary team at the BCPR camps during recruitment into the 175 register. 176

Outcome Measures assessed for both Groups A and B

The children and their primary caregivers were assessed at baseline at the time of 178 enrollment, after six-month period to measure the immediate outcome of the intervention 179 and after twelve months to evaluate long term effects of the intervention. A qualified 180 physiotherapist was trained by the study investigators on the assessments. The following 181instruments were used for assessment of the children and their primary caregivers, 182

1. Motor function: Gross Motor Function Measure (GMFM-66) and Gross Motor 184 Function Classification System (GMFCS) were used to evaluate changes in motor function 185 of the children with CP.[24,25] The GMFM is a criterion referenced assessment designed 186 and validated to measure change in gross motor function over time in children with CP. 187 GMFCS evaluates movement skills such as sitting, walking and use of mobility devices 188 and categorizes gross motor function into five levels. We reported and analyzed GMFCS 189

147

177

183

200

201

202 203

213

214

219

220

as ordinal and GMFM scores as continuous variables. GMFM-66 data was entered, stored, 190 and analyzed using the Gross Motor Ability Estimator (GMAE-2) Scoring Software for the 191 GMFM.[26] 192

2. Communication: Communication Function Classification System (CFCS), was used
194
to classify the everyday communication of the children with CP into one of five levels
according to effectiveness of communication which considers means of communication
196
including speech, gesture, facial expression and augmentative and alternative
communication.[27] A child classified at Level I is a more effective communicator than a
child classified at Level V. We reported and analyzed CFCS as an ordinal variable.

3. Speech: Viking Speech Scale (VSS) is used to classify children's speech production, We reported and analyzed as an ordinal variable.[28]

4. Mental wellbeing of primary caregivers: Depression Anxiety Stress Scale (DASS-21), a 204 21 item self-report instrument, was used to measure - depression, anxiety and stress - of 205 the primary caregivers of the children with CP.[29] The primary caregivers were asked to 206 use 4-point severity/frequency scales to rate the extent to which they have experienced 207 each state over the past week. DASS-21 scores were summed into 'depression', 'anxiety' 208 and 'stress' scale and categorized as 'normal', 'mild', 'moderate', 'severe' or 'extremely 209 severe' as per instrument protocol for analysis and interpretation. We reported and 210 analyzed Depression, Anxiety and Stress scores as continuous variable for Friedman 211 analysis. 212

Ethical considerations

Ethics approval for this study was obtained from the Bangladesh Medical Research215Council (BMRC) (Ref: BMRC/NREC/2016-2019/469) and Asian Institute of Disability and216Development (AIDD) (southasia-irb- 2016-1-07). Informed written consent was obtained217from the primary caregivers of the children.218

Statistical Analysis

Descriptive methods were used to summarize the cohort. Normality was checked 221 using Shapiro-Wilk and visual inspection of the box plots. Normality assumptions failed 222 therefore the non-parametric Friedman Test with post hoc pairwise comparisons was used 223 and reported using median and Interquartile Range (IQR). However, data was also 224 reported using the mean (SD) for clinical interpretation. There was a statistically 225 significant difference between ages for Group A and Group B therefore we could not run 226 any regression analysis using age as a covariate. Analysis was carried out using SPSS 227 version 24 (IBM Corporation, Chicago, Illinois, USA). A p-value of <0.05 was considered 228 significant. 229

3. Results

Between October 2016 and March 2017, 156 children with CP were recruited to the 231 study (77 in Group A and 79 in Group B). Seventy-seven children with CP enrolled in the 232 CSF "Shishu Shorgo" formed Group A and 79 children with CP who received standard 233 care (i.e., basic rehabilitation education with no structured rehabilitation) formed Group 234 B (mean (SD) age at baseline (i.e., 0 month), Group A vs Group B: 4.3 (2.9) vs 11.1 (4.0), p 235 value<0.001. Post hoc analysis to investigate the age differences were performed and 236 comparison within groups were also conducted where groups were not matched by age 237 and sample size. 238

239

The demographic characteristics of the study participants are summarized in Table 240 1. The male-female ratio in Group A and Group B were 1.4:1 and 1.5:1 respectively 241 (p=0.448). Both groups had similar sociodemographic characteristics; no significant 242 (p>0.05) difference was observed between the groups in terms of type of accommodation, 243 source of drinking water, sanitation, paternal education, and occupation. However, the 244 median monthly family income and maternal educational level were significantly lower 245 in Group B compared with Group A (p=0.005 and p=0.010 respectively).

Characteristic	Group A (n=77)	Group B (n=79)	<i>p</i> -value ^b	
Age				
Mean (SD)	4.3 (2.9)	11.1 (4.0)	<0.001°	
Median [IQR]	3.4 (2.4, 5.4)	11.6 (7.9, 14.1)	<0.001 ^d	
Sex				
Male	45 (58.4)	48 (60.8)	0.448 ^e	
Female	32 (41.6)	31 (39.2)		
Monthly family income (BDT)~ [USD] ^{a,f}				
Mean (SD)	9870.5 (6390.7) [~ 84.8 (75.3)]	8256.4 (6289.9) ~ 97.3 (74.1)	0.119 ^c	
Median [IQR]	8000.0 [6000.0, 10000.0] ~ 94.3 [70.7, 117.9]	6000.0 [5500.0, 8250.0] ~ 70.7 [64.8, 97.2]	0.005 ^d	
Monthly family income in BDT (USD) ^a				
Below 10,000 (below ~120)	49 (66.2)	60 (76.9)	0.052	
10,000-19,999 (~120-241)	15 (20.3)	15 (19.2)		
20,000-29,999 (~241-361)	9 (12.2)	1 (1.3)		
30000 and above (~361 & above)	1 (1.4)	2 (2.6)		
Type of accommodation ^a				
Temporary shelter (jhupri)	0 (0.0)	2 (2.6)	0.240	
Mud (kutcha) house	55 (74.3)	61 (78.2)		
Semi-permanent (semi-pucca) house	13 (17.6)	13 (16.7)		
Permanent brick (pucca) house	6 (8.1)	2 (2.6)		
Source of drinking water ^a				
Tap water	1 (1.4)	0 (0.0)	0.487	
Tube well	73 (98.6)	78 (100.0)		
Sanitation ^a				
No toilet facility	2 (2.7)	1 (1.3)	0.816	
Non-sanitary latrine	23 (31.1)	24 (30.8)		
Sanitary latrine	49 (66.2)	53 (67.9)		
Maternal education ^a				
Illiterate	20 (27.0)	40 (51.3)	0.010	
Primary	26 (35.1)	24 (30.8)		
Secondary	22 (29.7)	11 (14.1)		
Higher secondary and above	6 (8.2)	3 (3.8)		
Maternal occupation ^a				
Agriculture worker	1 (1.4)	2 (2.6)	0.720	
Garments worker/weaver/tailor	9 (12.2)	12 (15.4)		
Unemployed	64 (86.5)	64 (82.1)		
Paternal education ^a				
Illiterate	26 (35.1)	40 (51.3)	0.174	
Primary	23 (31.1)	19 (24.4)		
Secondary	18 (24.3)	16 (20.5)		

Table 1. Sociodemographic characteristics of the study participants

246

Characteristic	Group A (n=77)	Group B (n=79)	<i>p</i> -value ^b
Higher secondary and above	7 (9.5)	3 (3.8)	
Paternal occupation ^a			
Agriculture worker	11 (15.1)	23 (30.3)	0.207
Daily wage earners	10 (13.7)	12 (15.8)	
Business	16 (21.9)	16 (21.1)	
Garments worker/weaver/tailor	24 (32.9)	19 (25.0)	
Others	11 (15.1)	5 (6.6)	
Unemployed	1 (1.4)	1 (1.4)	

3.1 GROUP A

GMFM Score

The median GMFM total score significantly improved between baseline and 12 months (p<0.001) in Group A. Similar significant improvement was observed for children aged less than five years however, for children aged five years and more, the scores remained unchanged between baseline and 12 months (p=0.058).(Table 2 and 3)

GMFCS Level

Overall, 81.6% children had GMFCS level III-V at baseline, which reduced to 67.6%257at 6 months and again increased to 83.7% at 12 months (p<0.001) (Table 2). When</td>258disaggregated by age, the GMFCS level significantly improved among children aged less259than five years between 0 and 6 months (p<0.001), and then slightly deteriorated between</td>2606 and 12 months (p<0.001). Whereas for children aged five years and above in this group,</td>261the GMFCS level remained similar between 0 and 6 months (p=0.285) and significantly262deteriorated between 6 and 12 months (p=0.792) (Table 3).263



GMFM: Gross Motor Function Measure, CFCS: Communication Function Classification System, VSS: Viking Speech Scale, DASS-21: Depression Anxiety Stress Scale

248

249 250

251

252

253

254 255



CFCS Level

Overall, the CFCS level significantly improved between baseline and 12 months of 266 the study (p=0.002) (Table 2). Among the children aged less than five years, the proportion 267 of children with CFCS level V reduced from 41.1% (n=23) to 24.5% (n=12) whereas, CFCS 268 level I increased from 23.2% (n=13) to 26.5% (n=13) between baseline and 12 months 269 (p=0.007). A similar less pronounced change was observed among the children aged five 270 years and above (p=0.029) (Table 3).

VSS Level

Overall a significant improvement in VSS level was observed (p<0.001) (Table 2). 273 Upon disaggregation by age, the children aged less than five years showed significant 274 improvement between baseline and 6 months (p=0.041), 6 and 12 months (p<0.001), and 275 baseline and 12 months (p<0.001). Whereas the children aged five years and above, 276 showed a significant improvement between baseline and 6 months (p=0.014), followed by 277 deterioration between 6 and 12 months (p=0.070); there was an overall significant change 278 across all three time points (p=0.023) (Table 3). 279

Primary Caregiver DASS 21: Depression

At baseline overall 37.7% (n=29) of the caregivers had mild to extremely severe 282 symptoms of depression. Although this percentage decreased to 19.7% (n=15) at 6 months 283 of the study, we observed a marked increase (50.8%, n=34) in the proportion of caregivers with mild to extremely severe symptoms of depression at 12 months (p<0.001) (Table 2). 285 These scores on depression subscale significantly improved among caregivers of children 286 aged less than five years between baseline and 6 months (p<0.001), followed by a 287 significant increase between 6 and 12 months (p<0.001. However, the overall change 288 between baseline and 12 months was not significant. A similar pattern was observed for 289 caregivers of children aged five years and above (Table 3). Table 2. Descriptive findings of 290 the study participants 291

		Group A				Group B		
Timepoint	0 month	6 months	12 months	p value ^a	0 month	6 months	12 months	p value ^a
	n=77	n=77	n=67	-	n=79	n=73	n=67	-
GMFM Total score								
Mean (SD)	32.5 (18.6)	42.2 (20.1)	42.3 (18.2)	-	39.3 (22.0)	46.4 (23.5)	41.2 (19.4)	-
Median (IQR)	34.8 [16.0, 46.5]	43.6 [23.6, 54.4]	44.6 [30.5, 52.3]	<0.001	46.9 [16.0, 59.3]	46.9 [26.3, 65.8]	42.2 [22.7, 54.6]	<0.001
GMFCS, n [%]	76	77	67		79	73	67	
Level I	4 (5.3)	11 (14.3)	5 (7.5)	<0.001	1 (1.3)	7 (9.6)	1 (1.5)	<0.001
Level II	10 (13.2)	14 (18.2)	6 (9.0)		22 (27.8)	20 (27.4)	12 (17.9)	
Level III	14 (18.4)	19 (24.7)	20 (29.9)		19 (24.1)	11 (15.1)	16 (23.9)	
Level IV	16 (21.1)	14 (18.2)	17 (25.4)		7 (8.9)	16 (21.9)	15 (22.4)	
Level V	32 (42.1)	19 (24.7)	19 (28.4)		30 (38.0)	19 (26.0)	23 (34.3)	

265

271

272

281

280



		Group A			Group B			
Timepoint	0 month	6 months	12 months	p value ^a	0 month	6 months	12 months	p value ^a
	n=77	n=77	n=67	-	n=79	n=73	n=67	-
CFCS, n [%]					n=77 ^b			
Level I	15 (19.5)	24 (31.2)	16 (23.9)	0.002	20 (26.0)	29 (39.7)	19 (28.4)	0.095
Level II	12 (15.6)	7 (9.1)	8 (11.9)		11 (14.3)	5 (6.8)	6 (9.0)	
Level III	11 (14.3)	13 (16.9)	19 (28.4)		13 (16.9)	9 (12.3)	18 (26.9)	
Level IV	11 (14.3)	12 (15.6)	8 (11.9)		17 (22.1)	12 (16.4)	3 (4.5)	
Level V	28 (36.4)	21 (27.3)	16 (23.9)		16 (20.8)	18 (24.7)	21 (31.3)	
VSS, n [%]								
Level I	4 (5.2)	8 (10.4)	13 (19.4)	<0.001	15 (19.0)	28 (38.4)	20 (29.9)	0.232
Level II	6 (7.8)	7 (9.1)	11 (16.4)		13 (16.5)	0 (0.0)	6 (9.0)	
Level III	5 (6.5)	5 (6.5)	5 (7.5)		6 (7.6)	5 (6.8)	7 (10.4)	
Level IV	13 (16.9)	13 (16.9)	38 (56.7)		41 (51.9)	39 (53.4)	34 (50.7)	
NA [aged ≤4 years]	49 (63.6)	44 (57.1)	0 (0.0)		4 (5.1)	1 (1.4)	0 (0.0)	
DASS 21, n [%]								
Depression symptoms		n=76 ^b						
Normal	48 (62.3)	61 (80.3)	33 (49.3)	<0.001	73 (92.4)	53 (72.6)	31 (46.3)	<0.001
Mild	7 (9.1)	8 (10.5)	13 (19.4)		6 (7.6)	7 (9.6)	22 (32.8)	
Moderate	14 (18.2)	6 (7.9)	20 (29.9)		0 (0.0)	11 (15.1)	14 (20.9)	
Severe	3 (3.9)	0 (0.0)	1 (1.5)		0 (0.0)	2 (2.7)	0 (0.0)	
Extremely severe	5 (6.5)	1 (1.3)	0 (0.0)		0 (0.0)	0 (0.0)	0 (0.0)	
Anxiety symptoms		n=76 ^b				n=72 ^b		
Normal	45 (58.4)	56 (73.7)	4 (6.0)	<0.001	79 (100.0)	28 (38.9)	3 (4.5)	<0.001
Mild	14 (18.2)	6 (7.9)	4 (6.0)		0 (0.0)	10 (13.9)	2 (3.0)	
Moderate	2 (2.6)	7 (9.2)	27 (40.3)		0 (0.0)	22 (30.6)	32 (47.8)	
Severe	9 (11.7)	3 (3.9)	26 (38.8)		0 (0.0)	10 (13.9)	20 (29.9)	
Extremely severe	7 (9.1)	4 (5.3)	6 (9.0)		0 (0.0)	2 (2.8)	10 (14.9)	
Stress symptoms		n=76 ^b						
Normal	35 (45.5)	54 (71.1)	10 (14.9)	<0.001	34 (43.0)	56 (76.7)	5 (7.5)	<0.001
Mild	19 (24.7)	10 (13.2)	34 (50.7)		32 (40.5)	11 (15.1)	37 (55.2)	
Moderate	10 (13.0)	9 (11.8)	23 (34.3)		13 (16.5)	5 (6.8)	18 (26.9)	
Severe	8 (10.4)	1 (1.3)	0 (0.0)		0 (0.0)	1 (1.4)	7 (10.4)	
Extremely severe	5 (6.5)	2 (2.6)	0 (0.0)		0 (0.0)	0 (0.0)	0 (0.0)	

_

^a Friedman test, ^b Missing data

Primary Caregiver DASS 21: Anxiety

Overall, 41.6% (n=32) caregivers showed signs of mild to extremely severe symptoms 295 of anxiety at baseline which decreased to 26.3% (n=20) at 6 months followed by a sharp 296 rise to 94.0% (n=63) at 12 months (p<0.001) (Table 2). Among caregivers of children aged 297 less than five years, the median (IQR) of anxiety score slightly improved between baseline 298 and 6 months (p=0.046) but it increased significantly between 6 and 12 months (p<0.001). 299 Similar observation was made among caregivers of children aged five years and above 300 (Table 3). 301

Primary Caregiver DASS 21: Stress

Overall, a substantial improvement in the scores of stress subscale of the caregivers 304 was observed between baseline and 6 months, however, this was followed by a 305 deterioration between 6 months and 12 months (p<0.001) (Table 2). The median (IQR) 306 score on stress subscale reduced significantly between baseline and 6 months among the 307 caregivers of children aged less than five years (p<0.001), and caregivers of children aged 308 above five years (p=0.002). However, the score increased significantly between 6 and 12 months for both the groups (Table 3).

3.2 GROUP B

GMFM Score

The GMFM total score significantly deteriorated between baseline and 12 months among children in Group B (p<0.001). When disaggregated by age, the median GMFM 314 total score deteriorated between baseline and 6 months among both children aged less 315 than five years (p=0.600) and children aged five years and above (p<0.001). The pattern 316 remained unchanged between 6 and 12 months (p=0.345 for children aged less than five 317 years and p<0.001 for children aged five years and above) (Table 2 and Table 3). 318

GMFCS Level

Overall, significant deterioration in GMFCS level between baseline and 12 months 321 was also observed for children in Group B (GMFCS level III-V at baseline vs 12 months: 322 71.0% vs. 80.6%; p<0.001). However, the changes were not significant for children aged 323 less than five years (p=0.097) but significant for children aged five years and above 324 (p=<0.001) (Table 2 and Table 3). 325

293

294

302

303

309 310

311

312 313

319





Table 3. GMFM 66 and GMFCS among children in Group A (n=77) and Group B (n=79) according to their age at baseline*	326
*including data for the children lost to follow up and the deceased	327

	Group	A aged less than	n 5 years	Group	A aged 5 or mo	re years	Group	Group B aged less than 5 years		Group	B aged 5 or mo	re years	
Time point	0m (n=56)	6m (n=56)	12m (n=49)	0m (n=21)	6m (n=21)	12m (n=18)	0m (n=7)	6m (n=6)	12m (n=6)	0m (n=71)	6m (n=67)	12m (n=61)	
GMFM 66	n=56	n=56	n=49	n=21	n=21	n=18	n =7	n=6	n=6	n=71	n=67	n=61	
Mean (SD)	31.1 (18.3)	42.1 (20.1)	41.6 (17.8)	36.6 (19.3)	42.6 (20.6)	44.0 (19.7)	34.2 (24.4)	36.7 (30.0)	31.5 (21.3)	40.1 (21.9)	47.3 (22.9)	42.1 (19.1)	
Median [IQR]	30.2 [14.8, 46.0]	43.2 [24.8, 54.4]	46.1 [27.2, 52.4]	43.6 [17.8, 51.0]	44.5 [22.9, 59.1]	43.6 [29.0, 52.6]	36.0 [10.4, 58.6]	33.7 [11.1, 64.1]	29.8 [10.4, 50.2]	48.5 [16.0, 59.9]	47.7 [30.0, 66.7]	44.4 [24.0, 55.1]	
<i>p</i> value	Across all th 6m: <0.001	ree time points: , 6m-12m: 0.06 <0.001	: <0.001 , 0m - 8, 0m - 12m:	1, 0m - Across all three time points: 0.058, 0m - Across al · 12m: 6m: 0.009, 6m-12m:0.492, 0m - 12m: 0.093 6m: 0.60			Across all th 6m: 0.600,	Across all three time points: 0.607, 0m - 6m: 0.600, 6m-12m: 0.345, 0m - 12m: 0.463			Across all three time points: <0.001 , 0m - 6m: <0.001 , 6m-12m: <0.001 , 0m - 12m: 0.147		
GMFCS level*													
Level I	4 (7.3)	11 (19.6)	4 (8.2)	0 (0.0)	0 (0.0)	1 (5.6)	1 (14.3)	1 (16.7)	0 (0.0)	0 (0.0)	6 (9.0)	1 (1.6)	
Level II	5 (9.1)	9 (16.1)	6 (12.2)	5 (23.8)	5 (23.8)	0 (0.0)	1 (14.3)	1 (16.7)	1 (16.7)	21 (29.6)	19 (28.4)	11 (18.0)	
Level III	14 (25.5)	14 (25.0)	13 (26.5)	0 (0.0)	5 (23.8)	7 (38.9)	1 (14.3)	1 (16.7)	1 (16.7)	18 (25.4)	10 (14.9)	15 (24.6)	
Level IV	8 (14.5)	10 (17.9)	13 (26.5)	8 (38.1)	4 (19.0)	4 (22.2)	1 (14.3)	0 (0.0)	1 (16.7)	6 (8.5)	16 (23.9)	14 (23.0)	
Level V	24 (43.6)	12 (21.4)	13 (26.5)	8 (38.1)	7 (33.3)	6 (33.3)	3 (42.9)	3 (50.0)	3 (50.0)	26 (36.6)	16 (23.9)	20 (32.8)	
<i>p</i> value	Across all th 6m: <0.001 ,	ree time points: 6m-12m: <0.00 0.192	ts: <0.001, 0m -Across all three time points: 0.107, 0m -001, 0m - 12m:6m: 0.285, 6m-12m:0.792, 0m - 12m: 0.776		:: 0.107, 0m - m - 12m: 0.776	Across all three time points: 0.097, 0m - 6m: 0.317, 6m-12m: 0.180, 0m - 12m: 0.102		:: 0.097, 0m -), 0m - 12m:	Across all three time points: <0.001 , 0m - 6m: 0.001 , 6m-12m: <0.001 , 0m - 12m: 0.168		: <0.001 , 0m - 1 , 0m - 12m:		
CFCS level										n=69			
Level I	13 (23.2)	16 (28.6)	13 (26.5)	2 (9.5)	8 (38.1)	3 (16.7)	1 (14.3)	1 (16.7)	0 (0.0)	19 (27.5)	28 (41.8)	19 (31.1)	

	Group A aged less than 5 years			Group A aged 5 or more years			Group B aged less than 5 years			Group B aged 5 or more years		
Time point	0m (n=56)	6m (n=56)	12m (n=49)	0m (n=21)	6m (n=21)	12m (n=18)	0m (n=7)	6m (n=6)	12m (n=6)	0m (n=71)	6m (n=67)	12m (n=61)
GMFM 66	n=56	n=56	n=49	n=21	n=21	n=18	n =7	n=6	n=6	n=71	n=67	n=61
Level II	4 (7.1)	4 (7.1)	3 (6.1)	8 (38.1)	3 (14.3)	5 (27.8)	0 (0.0)	0 (0.0)	1 (16.7)	11 (15.9)	5 (7.5)	5 (8.2)
Level III	7 (12.5)	10 (17.9)	14 (28.6)	4 (19.0)	3 (14.3)	5 (27.8)	1 (14.3)	2 (33.3)	2 (33.3)	12 (17.4)	7 (10.4)	16 (26.2)
Level IV	9 (16.1)	9 (16.1)	7 (14.3)	2 (9.5)	3 (14.3)	1 (5.6)	3 (42.9)	1 (16.7)	0 (0.0)	13 (18.8)	11 (16.4)	3 (4.9)
Level V	23 (41.1)	17 (30.4)	12 (24.5)	5 (23.8)	4 (19.0)	4 (22.2)	2 (28.6)	2 (33.3)	3 (50.0)	14 (20.3)	16 (23.9)	18 (29.5)
<i>p</i> value	Across all three time points: <0.007 , 0m - 6m: <0.001 , 6m-12m: 0.73, 0m - 12m: <0.012			Across all three time points: 0.029 , 0m - 6m: 0.011 , 6m-12m:0.083, 0m - 12m: 0.705			Across all three time points: 0.807, 0m - 6m: 0.564, 6m - 12m: 0.414, 0m - 12m: 0.705			Across all three time points: 0.110, 0m - 6m: 0.177, 6m-12m: 0.015 , 0m - 12m: 0.131		
VSS												
Level I	2 (3.6)	2 (3.6)	11 (22.4)	2 (9.6)	6 (28.6)	2 (11.1)	0 (0.0)	0 (0.0)	0 (0.0)	15 (21.1)	28 (41.8)	20 (32.8)
Level II	1 (1.8)	3 (5.4)	6 (12.2)	5 (23.8)	4 (19.0)	5 (27.8)	0 (0.0)	0 (0.0)	1 (16.7)	13 (18.3)	0 (0.0)	5 (8.2)
Level III	1 (1.8)	2 (3.6)	2 (4.1)	4 (19.0)	3 (14.3)	3 (16.7)	0 (0.0)	1 (16.7)	2 (33.3)	6 (8.5)	4 (6.0)	5 (8.2)
Level IV	3 (5.4)	5 (8.9)	30 (61.2)	10 (47.6)	8 (38.1)	8 (44.4)	4 (57.1)	4 (66.7)	3 (50.0)	37 (52.1)	35 (52.2)	31 (50.8)
Not applicable	49 (87.5)	44 (78.6)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)	3 (42.9)	1 (16.7)	0 (0.0)	0 (0.0)	0 (0.0)	0 (0.0)
<i>p</i> value	Across all three time points: < 0.001 , 0m - 6m: 0.041 , 6m-12m: < 0.001 , 0m - 12m: < 0.001			Across all three time points: 0.023 , 0m - 6m: 0.014 , 6m-12m:0.070, 0m - 12m: 0.414			Across all three time points: 0.368, 0m - 6m: 0.180, 6m-12m: 1.000, 0m - 12m: 0.144			Across all three time points: 0.045 , 0m - 6m: 0.011 , 6m-12m: 0.197, 0m - 12m: 0.236		
DASS 21: Depression*					(n=20)							
Mean (SD)	4.9 (4.1)	2.3 (2.6)	5.3 (2.7)	5.9 (5.7)	2.3 (1.6)	4.7 (1.6)	2.6 (2.1)	0.8 (1.6)	5.7 (1.5)	2.4 (1.4)	2.8 (2.7)	4.8 (1.9)
Median [IQR]	4.00 [2.0, 7.0]	2.0 [1.0, 3.0]	5.0 [3.0, 8.0]	4.0 [2.0, 8.0]	2.0 [1.0, 3.7]	4.0 [3.7, 6.2]	3.0 [1.0, 4.0]	0.0 [0.0, 1.7]	5.0 [4.7, 7.2]	2.0 [2.0, 3.0]	2.0 [1.0, 4.0]	5.0 [3.0, 6.0]

	Group	A aged less thar	15 years	Group A aged 5 or more years			Group B aged less than 5 years			Group B aged 5 or more years		
Time point	0m (n=56)	6m (n=56)	12m (n=49)	0m (n=21)	6m (n=21)	12m (n=18)	0m (n=7)	6m (n=6)	12m (n=6)	0m (n=71)	6m (n=67)	12m (n=61)
GMFM 66	n=56	n=56	n=49	n=21	n=21	n=18	n =7	n=6	n=6	n=71	n=67	n=61
p value	Across all three time points: <0.001, 0m - 6m:< 0.001, 6m-12m: <0.001 , 0m - 12m: 0.189			Across all three time points: 0.016 , 0m - 6m: 0.003 , 6m-12m: 0.003 , 0m - 12m: 0.943			Across all three time points: 0.094, 0m - 6m: 0.197, 6m-12m: 0.042 , 0m - 12m: 0.072			Across all three time points: <0.001 , 0m - 6m: 0.725, 6m-12m: <0.001 , 0m - 12m: <0.001		
DASS 21: Anxiety*			(n=47)		(n=20)						n=66	
Mean (SD)	4.0 (3.9)	2.7 (3.1)	7.1 (2.2)	4.2 (4.8)	2.4 (3.6)	7.1 (2.5)	0.6 (0.5)	3.2 (3.8)	6.0 (2.2)	0.9 (0.9	4.9 (3.1)	7.2 (2.2)
Median [IQR]	2.5 [1.0, 5.0]	2.0 [0.0, 4.0]	7.0 [5.0, 9.0]	3.0 [1.0, 6.0]	1.0 [0.0, 2.7]	8.0 [5.5, 9.0]	1.0 [0.0, 1.0]	2.0 [0.0, 6.7]	7.0 [4.2, 7.2]	1.0 [0.0, 1.0]	5.0 [2.0, 7.0]	7.0 [5.0, 9.0]
<i>p</i> value	Across all three time points: <0.001 , 0m - 6m: 0 .046 , 6m-12m: <0.001 , 0m - 12m: <0.001			Across all three time points: 0.004 , 0m - 6m: 0.102, 6m-12m: 0.006 , 0m - 12m: 0.093			Across all three time points: 0.022 , 0m - 6m: 0.144, 6m-12m: 0.173, 0m - 12m: 0.027			Across all three time points: <0.001 , 0m - 6m: <0.001 , 6m-12m: <0.001 , 0m - 12m: <0.001		
DASS 21: Stress*			(n=48)		(n=20)							
Mean (SD)	8.2 (3.9)	5.8 (3.7)	8.4 (2.6)	9.7 (4.1)	5.5 (3.6)	8.1 (3.1)	6.4 (3.0)	3.0 (2.8)	7.7 (1.4)	7.5 (2.5)	5.4 (2.9)	9.3 (3.0)
Median [IQR]	7.0 [5.0, 11.0]	5.0 [3.0, 8.0]	8.5 [7.0, 10.0]	9.0 [7.5, 11.5]	5.5 [3.0, 8.0]	8.5 [5.7, 11.0]	7.0 [3.0, 9.0]	2.0 [0.7, 6.2]	7.5 [6.7, 8.5]	8.0 [6.0, 9.0]	5.0 [4.0, 8.0]	9.0 [7.0, 12.0]
p value	Across all three time points: <0.001 , 0m - 6m :<0.001 , 6m-12m: <0.001 , 0m - 12m: 0.608			Across all three time points: 0.010 , 0m - 6m: 0.002 , 6m-12m: 0.058, 0m - 12m: 0.382			Across all three time points: 0.066, 0m - 6m: 0.206, 6m-12m: 0.066, 0m - 12m: 0.136			Across all three time points: <0.001 , 0m - 6m: <0.001 , 6m-12m: <0.001 , 0m - 12m: 0.009		

CFCS Level

Overall, the CFCS level deteriorated between baseline and 12 months (p=0.095). Upon disaggregation by age, no significant difference was observed in CFCS levels among children aged less than five years. Among children aged five years and above the CFCS level deteriorated significantly between 6 and 12 months (p=0.015) (Table 2 and Table 3).

VSS Level

Overall, the VSS level slightly improved among children between baseline and 6 months but then deteriorated between 6 and 12 months, however, this change between these timepoints were not statistically significant (p=0.232). When disaggregated by age, slight improvement in VSS level was observed for children aged less than five years between baseline and 6 months (p=0.180), 6 and 12 months (p=1.00) and baseline and 12 months (p=0.144). Children aged 5 years and more showed improvement in VSS level between baseline and 6 months (p=0.011), however it slightly deteriorated between 6 and 12 months (p=0.197) (Table 2 and Table 3).

Primary Caregiver DASS 21: Depression

The proportion of caregiver with depressive symptoms gradually increased across the three timepoints (p<0.001). The score on depression subscale decreased between baseline and 6 months (p=0.197) and increased between 6 and 12 months (p=0.042) among caregivers of children aged less than five years. However, the median score remained unchanged between baseline and 6 months for caregivers of children aged five years and above (p=0.725) but significantly increased between 6 and 12 months (p<0.001) (Table 2 and Table 3).

Primary Caregiver DASS 21: Anxiety

The proportion of mild to extremely severe anxiety symptoms gradually increased between baseline, 6 and 12 months (p<0.001) among primary caregivers of children. When disaggregated by age of children, the anxiety score increased between baseline and 6 months (p=0.144) and between 6 and 12 months (p=0.173) among caregivers of children aged less than five years. However, among caregivers of children aged five years and above, the median anxiety score increased between baseline and 6 months (p<0.001) and 6 and 12 months (p<0.001) (Table 2 and Table 3).

Primary Caregiver DASS 21: Stress

The mild to extremely severe symptoms of stress decreased between baseline and 6 months but increased between 6 and 12 months (p<0.001) among the primary caregivers. The median stress score decreased between baseline and 6 months for both subgroups (p=0.206 for children aged less than

five years and p<0.001 for children aged five years and above). This was followed by an increase between 6 and 12 months in both age groups (p=0.066 and p<0.001 respectively) (Table 2 and Table 3).

3.3 Mortality

Three children died during the study period: one from Group A (male, age: 7.2 years) and two from group B (female, age: 10.2 years and male, age: 7.5 years). All three children had severe motor (GMFCS Level V), communication (CFCS level V) and speech impairment (VSS level IV). All three families were living below the poverty line and residing in mud houses. All three mothers were unemployed even though two of them had higher level of education compared to their husbands who were employed.

4. Discussion

To the best of our knowledge this is one of few studies to systematically examine the outcome of a community-based early interventional and rehabilitation program for children with CP in Bangladesh and similar economy countries. The intervention in the present study incorporated several key components to address some of the barriers to intervention in low resource settings.[6,17,19-22] It is therefore scalable and easily replicable in other parts of Bangladesh and similar settings in other LMICs. The intervention is provided free of cost and accompanied by transportation service. It is delivered by community health workers, therefore, sustainable in absence of highly skilled allied health workers. The intervention concurrently aims to empower the primary caregivers, to be able to continue therapy beyond the scope of the study and better equip them for the caregiving role of their child's lifelong condition. The intervention also includes initial goal setting for each child with the primary caregivers at the time of enrollment - this is best practice for CP due to the heterogeneous nature of the condition and it enables individualization of care for optimal outcomes.[30,31]

The children who received the intervention showed significant improvement in GMFM total scores in the first six months. This increment was sustained among children who were aged less than five years. However, although not statistically significant there was a decline observed among those who were aged five years and above. In contrast, among those who did not receive intervention a decline in the median GMFM scores was observed across the three timepoints. This decline was statistically significant for children ages five years or more. However, this could be due to the difference in sample size between the two subgroups. This finding in Group A contrasts to some extent, with what has been previously published on motor function trajectories of children with CP. With the exception of GMFCS V, children with CP typically continue to increase in GMFM scores until about 5-7 years and then reach a plateau. It is later in adolescence that decline in motor function for those GMFCS III-V occurs. Studies have shown that motor function is strongly linked to the nutritional status and quality of life of children with CP in LMICs.[32,33] Findings from the Bangladesh Cerebral Palsy Register have shown that majority of children with CP in Bangladesh have severe chronic undernutrition.[32] Another study in Bangladesh reported that individuals with CP are at high risk of poor health-related quality of life and mental health problems.[33] Therefore, the intervention provided in the present study which resulted in improved motor outcomes additionally holds the potential for better nutritional outcomes and enhanced wellbeing among these children.

There was significant improvement in communication and speech among the children who received the intervention. Upon stratification by age, better outcomes were observed among children aged less than five years. This finding reinforces the importance of early intervention before five years of age. Meanwhile a different pattern was observed among the children who did not receive the intervention in Group B. Those who were under five years of age did not show any significant change in communication, while the children over five years of age deteriorated significantly over time. There was no significant change observed for speech in this group. However, interpretations should be made with caution as the sample size for children under the age of five years in this group was considerably smaller than those above five years of age (i.e. 7,6 and 6 vs 71, 67 and 61 at 0, 6 and 12 months respectively). These findings underpin the need for more trained speech therapist in countries like Bangladesh and the inclusion of speech therapy in early intervention programs.

In addition to the direct benefits for the children with CP, the intervention was seen to significantly improve the DASS scores of their primary caregivers from the intervention group. This potentially reflects the beneficial effects of peer to peer support through the formation of caregiver networks within the intervention program at CSF "Shishu Shorgo" rehabilitation centres. Having a child with CP impacts the entire family, particularly the mothers who bear the majority of the burden of caregiving. Cultural factors, misconceptions and stigma around disability add further to the plight of the mothers. Furthermore, 86% and 82% of mothers in group A and B were unemployed respectively. According to the World Bank data female unemployment rate in Bangladesh is ~6%. This marked difference is most likely owing to their primary caregiver role to their children with CP.

A study in Bangladesh reported that primary caregivers of adolescents with CP are at significantly higher risk of depression and stress compared to the caregivers of their peers without CP.(11) In recent times, there has been an increasing shift towards greater participation of primary caregivers in therapy for their children.(31, 32) Studies reported improved child and caregiver outcomes through this approach. However, caregiver mental wellbeing is crucial for them to effectively deliver such interventions.(33) Several studies showed that interventions for children with CP can significantly improve parental wellbeing.(31) Our findings support the recommendations from a previous study conducted in Bangladesh which highlights the importance of supporting caregiver mental wellbeing when designing interventions for individuals with CP.(11) Furthermore, interventions including holistic measures for poverty alleviation and improvement of social and economic capital of the families can also potentially yield better functional outcomes for children with CP.(34)

Early diagnosis and intervention improve outcomes, particularly among children with mild CP.(12) However, children with severe CP, who also tend to have more severe associated impairments, may make smaller gains with intervention . Population-based surveillance in Bangladesh have found increased vulnerability to mortality with greater motor severity and more associated impairments.(35) This is consistent with the findings in our study where all three children who died during the study period had severe gross motor, communication and speech impairment. Moreover, all three families were living in impoverished conditions and their family income was below the poverty line. These findings highlight the relationship between health outcomes and social determinants of health and the importance of addressing the prevailing inequities to reduce needless deaths among children with CP.

Study limitations

One of the major limitations of the study was that Group A and the Group B were significantly different in terms of age (mean age: 3.7 vs 9.6 years) which inhibited the statistical analysis; there was also variation in the baseline status of their motor and communication functions. These factors may have collectively influenced the study findings. There was also bias introduced due to the method of recruitment of study participants. The intervention was offered to all children with CP in need of therapy identified through the BCPR during the study period. Those who were able to take-up the provided intervention formed Group A, whereas, the families that were unable to take this opportunity due to personal or family circumstances formed Group B. Therefore, interpretation of the differences in outcome between the two groups should be made with caution considering that there are several differences between the children who received the intervention and those who did not including their age and severity of impairments at baseline. Yet, Group A and Group B did not show significant differences across most demographic characteristics and both groups showed short term improvement in motor outcomes.

The improvement among the children in Group B could possibly be due to the primary caregiver education provided at the time of registration in the BCPR; the registered children are also ensured access to assistive devices and have access to weekly physiotherapy clinics in the surveillance area. It could additionally be due to attrition bias as some of the more severe cases were lost to follow up or died during the study period. Moreover, the heterogeneous nature of CP poses a challenge in the application and evaluation of interventions. Further exploration through large scale randomized controlled trial using the BCPR as the sampling frame to evaluate whether the beneficial effects of the intervention are sustained over long-term can generate more robust population level evidence.

Despite these limitations, this study was based on a population-based cohort (unlike institutional cohort subject to selection bias) and therefore represents the true nature and likely outcomes of community based and intervention programs in LMICs. As a pragmatic quasi experimental study, findings from this study including our reported limitations can inform clinicians and researchers in developing more holistic programs and welldesigned experimental studies including randomized controlled trials.

5. Conclusions

The intervention in the present study has resulted in promising outcomes, particularly for young children with CP under five years of age. There is a need for such caregiver led community-based rehabilitation program for children with CP in LMICs such as Bangladesh, particularly in the rural settings, where services and trained health workers are scarce. This will ensure access to services to one of the most marginalized populations and promote optimal use of the limited resources. This work can potentially underpin the development of a sustainable model for rehabilitation of children with CP which can be scaled up in low resource settings.

Author Contributions: Conceptualization, GK and MM; methodology, GK; software, CM, TK; formal analysis, TK, IJ, CG; investigation, TK; resources, GK, TK; data curation, GK, TK; writing—original draft preparation, TK; writing—review and editing, GK, NB, MM, CM, HSS, CG, TK, IJ.; visualization, GK, NB, CM, CG, TK, IJ; supervision, GK,NB; funding acquisition, GK, NB, MM. All authors have read and agreed to the published version of the manuscript.

Funding: This study has been conducted as part of the BCPR. BCPR is funded by the Research Foundation of Cerebral Palsy Alliance (PG4314, PG16917), Sydney Medical School Foundation and internal funding from CSF Global, Bangladesh. The funding body played no role in the design of the study and collection, analysis, and interpretation of data and in the preparation of the manuscript. TK is supported by the Research Foundation of Cerebral Palsy Alliance (CDG04617, PHD02119). HSS received salary support through a National Health and Medical Research Council of Australia Early Career Fellowship (1144566) and Australasian Cerebral Palsy Clinical Trials Network. **Institutional Review Board Statement:** The study was conducted according to the guidelines of the Declaration of Helsinki and approved by the Bangladesh Medical Research Council (BMRC) (Ref: BMRC/NREC/2016-2019/469) and Asian Institute of Disability and Development (AIDD) (southasia-irb- 2016-1-07).

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Data is available upon reasonable request from the corresponding author.

Acknowledgments: We would like to acknowledge the CSF Global team in Bangladesh for their support in the implementation of the project and for supporting the children with CP and their families in access to services through a strong referral system.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Rosenbaum, P.; Paneth, N.; Leviton, A.; Goldstein, M.; Bax, M.; Damiano, D.; Dan, B.; Fabiola, R.; Jacobsson, B. A report: the definition and classification of cerebral palsy April 2006. *Developmental Medicine & Child Neurology Suppl.* 2007, 49, 8-14, doi:10.1111/j.1469-8749.2007.tb12610.x.
- 2. Ahlin, K.; Himmelmann, K.; Nilsson, S.; Sengpiel, V.; Jacobsson, B. Antecedents of cerebral palsy according to severity of motor impairment. *Acta obstetricia et gynecologica Scandinavica* **2016**, *95*, 793-802.
- McIntyre, S.; Taitz, D.; Keogh, J.; Goldsmith, S.; Badawi, N.; Blair, E. A systematic review of risk factors for cerebral palsy in children born at term in developed countries. *Dev Med Child Neurol* 2013, 55, 499-508, doi:10.1111/dmcn.12017.
- 4. Korzeniewski, S.J.; Birbeck, G.; DeLano, M.C.; Potchen, M.J.; Paneth, N. A systematic review of neuroimaging for cerebral palsy. *Journal of child neurology* **2008**, 23, 216-227.
- Mesterman, R.; Leitner, Y.; Yifat, R.; Gilutz, G.; Levi-Hakeini, O.; Bitchonsky, O.; Rosenbaum, P.; Harel, S. Cerebral palsy—long-term medical, functional, educational, and psychosocial outcomes. *Journal of child neurology* 2010, 25, 36-42.
- 6. Khandaker, G.; Muhit, M.; Karim, T.; Smithers-Sheedy, H.; Novak, I.; Jones, C.; Badawi, N. Epidemiology of cerebral palsy in Bangladesh: a population-based surveillance study. *Dev Med Child Neurol* **2019**, *61*, 601-609, doi:10.1111/dmcn.14013.
- 7. Korzeniewski, S.J.; Slaughter, J.; Lenski, M.; Haak, P.; Paneth, N. The complex aetiology of cerebral palsy. *Nature Reviews Neurology* **2018**, *14*, 528-543.
- 8. Jahan, I.; Muhit, M.; Hardianto, D.; Laryea, F.; Chhetri, A.B.; Smithers-Sheedy, H.; McIntyre, S.; Badawi, N.; Khandaker, G. Epidemiology of cerebral palsy in low-and middle-income countries: preliminary findings from an international multi-centre cerebral palsy register. *Developmental Medicine & Child Neurology* 2021.

- Novak, I.; Morgan, C.; Adde, L.; Blackman, J.; Boyd, R.N.; Brunstrom-Hernandez, J.; Cioni, G.; Damiano, D.; Darrah, J.; Eliasson, A.C.; et al. Early, Accurate Diagnosis and Early Intervention in Cerebral Palsy: Advances in Diagnosis and Treatment. *JAMA Pediatr* 2017, 171, 897-907, doi:10.1001/jamapediatrics.2017.1689.
- 10. Ulrich, B.D. Opportunities for early intervention based on theory, basic neuroscience, and clinical science. *Physical therapy* **2010**, *90*, 1868-1880.
- 11. McIntyre, S.; Morgan, C.; Walker, K.; Novak, I. Cerebral palsy-don't delay. *Developmental disabilities research reviews* **2011**, *17*, 114-129.
- Rosenbaum, P.L.; Walter, S.D.; Hanna, S.E.; Palisano, R.J.; Russell, D.J.; Raina, P.; Wood, E.; Bartlett, D.J.; Galuppi, B.E. Prognosis for gross motor function in cerebral palsy: creation of motor development curves. *Jama* 2002, *288*, 1357-1363.
- 13. Morgan, C.; Fetters, L.; Adde, L.; Badawi, N.; Bancale, A.; Boyd, R.N.; Chorna, O.; Cioni, G.; Damiano, D.L.; Darrah, J. Early Intervention for Children Aged 0 to 2 Years With or at High Risk of Cerebral Palsy: International Clinical Practice Guideline Based on Systematic Reviews. *JAMA pediatrics* **2021**.
- 14. Power, R.; Muhit, M.; Heanoy, E.; Karim, T.; Galea, C.; Badawi, N.; Khandaker, G. Depression, anxiety and stress among caregivers of adolescents with cerebral palsy in rural Bangladesh. *Disability and Rehabilitation* **2019**, 1-8.
- 15. Polack, S.; Adams, M.; O'Banion, D.; Baltussen, M.; Asante, S.; Kerac, M.; Gladstone, M.; Zuurmond, M. Children with cerebral palsy in Ghana: malnutrition, feeding challenges, and caregiver quality of life. *Dev Med Child Neurol* **2018**, *60*, 914-921, doi:10.1111/dmcn.13797.
- 16. Dambi, J.M.; Mlambo, T.; Jelsma, J. Caring for a child with Cerebral Palsy: The experience of Zimbabwean mothers. *African journal of disability* **2015**, *4*, 1-10.
- 17. Maulik, P.K.; Darmstadt, G.L. Childhood disability in low-and middle-income countries: overview of screening, prevention, services, legislation, and epidemiology. *Pediatrics* **2007**, *120*, S1-S55.
- Novak, I.; Morgan, C.; Fahey, M.; Finch-Edmondson, M.; Galea, C.; Hines, A.; Langdon, K.; Namara, M.M.; Paton, M.C.; Popat, H.; et al. State of the Evidence Traffic Lights 2019: Systematic Review of Interventions for Preventing and Treating Children with Cerebral Palsy. *Curr Neurol Neurosci Rep* 2020, 20, 3, doi:10.1007/s11910-020-1022-z.
- 19. Nuri, R.P.; Ghahari, S.; Aldersey, H.M.; Huque, A.S. Exploring access to government-led support for children with disabilities in Bangladesh. *PLOS ONE* **2020**, *15*, e0235439, doi:10.1371/journal.pone.0235439.
- 20. Ahmed, S.M.; Hossain, M.A.; RajaChowdhury, A.M.; Bhuiya, A.U. The health workforce crisis in Bangladesh: shortage, inappropriate skill-mix and inequitable distribution. *Human resources for health* **2011**, *9*, 1-7.
- 21. Spittle, A.J.; Morgan, C. Early intervention for children with cerebral palsy. In *Cerebral palsy*; Springer: 2018; pp. 193-200.
- 22. Al Imam, M.H.; Jahan, I.; Das, M.C.; Muhit, M.; Smithers-Sheedy, H.; McIntyre, S.; Badawi, N.; Khandaker, G. Rehabilitation status of children with cerebral palsy in Bangladesh: Findings from the Bangladesh Cerebral Palsy Register. *Plos one* **2021**, *16*, e0250640.
- 23. Khandaker, G.; Smithers-Sheedy, H.; Islam, J.; Alam, M.; Jung, J.; Novak, I.; Booy, R.; Jones, C.; Badawi, N.; Muhit, M. Bangladesh Cerebral Palsy Register (BCPR): a pilot study to develop a national cerebral palsy (CP) register with surveillance of children for CP. *BMC neurology* **2015**, *15*, 173.
- Palisano, R.J.; Rosenbaum, P.; Bartlett, D.; Livingston, M.H. Content validity of the expanded and revised Gross Motor Function Classification System. *Developmental Medicine & Child Neurology* 2008, 50, 744-750.
- 25. Alotaibi, M.; Long, T.; Kennedy, E.; Bavishi, S. The efficacy of GMFM-88 and GMFM-66 to detect changes in gross motor function in children with cerebral palsy (CP): a literature review. *Disability and rehabilitation* **2014**, *36*, 617-627.
- 26. Russell, D.; Rosenbaum, P.; Wright, M.; Avery, L. Measure (GMFM-66 & GMFM-88) User's Manual, London, UK: Mac Keith Press, 2013. 290 pp. \$115.00. ISBN: 978-1908316882.
- Hidecker, M.J.C.; Paneth, N.; Rosenbaum, P.L.; Kent, R.D.; Lillie, J.; Eulenberg, J.B.; CHESTER, J., KEN; Johnson, B.; Michalsen, L.; Evatt, M. Developing and validating the Communication Function Classification System for individuals with cerebral palsy. *Developmental Medicine & Child Neurology* 2011, 53, 704-710.
- Pennington, L.; Virella, D.; Mjoen, T.; da Graca Andrada, M.; Murray, J.; Colver, A.; Himmelmann, K.; Rackauskaite, G.; Greitane, A.; Prasauskiene, A.; et al. Development of The Viking Speech Scale to classify the speech of children with cerebral palsy. *Res Dev Disabil* 2013, 34, 3202-3210, doi:10.1016/j.ridd.2013.06.035.
- 29. Lovibond, P.F.; Lovibond, S.H. The structure of negative emotional states: Comparison of the Depression Anxiety Stress Scales (DASS) with the Beck Depression and Anxiety Inventories. *Behaviour research and therapy* **1995**, *33*, 335-343.

4.3 CHAPTER SYNOPSIS

In this chapter, the outcome of a community-based parent led early intervention program was assessed. The intervention demonstrated promising outcomes, particularly for children with CP under five years of age. However, further research is needed to assess the broader implications for scale in Bangladesh and other low resource settings.

The first four chapters of this thesis have involved an exploration of epidemiology, early detection, and early intervention in a LMIC, Bangladesh. This research has used the BCPR as a data resource and a sampling frame for clinical research. However, in most LMIC regions, there are no equivalent CP registers to allow enhanced understanding of the clinical characteristics and inform the development of intervention and rehabilitation programs. Therefore, in the following chapter of this thesis, we shift our focus to an alternative method for understanding children with CP. A hospital-based surveillance program was instituted in Vietnam. In chapter 4, results from this program regarding the clinical characteristics and nutritional status of children with CP in Vietnam will be described.

CHAPTER 5. HOSPITAL-BASED SURVEILLANCE OF CHILDREN WITH CEREBRAL PALSY IN VIETNAM





5.1 INTRODUCTION

The BCPR demonstrated what can be achieved with a CP register in a LMIC. Surveillance for children with CP is imperative in other LMICs where similar research and service gaps exist.

Upon commencement of my doctoral studies in July 2018, I relocated to Sydney. I had the opportunity to work with Professor Elizabeth Elliott and a team in Vietnam. This enabled the extension of my lessons and experience from the BCPR study to Vietnam. This new work would help us support the establishment of CP registers in multiple LMICs which could then collaborate with each other and with those from HICs.

A hospital-based surveillance of children with CP in Vietnam is described in this chapter. The protocol for the surveillance has been previously published.(42) This study enabled exploration of the clinical characteristics, etiology, rehabilitation, educational and nutritional status of children with CP in Vietnam. These findings have been described in the following publications.

5.2 PAPER #5 DATA ON CEREBRAL PALSY IN VIETNAM: FINDINGS FROM PROSPECTIVE HOSPITAL-BASED SURVEILLANCE IN VIETNAM

5.3 Paper #6. NUTRITIONAL STATUS OF CHILDREN WITH CEREBRAL PALSY: FINDINGS FROM PROSPECTIVE HOSPITAL-BASED SURVEILLANCE IN VIETNAM



ISSN: (Print) (Online) Journal homepage: https://www.tandfonline.com/loi/idre20

Data on cerebral palsy in Vietnam will inform clinical practice and policy in low and middleincome countries

Tasneem Karim , Rachael Dossetor , Nguyen Thi Huong Giang , Trinh Quang Dung , Tran Vinh Son , Nguyen Xuan Hoa , Nguyen Hong Tuyet , Nguyen Thi Van Anh , Cao Minh Chau , Nguyen Van Bang , Nadia Badawi , Gulam Khandaker & Elizabeth Elliott

To cite this article: Tasneem Karim , Rachael Dossetor , Nguyen Thi Huong Giang , Trinh Quang Dung, Tran Vinh Son, Nguyen Xuan Hoa, Nguyen Hong Tuyet, Nguyen Thi Van Anh, Cao Minh Chau, Nguyen Van Bang, Nadia Badawi, Gulam Khandaker & Elizabeth Elliott (2021): Data on cerebral palsy in Vietnam will inform clinical practice and policy in low and middle-income countries, Disability and Rehabilitation, DOI: 10.1080/09638288.2020.1854872

To link to this article: https://doi.org/10.1080/09638288.2020.1854872



Published online: 04 Jan 2021.

|--|

Submit your article to this journal 🗹

Article views: 30

View related articles 🗹



則 🛛 View Crossmark data 🗹

ORIGINAL ARTICLE

Check for updates

Taylor & Francis

Taylor & Francis Group

Data on cerebral palsy in Vietnam will inform clinical practice and policy in low and middle-income countries

Tasneem Karim^{a,b,c,d}* (**b**), Rachael Dossetor^a*, Nguyen Thi Huong Giang^e, Trinh Quang Dung^e, Tran Vinh Son^e, Nguyen Xuan Hoa^e, Nguyen Hong Tuyet^e, Nguyen Thi Van Anh^f, Cao Minh Chau^{g,h}, Nguyen Van Bang^{i,j}, Nadia Badawi^{a,d}, Gulam Khandaker^{a,b,c,k}; and Elizabeth Elliott^{a,l}

^aDiscipline of Child and Adolescent Health, Faculty of Medicine and Health, The University of Sydney, Sydney, Australia; ^bCSF Global, Dhaka, Bangladesh; ^cAsian Institute of Disability and Development (AIDD), University of South Asia, Dhaka, Bangladesh; ^dCerebral Palsy Alliance Research Institute, Specialty of Child & Adolescent Health, Sydney Medical School, Faculty of Medicine & Health, The University of Sydney, Sydney, Australia; ^eRehabilitation Department, National Children's Hospital, Hanoi, Vietnam; ^fMedical Education and Skills-Lab Department, Hanoi Medical University, Hanoi, Vietnam; ^gRehabilitation Department, Hanoi Medical University (Bach Mai Hospital), Hanoi, Vietnam; ^hFaculty of Health Sciences, Phenikaa University, Hanoi, Vietnam; ⁱDepartment of Paediatrics, Hanoi Medical University, Hanoi, Vietnam; ^jFaculty of Medicine, Hanoi University of Business and Technology, Hanoi, Vietnam; ^kCentral Queensland Public Health Unit, Central Queensland Hospital and Health Service, Rockhampton, Australia; ^lKid's Research, The Sydney Children's Hospitals Network (Westmead), Sydney, Australia

ABSTRACT

Purpose: To document known risk factors, clinical severity, associated impairments and rehabilitation status of children presenting with cerebral palsy (CP) to the National Children's Hospital (NCH) in Hanoi, Vietnam.

Materials and methods: Active prospective ascertainment of cases of CP presenting between June and November 2017 to the NCH using surveillance modelled on the Paediatric Active Enhanced Disease Surveillance system in Australia.

Results: Data were collected on 765 children with CP (mean age: 2 years 7 months (SD 2 y 6 mo). Mean age at diagnosis was 1 year 8 months (SD 1 y 9 mo). Children predominantly had spastic CP (95.2%, n=729), most were quadriplegic (69.6%, n=532) and 60.3% (n=454) were Gross Motor Functional Classification System level III-V. Of the children 76.2% (n=583) had one/more associated impairments. 36.3% (n=276) had presumed perinatal asphyxia, 26.5% (n=202) were preterm. Physiotherapy (94.3%, n=663) was the most common form of intervention used. Only 2.6% (n=12) of the children who would have benefitted from assistive devices had wheelchairs.

Conclusion: We established hospital-based surveillance of CP in Hanoi and confirmed a high burden and severity of CP with potentially preventable risk factors. These data will inform clinician training and health policy and identify need for evidence-based care and assistive devices.

► IMPLICATIONS FOR REHABILITATION

- We identified a high number of children with severe forms of cerebral palsy (CP) in Hanoi, Vietnam through hospital-based surveillance.
- There is an urgent need for clinician training and access to and use of evidence-based interventions including assistive technology.
- This study will inform local capacity building and health policy for improved diagnosis and care of children with CP in Vietnam and other low and middle-income countries.

Introduction

Cerebral palsy (CP) is a group of disorders resulting from an insult to the developing brain and characterised by permanently disordered movement, co-ordination, balance or posture. Children with CP often have associated impairments including problems with development, vision, speech, cognition, epilepsy and feeding. CP is the leading cause of physical disability and data from highincome countries (HICs) show that the pooled prevalence is 2.9 per 1000 children, CP affects 2.1 per 1000 live births [1]. The prevalence of CP in Europe decreased from 1.9 to 1.8 per 1000 live births between 1980 to 2003 [2]. Recent findings from the Australian CP Register (ACPR) showed a decline in rates from 2.1 in 1995–1997 to 1.4 children per 1000 live births/neonatal survivors in the 2010–2012 period [3]. The decline in CP rates and severity in the HICs is the cumulative result of the impact advances in maternal and child health care and robust public health initiatives, guided by empirical evidence [3]. Furthermore, there are probably additional factors affecting CP prevalence that are not

ARTICLE HISTORY

Received 10 January 2020 Revised 18 November 2020 Accepted 19 November 2020

KEYWORDS

Cerebral palsy; children; register; surveillance; Vietnam

CONTACT Tasneem Karim staneem.karim.tk@gmail.com stance.bicipline of Child and Adolescent Health, Faculty of Medicine and Health, The University of Sydney, Level 3, The Children's Hospital at Westmead, Corner Hawkesbury Road and Hainsworth Street, Sydney, NSW, 2145, Australia *Tasneem Karim and Rachael Dossetor are joint first authors on this paper.

This article has been corrected with minor changes. These changes do not impact the academic content of the article.

yet identified. Neuroprotective approaches (e.g., administration of magnesium sulphate to mothers at risk of a preterm birth) and innovation in perinatal practice (e.g., cooling or induced mild hypothermia to reduce the impact of brain injury) are the key to the reducing the burden of CP, particularly due to known preventable risk factors, and multidisciplinary care is crucial to improving health and quality of life [4,5]. These interventions are often unavailable in low and middle-income countries (LMICs). Data are limited, particularly on trends, and there is no evidence of declining rates of CP in LMICs.

Service delivery for CP is seen as a public health priority by the Ministry of Health in Vietnam, however the lack of epidemiological data limits an evidence-based approach [6]. There is an urgent need for surveillance systems to collect local data to inform diagnoses, intervention and health policies in Vietnam. Furthermore, comprehensive pooled data from LMICs will inform regionally applicable strategies and models of care for children with CP and ensure optimal use of limited resources.

We aimed to identify children with CP at the National Children's Hospital (NCH), describe their demographic data and the aetiology, motor function, clinical severity, associated impairments, and rehabilitation status associated with CP. These data will form the basis of a national CP register in Vietnam.

Materials and methods

We established a method for active, hospital-based surveillance to identify children with CP, modelled on the Paediatric Active Enhanced Disease Surveillance (PAEDS) system operating in Australia [7]. Active, prospective case finding involves identification of cases on a daily basis as they come to the National Children's Hospital. Dedicated, trained clinicians at the hospital are responsible for actively seeking children with CP and collecting data using a standardised record form following thorough clinical assessment, history taking and review of existing medical records. Additional methodological details can be found in our study protocol and other publications [6,7].

Study site and participants

NCH is an 1800 bed tertiary paediatric hospital in Hanoi that provides services for about 100 000 inpatients and 1 million outpatients each year. The rehabilitation department sees approximately 80 to 100 children per day.

Baseline case identification by Vietnamese paediatricians took place between June and November 2017. Participants were children aged less than 18 years, who attended the rehabilitation department at NCH and were diagnosed with CP, according the definition used by the Surveillance of Cerebral Palsy in Europe and the Australian Cerebral Palsy Register (ACPR) [2,3]. Diagnosis of CP was confirmed by the paediatricians based on clinical assessment, history taking and review of existing hospital records and investigations using our study protocol and hospital guidelines. The definition, inclusion and exclusion criteria used for CP are described in the protocol for this study [6].

Detailed assessment and data collection

The questionnaire used for data collection is a modified version of the ACPR record form which was used for the Bangladesh CP Register (BCPR) Study [3,5]. The clinical information was collected by trained Vietnamese paediatricians at the National Children's Hospital. Some information provided in the questionnaire came from existing medical records. Responses were recorded in Vietnamese and translated into English by two of the investigators. Translated data were entered into a database by study investigators.

Sociodemographic details were collected. Families living on an average income of less than or equal to 653 000 Vietnam Dong (VND) per person, per month were defined as living below the poverty line [8].

Data were collected on the perinatal period as was clinical information including the age of CP diagnosis, presumed timing of injury causing CP and type of CP. The Gross Motor Function Classification System (GMFCS) and Manual Ability Classification System (MACS) were used to assess functional severity. MACS was determined for children over 4 years of age [9,10]. MRI scans were performed in some children and anthropometric measurements were documented to assess the nutritional status of the children and these data have been reported separately.

The presence and severity of associated impairments were documented based on clinical history, detailed assessment conducted by clinicians and report by the parents/primary caregivers.

Diagnosis of epilepsy was based on a clinical history of one or more unprovoked seizures in the previous 3 months beyond the neonatal period. An EEG assessment was done, clinical history was taken and relevant medical records were reviewed.

Intellectual impairment or delayed cognitive development was determined by clinicians based on clinical history and assessment or report by the parents/primary caregivers or review of medical records.

Visual impairment was determined through history, available medical records and clinical assessment of visual acuity and functional vision including counting fingers, hand motion and light perception. The monocular cover-uncover test was conducted to identify strabismus.

Assessment of hearing impairment was based on hearingrelated medical history, including the child's response to namecall, clap and vehicle horns; previous ear infections, symptoms of ear pain or purulent drainage; family history; prescription and use of over-the-counter ototoxic medication; and previous hearing loss interventions. Otoscopic examination was performed to evaluate the pinna, external auditory canal, and tympanic membrane for any conditions that could be contributing to hearing loss or that may require further evaluation and treatment. Audiology was not performed.

Assessment of speech was based on the history taken from the primary caregiver and speech and language assessment by a medical practitioner. Expressive and respective language, naming guality, and guality of conversational speech were also observed.

Perinatal asphyxia was defined as a newborn failing to cry at the time of birth, experiencing delayed onset of breathing (>1min), or requiring assistance to initiate breathing.

Early feeding difficulty was recorded when a neonate was unable to feed in the first month of life due to poor sucking abilities.

Ethical considerations

Ethics approval was obtained from the University of Sydney Human Research Ethics Committee (HREC) (2016/456), Hanoi Medical University (HMU) (1722/QD-QHYHN) and the NCH in Hanoi (812/QD-BVNTU). Informed consent was obtained from parents or caregivers of all the study participants.

Statistical analysis

Data were checked for missing data and cleaned. We performed descriptive analyses (mean, standard deviation (SD), median and proportion). All analyses were performed using SPSS Statistics software version 24 (IBM Corporation, Chicago, Illinois, USA).

Results

Data were collected between June and November 2017 on 765 children with a clinical diagnosis of CP. The mean age of participants was 2 years 7 months (SD 2 y 6 mo; range: 2.4 mo-13 y 5 mo), 64.2% were male (n = 491). The mean age at diagnosis of CP was 1 year 8 months (SD 1 y 9 mo).

Of the children, 90.2% (n = 533) were from the majority ethnic group in Vietnam, Kinh. Most (94.4%, n = 711) of the cohort were living in houses with finished floors. Sources of drinking water predominantly included piped water (55.6%, n = 423) and well water (37.3%, n = 284). Most (83.4%, n = 633) had a flushing toilet. The mean monthly income of families was 113.2 USD (SD 106.0 USD; range: 4.4–1565.2 USD). In total, 68.8% (n = 501) of families were living below the poverty line.

The mean maternal age was 26 years 10 months (SD 5 y 2.2 mo; range: 12 y 7.2 mo-45y 10.8 mo). Over a third of mothers (42.4%, n = 323) and fathers (40.5%, n = 301) had completed higher secondary education. However, 35.0% (n = 266) of mothers and 24.5% (n = 183) of fathers were unemployed. Detailed demographic characteristics are outlined in Table 1.

CP motor type and severity

The majority of children (95.2%, n = 729) were diagnosed with spastic CP, of whom 73.0% (n = 532) had quadriplegia, 21.5% (n = 157) had monoplegia/hemiplegia and 5.5% (n = 40) had diplegia. According to GMFCS, 60.3% (n = 454) were level III-V. Among these children who would have benefitted from assistive devices only 2.6% (n = 12) of them had access to a wheelchair. Of the 150 children over 4 years of age 22.0% (n = 33) had MACS level III-V (Table 2).

Associated impairments

Of the children, 76.3% (n = 583) had one or more associated impairments: 59.1% (n = 450) had speech impairment recorded by clinicians (including 258 children who were non-verbal), 57.8% (n = 439) had intellectual impairment, 1.7% (n = 13) had hearing impairment (including 3 children with bilateral deafness), 5.5% (n = 41) had visual impairment (including one child with functional blindness) and 10.8% (n = 82) had epilepsy (Table 2).

Risk factors

Over half (55.2%, n = 416) of the children had a prenatal or perinatal risk factor reported, 13.0% (n = 98) had a postnatal risk factor and the timing of the brain injury was unknown for 31.7% (n = 239). Potentially preventable risk factors including presumed perinatal asphyxia (36.3%, n = 276), neonatal jaundice (10.7%, n = 81), infection (9.1%, n = 69), brain haemorrhage (1.4%, n = 11) and traumatic/accidental brain injury (1.4%, n = 11) were reported (Table 3).

Of the mothers, 1.8% (n = 13) had assisted conception, 86.5% (n = 652) received regular antenatal care and 4.1% (n = 31) did not receive any antenatal care. Maternal complications during childbirth were reported in 23.1% (n = 170). The majority (90.8%,

n = 687) of children were born in hospital and 66.7% (n = 504) of births were by spontaneous vaginal delivery (Table 3). Of the 4.2% (n = 31) multiple births, 8 resulted from assisted conception (5 *in vitro* fertilisations, 1 artificial insemination, 1 gamete intrafallopian transfer and 1 ovulation stimulation).

The mean GA at birth was 37.2 weeks (SD 4.5wks; range: 25–43wks): 26.5% (n = 202) were born preterm (<37wks) of whom 37.6% (n = 76) were very preterm (28–31wks). The mean birthweight was 2783.0 grams (SD 708.0 g) and 26.3% (n = 200) of children were low birthweight (<2500 g). Of the 168 children 22.6% had early feeding difficulties.

In total, 85.5% (n=460) of the cohort required a hospital transfer in the neonatal period: 27.8% (n=128) were transferred to the only national central level hospital i.e., NCH, 49.1% (n=226) to a provincial level hospital and 23.0% (n=106) to a district level hospital. Over half (53.7%, n=411) of the children received only routine care, however, 31.6% (n=242) were treated in the neonatal intensive care unit and 5.6% (n=43) received special care (Table 3).

Birth defects were reported in 14.8% (n = 112) including cardiovascular defects in 29 children. An MRI head scan was performed in 56.1% (n = 424) and reports were available for 36.1% (n = 273). A variety of structural anomalies and insults were identified. These data are currently being analysed and will be reported in detail elsewhere.

Immunisation

Of the children, 82.7% (n = 623) were fully vaccinated as per the national immunization schedule recommended for children in Vietnam and 94.0% (n = 701) had a BCG vaccine mark [11]. Of the children over 9 months of age and therefore eligible, 72.9% (n = 373) received the Measles-Rubella (MR) vaccine during the 2015 National MR campaign (Table 4).

Education and rehabilitation

Of the 75 children over 6 years of age who are eligible to attend school, 30.7% (n=23) do not attend school and only 1.2% (n=9) attend a special school. All the parents of children who received no form of rehabilitation reported that they were unaware of available therapies. Among the 92.6% (n=703) who received rehabilitation, 94.3% (n=663) received physiotherapy (however, no standard early intervention protocol was followed), 21.5% (n=151) received advice and 0.7% (n=5) underwent surgery. Only 11.5% (n=81) had used an assistive device which rarely included a wheelchair or walker. Five children received treatment with muscle relaxants, and one received acupuncture (Table 4).

Discussion

We established an ongoing hospital-based surveillance system enabling collection of robust data which can guide health service planning, capacity building of health professionals and family support. The PAEDS mechanism is efficient, low cost and suitable for a hospital such as NCH [7]. It could be scaled up to hospitals across Vietnam and is the first step towards the establishment of a national CP register. It could additionally be used to monitor other conditions of public health importance as demonstrated in Australia [7,12].

Global collaborative initiatives in cerebral palsy research have been established in HICs. For example, the Surveillance of Cerebral Palsy in Europe was established in 1998 as a

Table 1. Demographic characteristics and living conditions of children and families.

Characteristics		n (%)
Ethnic group ($n = 591$)		
Majority group Kinh		533 (90.2)
Minority group Khac		58 (9.8)
Region (<i>n</i> = 756)		
Red River Delta		421 (55.7)
Northeast		146 (19.3)
North Central		115 (15.2)
Northwest		65 (8.6)
Central Highlands		6 (0.8)
Soulhedsi Makang Biyar Dalta		2 (0.3)
South Control Coast		1(0.1)
Mean narental age at time of hirth (years) and range		0 (0.0)
Mother ($n = 764$)		269+52.126-459
Eather $(n = 75)$		30.2 + 5.8; 16.2-56.9
Education level of mother ($n = 762$)		
Illiterate		2 (0.3)
Primary		36 (4.7)
Secondary/higher secondary		518 (68.0)
Graduate/post-graduate		130 (17.1)
Diploma/other trade qualification		76 (10.0)
Education level of father ($n = 743$)		
Illiterate		1 (0.1)
Primary		35 (4.7)
Secondary/higher secondary		522 (70.2)
Graduate/post-graduate		128 (17.2)
Diploma/other trade qualification		57 (7.7)
Primary carer $(n = 596)$		405 (82.1)
Faleni(s) Grandparent(s)		495 (83.1) 62 (10.4)
Parent(s) and grandparent(s)		37 (6 2)
Other		2 (0.3)
Mean (+ SD) number of household members and range		4.7 + 1.3; 2 - 10
Mean $(\pm SD)$ number of rooms per house and range		2.5 ± 1.2 ; 1–10
Source of drinking water $(n = 761)$		
Piped water		423 (55.6)
Well water		284 (37.3)
Other sources (pond, river, lake)		54 (7.1)
Sanitation ($n = 759$)		
Flush toilet		633 (83.4)
Pit toilet		126 (16.6)
Flooring type $(n = 753)$		
Finished floor		711 (94.4)
Rough wood/bamboo		29 (3.9)
Edrin/Sdnu Monthly family income (n. 728)		13 (1.7)
Nonthing family income $(n = 726)$		
Mean + SD		2 603 287 2 + 2 436 897 5
Range		100 000-36 000 000
USD		
Mean ± SD		113.2 ± 106.0
Range		4.4–1565.2
Families living below the poverty line		501 (68.8)
		501 (00.0)
		rather (<i>I</i> = 740)
Unemployed	266 (35.0)	183 (24.5)
Elementary occupations	244 (32.1)	2/9 (3/.4)
Protessional	148 (19.5)	94 (12.6)
Service and sales workers Craft and related trade workers	27 (3.8) 23 (2.0)	(5.2) 40 (0.2) 52
Clerical support workers	23 (3.0) 15 (2.0)	22 (2.9) 17 (1 6)
Managers	15 (2.0)	1 (0.1)
Armed forces occupations	4 (0.5)	6 (0.8)
Plant and machine operators and assemblers	1 (0.1)	17 (2.3)
Technicians and associate professionals	0 (0.0)	52 (7.0)
Other	15 (2.0)	11 (1.5)

collaboration of professionals and researchers working with CP registries across Europe [2]. Such initiatives have only recently been explored in LMIC. In this paper we cite evidence from Bangladesh and Indonesia collected by the authors of this paper [5], which also use a modified version of the ACPR record form

for data collection, enabling comparison of study findings. We will also be able to compare data collected in our study with that collected through the Surveillance of Cerebral Palsy in Europe [2]. However, comparisons must be made with caution due to disparities resulting from methodological differences.

	n (%)
Sex (n = 765)	
Male	491 (64.2)
Mean (± SD) age of assessment at NCH (years)	2.6 ± 2.5
Age group ($n = 765$)	
<1	201 (26.3)
1 to <2	224 (29.3)
2 to <5	219 (28.6)
5 to <10	101 (13.2)
10 to <15	20 (2.6)
Mean (\pm SD) age of diagnosis of CP (years)	1.7 ± 1.8
Presumed timing of injury causing CP $(n = 753)$	
Pre and perinatal	416 (55.2)
Postnatal	98 (13.0)
Unknown	239 (31.7)
Type of CP ($n = 765$)	. ,
Spastic	729 (95.3)
Ouadriplegia	532 (69.6)
Monoplegia/hemiplegia	157 (20.5)
Diplegia	40 (5.2)
Triplegia	_
Dyskinetic	35 (4.6)
Ataxic	1 (0.1)
Hypotonic	_
GMFCS level $(n = 754)$	
	179 (23.7)
II.	121 (16.0)
III	103 (13.7)
IV	131 (17.4)
V	220 (29.2)
MACS level $(n = 150)^{a}$	()
	55 (36.7)
II	62 (41.3)
III	21 (14.0)
IV	11 (7.3)
V	1 (0.7)
Associated impairments ^b	
One or more associated impairment(s) $(n = 764)$	582 (76.2)
Epilepsy ($n = 758$)	82 (10.8)
Speech impairment ($n = 761$)	449 (59.0)
Intellectual impairment ($n = 759$)	439 (57.8)
Strabismus ($n = 452$)	49 (10.8)
Visual impairment $(n = 743)$	40 (5.4)
Hearing impairment ($n = 759$)	13 (1.7)
Swallowing difficulty ($n = 750$)	60 (8.0)
Gastro-oesophageal reflux disease (GORD) ($n = 748$)	21 (2.8)

Data are n (%). MACS: Manual Ability Classification System; GMFCS: Gross Motor Function Classification System.

^aMACS was applicable for 169 children > 4 years.

^bMultiple impairments were reported for several children.

Although data from LMICs are scarce, recent studies show that the burden of CP is substantially higher than in HICs. Populationbased prevalence of 3.4 per 1000 children was reported in Bangladesh and 2.9 per 1000 children in Uganda [5,13]. In contrast, a study predominantly including HICs reported the overall worldwide prevalence of CP is 2.1 per 1000 live births [1], the lowest rate (1.4 children per 1000 live births/neonatal survivors) being observed in Australia [3]. Our data provide valuable insights on the high burden and severity of CP in a hospital-based cohort in Vietnam. This is possibly reflecting the greater burden of perinatal risk factors (e.g., asphyxia, infections, prematurity) and poor access to perinatal care and rehabilitation commonly observed in LMICs [5,14–18].

The complete causal pathway of the brain injury causing CP is poorly understood. CP is associated with numerous prenatal and perinatal factors, such as perinatal asphyxia, neonatal jaundice, infections, preterm birth, low birth weight, multiple pregnancy and birth defects [18]. We reported the proportions of our cohort with these prenatal and perinatal factors in Table 3. Identification of known aetiologies of CP can be challenging in low resource settings in absence of the tools available in HICs [17]. The presumed timing of the brain injury causing CP was unknown for 31.7% of our cohort. Studies have defined prenatal, perinatal and postnatal causes variably and cases that have not been identified to have any known postnatal cause are often believed to have resulted from brain injury during the prenatal or perinatal period [17].

Findings from similar settings can also vary due to methodological differences. In our hospital-based cohort, majority of the mothers received regular antenatal care (86.5%, n = 652) and had institutional delivery of their child in a hospital (90.8%, n = 687). These findings markedly differ from the community-based studies conducted in Bangladesh and Indonesia where considerably fewer mothers received regular antenatal care (Bangladesh: 30.3%, Indonesia: 55.4%) and the majority (Bangladesh: 72.7%, Indonesia: 72.3%) of the children were delivered at home [5,16].

We have identified modifiable risk factors for CP and poor outcomes in many children, including hyperbilirubinemia, poor nutrition, lack of mobility devices and limited access to multidisciplinary services. This suggests opportunities for prevention of CP and its co-morbidities, decreasing disease severity, and improving quality of life for children with CP in Vietnam. In addition, we describe the clinical type, severity, associated impairments and rehabilitation status of children with CP. This information will inform clinician education and training, health policy and service planning and identify opportunities for evidence-based diagnosis and therapy.

evidence-based diagnosis and management Although approaches to CP have proven effective in HICs [19], there is limited information about their use in Vietnam. There are practical limitations to the use of some assessment methods and treatments. For example, the lack of dietitians precludes adequate nutritional assessment and treatment and, interventions, such as continuous nasogastric feeding, are not readily available [20]. Similarly, access to mobility aids is limited in Vietnam. Only 12 of the 454 children from our cohort with GMFCS III to V had access to assistive devices e.g. wheelchairs. The lack of availability of allied health professionals in Vietnam, including speech and occupational therapists, is a major limitation to optimal, multidisciplinary management of children with CP and diverse needs. Integration of these evidence-based approaches into current practice would improve management of children with CP in Vietnam substantially. Clinical guidelines for the management of children with CP have been developed and approved by the Vietnamese Ministry of Health and their implementation is currently underway across Vietnam [21].

Early intervention is limited by delayed diagnosis in LMICs, which contributes to poor motor and cognitive outcomes, is associated with secondary complications including contractures and malnutrition and negatively impacts carer wellbeing [22]. The mean age of diagnosis of CP was 1.7 years in our cohort. Owing to the hospital-based recruitment of our cohort we are unaware of the timing of CP diagnosis for majority of the community cases. It is likely to be delayed due to a range of reasons which are commonly observed in LMICs, such as poverty, poor educational status, lack of awareness and limited access to health information [5,23]. Community-based studies from other LMICs have documented delayed diagnosis at 5.2 years and 6.4 years in Bangladesh and Indonesia respectively [5,16]. In contrast, HICs such as Australia reported a mean age of CP diagnosis at 1.4 years [22]. Use of evidence-based approaches including the General

Table 3. Characteristics of pregnancy and perinatal period.

	n	(%)
Mothers with previous stillbirths more than 20 weeks ($n = 729$)	31	(4.3)
Mothers with previous miscarriages less than 20 weeks ($n = 747$)) 125	(16.7)
Family member with a disability $(n = 748)$	76	(10.2)
Sibling	29	(3.9)
One or both parents	18	(2.4)
Cerebral palsy	21	(2.8)
Agent Urange poisoning	3	(0.4)
Parent consanguinity $(n = 744)$	11	(1.5)
Assisted conception $(n = 736)$	13	(1.8)
Ovulation stimulation only	2	(1.0)
Artificial insemination	2	(0.4)
Gamete intrafallonian transfer (GIFT)	1	(0.3)
Antenatal care ($n = 754$)		(0.1)
No	31	(4.1)
Regular	652	(86.5)
Irregular (>2 visits)	60	(8.0)
Irregular (<2 visits)	11	(1.5)
Maternal febrile illness during pregnancy ($n = 762$)	102	(13.4)
Maternal febrile rash ($n = 680$)	18	(2.6)
Maternal febrile illness during labour ($n = 752$)	16	(2.1)
Place of birth ($n = 757$)		
Hospital	687	(90.8)
Health care centre	60	(7.9)
Home	10	(1.3)
Mode of delivery $(n = 756)$		(
Vaginal birth	504	(66./)
Caesarean section	238	(31.5)
Instrumental delivery (forceps or ventouse suction)	14	(1.9)
Mean destational age and range $(n - 762)$	27 2 1/0	(23.1) 15.75 /2
Term $(>37-41 \text{ weeks})$	57.2±4 541	+.3, 23-43 (71 0)
Preterm (\leq 36 weeks)	202	(71.0)
Most preterm (<27 weeks)	11	(5 4)
Very preterm (28–31 weeks)	64	(31.7)
Moderate preterm (32–33 weeks)	46	(22.8)
Late preterm (34–36 weeks)	81	(40.1)
Post-term (42 – 43 weeks)	19	(2.5)
Mean birthweight and SD (grams)	2783	± 708.0
Low birthweight (<2500 g)	200	(26.3)
Multiple birth $(n = 733)^a$	31	(4.2)
Early feeding difficulties ($n = 742$)	168	(22.6)
Birth defects	112	(14.8)
Brain anomalies	5	(4.5)
Known syndromes $(n = 726)$	6/	(9.2)
Perinatal aspnyxia ($n = 761$)	2/6	(30.3)
	60	(10.7)
Brain baemorrhage	209	(9.1)
Traumatic/accidental brain injury	11	(3.7) (1.4)
Hospital peopatal transfer ($n = 538$)	460	(85 5)
Central/national	128	(27.8)
District	106	(23.0))
Provincial	226	(49.1)
Mean length of stay at hospital of neonatal transfer (days)	18.7	± 16.2;
and range	1-	-114
Type of care received at hospital of neonatal transfer ($n = 696$)		
NICU	242	(31.6)
Special care	43	(5.6)
Routine care only	411	(53.7)
Mean number of hospitalisations for chest infections/	0.4±1	1.0; 1–10
respiratory infections in the last 6 months and		
range ($n = /45$)		

^aone triplet and 30 twin pregnancies.

^ba birth defect is defined as a structural anomaly present at birth, involving the face, limbs or any organ system. The presence of birth defects was determined by a paediatrician based on clinical assessment, review of hospital records, relevant investigations and parent report.

Movements Assessment, Hammersmith Infant Neurological Examination and MRI in HICs make the diagnosis of CP possible with high predictive value as early as 5 months of age [22]. Even

Table 4. Immunisation, education and rehabilitation status.

	n (%)
Fully vaccinated ($n = 753$)	623 (82.7)
BCG (n = 746)	701 (94.0)
MR (<i>n</i> = 512)	373 (72.9)
Reason for no vaccination $(n = 80)$	
Child was ill	64 (80.0)
No access at local centre or no transport to travel	3 (3.8)
Convulsions/post injection seizures	5 (6.3)
Family refused	2 (2.5)
Forgot immunization date	1 (1.3)
Unknown	5 (6.3)
Attend mainstream school $(n = 75)^{a}$	
No	23 (30.7)
Yes	52 (69.3)
Primary	46 (88.4)
Secondary	5 (9.6)
Vocational/other	1 (1.9)
Attend special school ($n = 725$)	9 (1.2)
Received rehabilitation $(n = 703)^{b}$	
Therapy exercises	663 (94.3)
Advice	151 (21.5)
Assistive device ^c	81 (11.5)
Muscle relaxant	5 (0.7)
Surgery	5 (0.7)
Acupuncture	1 (0.1)
Age of first rehab and range (years) ($n = 694$)	1.1 ± 1; 0–13
Location of rehabilitation service $(n = 659)^{d}$	
Home	585 (88.8)
Hospital	334 (50.7)
Private clinic	4 (0.6)

^aSchool attendance is applicable for 75 children aged \geq 6 years. ^bMany participants received more than one type of rehabilitation.

^cAssistive devices included wheelchairs and walkers.

^dMany participants received treatment at more than one location.

with the use of these approaches, however early diagnosis of CP can be challenging in HICs, particularly in milder cases and children with multiple associated impairments.

It was previously believed that improved neonatal care may increase survival rates and adverse neurodevelopmental outcomes including CP and poor quality of life. However, recent findings from Australia and Europe show that following an initial spike in the rate of CP among survivors, in recent years optimal NICU care is associated with decreasing rates of CP [24]. HICs have observed reductions in both the burden and severity of CP over recent decades with advances in evidence-based healthcare practices and public health initiatives [2,21]. Findings from our study can potentially guide the warranted strategies in Vietnam to achieve the successes observed in HICs in addressing the burden of CP.

In HICs the administration of magnesium sulphate to mothers at risk of a preterm birth has proven effective and holds tremendous potential in LMICs [25]. Preterm births, a leading risk factors for CP in HICs, are underrepresented in our cohort (Vietnam: 26.5%, Australia: 43.0%) [3]. This is possibly due to the high mortality among preterm babies born in LMICs and the failure to prevent, identify and treat preventable risk factors such as infection, asphyxia and hyperbillirubinemia [5,19]. However, the rate of preterm births among our cohort is significantly higher than the proportion of 9.0% among the general population of Vietnam [26]. Furthermore, children with CP are at high risk of incomplete and delayed immunisation compared to the general population which is consistent with our findings [27].

Although Vietnam has seen improved nutrition and child and maternal survival rates, health disparity across sociodemographic gradients is significant [28]. Over two thirds (68.8%) of our cohort were living below the national poverty line, which is substantially higher than the 20.7% among the general population [8].

Socioeconomic deprivation is often both a cause and consequence of having a child with CP [29]. Low socioeconomic status is associated with increased risk of CP and contributes to more severe functional limitations owing to greater exposure to risk factors and limited access to healthcare [30,31]. High parental unemployment rates (maternal: 35.0% and paternal: 24.5%) were observed in our cohort [32]. This is substantially higher than the national unemployment rate of 3.2%. Larger than average household size is known to be associated with poverty and we observed a household size of 4.7 members per household in our cohort compared to the national average of 3.5 members per household [33,34]. The Kinh ethnic group accounts for 85.3% of the total population in Vietnam and 97.9% of the total population of the Red River Delta region where our study site is located; we speculate that the burden and severity of CP would be even greater than in our cohort among ethnic minority groups and children living in remote or disadvantaged provinces in Vietnam who are more vulnerable to the known risk factors associated with CP [35].

Limitations

Although this study has yielded valuable new insights into the epidemiology of CP in Vietnam, the cohort is not population representative, thus we were unable to document prevalence. Furthermore, we do not know how generalisable these findings are to other provinces. Hospital-based recruitment, disparity in access to health care, and variable health-seeking behaviour may have resulted in selection bias. For example, there may have been underreporting of milder cases, children from poorer or less educated families and children from rural or remote settings. Similarly, a high proportion of our cohort had guadriplegia which may reflect overreporting of severe cases. In LMICs, premature mortality is high among children with CP and occurs well before the average age of CP diagnosis in LMICs [36]. Our surveillance does not capture children who died prior to diagnosis. Despite these limitations these data were collected using a well-established method, from the main paediatric referral hospital (NCH) in the most densely populated region of Vietnam (the Red River Delta Region) and serves the majority of that child population.

Although cognitive delay may be evident in young children, it may be transient. Intellectual impairment cannot be formally confirmed until \sim 5 years of age, when standardised, valid measures that are predictive and reliable can be used [37]. In our study, assessment of intellectual impairment or cognitive delay was based on clinical assessment, report by the parents/primary caregivers or review of existing medical records, as most of our cohort is under 5 years of age. We acknowledge that a limitation of the study is the inability to accurately estimate the proportion who will have ongoing intellectual impairment.

We were unable to conduct detailed assessment of the associated impairments due to limited access to audiology, formal testing of speech and language and cognitive testing. We relied on clinical history, detailed assessment conducted by clinicians and report by the parents/primary caregivers. History taking from parents by the clinicians on retrospective data is subject to recall bias.

Despite its limitations, this study provides baseline descriptive data on a large cohort of children with CP in Vietnam. Although there is likely selection bias, our findings demonstrate a high burden of CP on the hospital system and in the absence of any such data on children with CP in Vietnam we believe our study findings are valuable in identifying opportunities for prevention, needs for diagnostic tools, therapy and equipment, and service planning and training needs.

Conclusion

We established a hospital-based, prospective surveillance system to identify and describe children with CP. Our novel data reveal a high burden and severity of CP in this hospital setting and opportunities for prevention. Our data will inform clinician education and training, health policy, service planning and evidence-based diagnosis and therapy.

Disclosure statement

No potential conflict of interest was reported by the author(s).

Funding

The study sponsor played no role in study design or the collection, analysis, and interpretation of data; in the writing of the report; and in the decision to submit the paper for publication. EE is supported by a Practitioner Fellowship from the National Health and Medical Research Council of Australia [1021480]. TK is supported by Cerebral Palsy Alliance Research Foundation [CDG04617, PHD02119].

ORCID

Tasneem Karim (b) http://orcid.org/0000-0003-1480-8409

Data availability statement

The data that support the findings of this study are available from the corresponding author upon reasonable request.

References

- Oskoui M, Coutinho F, Dykeman J, et al. An update on the prevalence of cerebral palsy: a systematic review and meta-analysis. Dev Med Child Neurol. 2013;55(6):509–519.
- [2] Sellier E, Platt MJ, Andersen GL, et al. Decreasing prevalence in cerebral palsy: a multi-site European populationbased study, 1980 to 2003. Dev Med Child Neurol. 2016; 58(1):85–92.
- [3] ACPR. Report of the Australian Cerebral Palsy Register, birth years 1995–2012. Australia. 2018 [cited 2018 Dec]. Available from: https://cpregister.com/wp-content/uploads/2019/02/ Report-of-the-Australian-Cerebral-Palsy-Register-Birth-Years-1995-2012.pdf
- [4] Reid SM, Meehan E, McIntyre S, et al. Temporal trends in cerebral palsy by impairment severity and birth gestation. Dev Med Child Neurol. 2016;58:25–35.
- [5] Khandaker G, Muhit M, Karim T, et al. Epidemiology of cerebral palsy in Bangladesh: a population-based surveillance study. Dev Med Child Neurol. 2019;61(5):601–609.
- [6] Khandaker G, Van Bang N, Dũng TQ, et al. Protocol for hospital based-surveillance of cerebral palsy (CP) in Hanoi using the Paediatric active enhanced disease surveillance mechanism (PAEDS-Vietnam): a study towards developing hospital-based disease surveillance in Vietnam. BMJ Open. 2017;7(11):e017742.

- [7] Zurynski Y, McIntyre P, Booy R, et al. Paediatric active enhanced disease surveillance: a new surveillance system for Australia. J Paediatr Child Health. 2013;49(7):588–594.
- [8] The World Bank, Poverty reduction in Vietnam: Remarkable progress, Emerging challenges. 2013 [cited 2019 May 6]. Available from: http://www.worldbank.org/en/news/feature/ 2013/01/24/poverty-reduction-in-vietnam-remarkable-progress-emerging-challenges
- [9] Rosenbaum PL, Palisano RJ, Bartlett DJ, et al. Development of the gross motor function classification system for cerebral palsy. Dev Med Child Neurol. 2008;50(4):249–253.
- [10] Eliasson A-C, Krumlinde-Sundholm L, RöSblad B, et al. The Manual Ability Classification System (MACS) for children with cerebral palsy: scale development and evidence of validity and reliability. Dev Med Child Neurol. 2006;48(7): 549–554.
- [11] Recommended national immunization schedule for Viet Nam's children, UNICEF. 2015 [cited 2019 Feb] Available from: http://www.wpro.who.int/vietnam/mediacentre/ releases/2015/childhood_immunization_epi_schedule_ march 2015 vietnam v2.pdf
- [12] Britton PN, Dale RC, Blyth CC, Clark JE, et al. Causes and clinical features of childhood encephalitis: a multicenter, prospective cohort study. Clin Infect Dis. 2020;70(12): 2517–2526.
- [13] Kakooza-Mwesige A, Andrews C, Peterson S, et al. Prevalence of cerebral palsy in Uganda: a population-based study. Lancet Global Health. 2017;5(12):e1275–e1282.
- [14] Singhi P, Saini AG. Changes in the clinical spectrum of cerebral palsy over two decades in North India-an analysis of 1212 cases. J Trop Pediatr. 2013;59(6):434–440.
- [15] Benfer KA, Jordan BP, Bandaranayake S, et al. Motor severity in children with cerebral palsy studied in a highresource and low-resource country. Pediatrics. 2014;72:14.
- [16] Jahan I, Al Imam MH, Karim T, et al. Epidemiology of cerebral palsy in Sumba Island, Indonesia. Dev Med Child Neurol. 2020;62(12):1414–1422.
- [17] Gladstone M. A review of the incidence and prevalence, types and aetiology of childhood cerebral palsy in resource-poor settings. Ann Trop Paediatr. 2010;30(3): 181–196.
- [18] Eunson P. Aetiology and epidemiology of cerebral palsy. Paediatr Child Health. 2012;22(9):361–366.
- [19] Novak I, Mcintyre S, Morgan C, et al. A systematic review of interventions for children with cerebral palsy: state of the evidence. Dev Med Child Neurol. 2013;55(10):885–910.
- [20] Karim T, Jahan I, Dossetor R, et al. Nutritional status of children with cerebral palsy—findings from prospective hospital-based surveillance in Vietnam indicate a need for action. Nutrients. 2019;11(9):2132.
- [21] Humanity and inclusion, Vietnam national rehabilitation guidelines. 2018 [cited Dec 2018]. Available from: https:// hivndrive.wixsite.com/hivn/rehabguidelines
- [22] Novak I, Morgan C, Adde L, et al. Early, accurate diagnosis and early intervention in cerebral palsy: advances in diagnosis and treatment. JAMA Pediatr. 2017;171(9):897–907.

- [23] Donald KA, Samia P, Kakooza-Mwesige A, et al. Pediatric cerebral palsy in Africa: a systematic review. Semin Pediatr Neurol. 2014;21(1):30–35.
- [24] Wilson-Costello D, Friedman H, Minich N, et al. Improved survival rates with increased neurodevelopmental disability for extremely low birth weight infants in the 1990s. Pediatrics. 2005;115(4):997–1003.
- [25] Conde-Agudelo A, Romero R. Antenatal magnesium sulfate for the prevention of cerebral palsy in preterm infants less than 34 weeks' gestation: a systematic review and metaanalysis. Am J Obstet Gynecol. 2009;200(6):595–609.
- [26] Blencowe H, Cousens S, Oestergaard M, et al. National, regional and worldwide estimates of preterm birth rates in the year 2010 with time trends since 1990 for selected countries: a systematic analysis and implications. Lancet. 2012;379(9832):2162–2172.
- [27] Greenwood VJ, Crawford NW, Walstab JE, et al. Immunisation coverage in children with cerebral palsy compared with the general population. J Paediatr Child Health. 2013;49(2):E137–41.
- [28] Målqvist M, Hoa DT, Thomsen S. Causes and determinants of inequity in maternal and child health in Vietnam. BMC Public Health. 2012;12(1):641.
- [29] Solaski M, Majnemer A, Oskoui M. Contribution of socioeconomic status on the prevalence of cerebral palsy: a systematic search and review. Dev Med Child Neurol. 2014; 56(11):1043–1051.
- [30] McIntyre S, Taitz D, Keogh J, et al. A systematic review of risk factors for cerebral palsy in children born at term in developed countries. Dev Med Child Neurol. 2013;55(6): 499–508.
- [31] Durkin MS. Socio-economic disparities and functional limitations of children with cerebral palsy. Dev Med Child Neurol. 2016;58(2):115–115.
- [32] World Health Organisation. Human resources for health country profiles: Viet Nam. Manila: WHO Regional Office for the Western Pacific. 2016. Available from: https://apps.who. int/iris/handle/10665/259990
- [33] Lanjouw P, Ravallion M. Poverty and household size. Economic J. 1995;105(433):1415–1434.
- [34] General Statistic Office of Vietnam. Population and housing census in 2019. [cited 2019 Sep 29]. Available from: https:// www.gso.gov.vn/default_en.aspx?tabid=515&idmid=5& ltemID=19281
- [35] Eriksson L, Nga NT, Målqvist M, et al. Evidence-based practice in neonatal health: knowledge among primary health care staff in northern Viet Nam. Hum Resour Health. 2009; 7(1):36.
- [36] Jahan I, Karim T, Das MC, et al. Mortality in children with cerebral palsy in rural Bangladesh: a population-based surveillance study. Dev Med Child Neurol. 2019;61(11): 1336–1343.
- [37] Moeschler JB, Shevell M, Committee on Genetics. Comprehensive evaluation of the child with intellectual disability or global developmental delays. Pediatrics. 2014; 134(3):e903–e918.
- [38] ILO. 2007. International standard classification of occupations: ISCO-08. Geneva: International Labour Office.



Article

Nutritional Status of Children with Cerebral Palsy—Findings from Prospective Hospital-Based Surveillance in Vietnam Indicate a Need for Action

Tasneem Karim ^{1,2,3,4,*}, Israt Jahan ^{2,3}, Rachael Dossetor ¹, Nguyen Thi Huong Giang ⁵, Nguyen Thi Van Anh ⁶, Trinh Quang Dung ⁵, Cao Minh Chau ^{7,8}, Nguyen Van Bang ^{9,10}, Nadia Badawi ^{1,4}, Gulam Khandaker ^{1,2,3,11} and Elizabeth Elliott ^{1,12}

- ¹ Discipline of Child and Adolescent Health, Faculty of Medicine and Health, The University of Sydney, Sydney 2145, Australia
- ² CSF Global, Dhaka 1213, Bangladesh
- ³ Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka 1213, Bangladesh
- ⁴ Cerebral Palsy Alliance Research Institute, The University of Sydney, Sydney 2050, Australia
- ⁵ Rehabilitation Department, National Children's Hospital, Hanoi 117200, Vietnam
- ⁶ Department of Medical Education and Skills Lab, Hanoi Medical University, Hanoi 116177, Vietnam
- ⁷ Rehabilitation Department, Hanoi Medical University (Bach Mai Hospital), Hanoi 116177, Vietnam
- ⁸ Faculty of Health Sciences, Phenikaa University, Hanoi 152308, Vietnam
- ⁹ Department of Paediatrics, Hanoi Medical University, Hanoi 116177, Vietnam
- ¹⁰ Faculty of Medicine, Hanoi University of Business and Technology, Hanoi 113400, Vietnam
- ¹¹ Central Queensland Public Health Unit, Central Queensland Hospital and Health Service, Rockhampton, Queensland 4700, Australia
- ¹² The Sydney Children's Hospitals Network, Westmead, Sydney 2145, Australia
- * Correspondence: tasneem.karim.tk@gmail.com; Tel.: +6-1431-959-050

Received: 28 June 2019; Accepted: 29 August 2019; Published: 6 September 2019



Abstract: Background: Lack of evidence on the burden and risk factors for malnutrition among children with cerebral palsy (CP) in Vietnam limits evidence-based interventions. We aimed to define the nutritional status of children with CP in Vietnam. Materials and Methods: The study utilized data from active prospective hospital-based surveillance modelled on the Pediatric Active Enhanced Disease Surveillance system. Children (0-18 years) with CP attending the National Children's Hospital Hanoi, Vietnam between June-November 2017 were included. Data on demographic, clinical and rehabilitation status were collected following detailed neurodevelopmental assessment. Anthropometric measurements were taken. Nutritional status was determined using the World Health Organization guideline. Results: Of 765 children (the mean (SD) age was 2.6 (2.5) years; 35.8% were female), 28.9% (n = 213) were underweight and 29.0% (n = 214) stunted. The odds of underweight were significantly higher among children aged >5 years and/or having a monthly family income of <50 USD. Underweight and/or stunting was high among children with quadriplegia (81%, n = 60 and 84.5%, n = 87) and/or Gross Motor Functional Classification System (GMFCS) level IV-V (62.5%, n = 45 and 67.0%, n = 67). Nearly one-third of intellectually impaired and more than half of hearing-impaired children were underweight and/or stunted. Conclusions: Poor economic status and increased motor severity increased vulnerability to malnutrition. Our findings will inform nutritional rehabilitation programs among these vulnerable children.

Keywords: cerebral palsy; hospital-based surveillance; malnutrition; Vietnam; children



1. Introduction

Poor nutrition and growth faltering disrupt cognition and development and are commonly observed among children with Cerebral Palsy (CP). CP is a major cause of childhood disability. It is a non-progressive neurological disorder resulting from an insult to the developing brain that affects posture and movement [1]. Oromotor dysfunction and gastrointestinal problems, often associated with motor impairment, hinder optimum nutrition of children with CP [2,3]. The worldwide prevalence of CP is 2.1 per 1000 live births with a rate of 1.4 per 1000 live births observed in Australia [4,5]. However, a recent population-based study in Bangladesh reported a prevalence of 3.4 per 1000 live births [6]. This is substantially higher than the global estimate, which largely reflects findings from high income countries (HICs). This disparity is most likely due to increased risk factors, including birthing practices, in a poor socio-economic context [7,8], which is also a key underlying risk factor for malnutrition among children [9].

Maintenance of optimal nutrition is challenging among children with CP. Although, studies from both HICs and low and middle-income countries (LMICs) have demonstrated high prevalence of malnutrition among children with CP compared to the general population [10,11], evidence suggests that the burden is considerably higher in LMICs than in HICs [10–12]. Yet the underlying risk factors in LMICs are poorly understood. Severe motor impairment, measured using the Gross Motor Functional Classification System (GMFCS), presence of associated impairments and oropharyngeal dysphagia affect the nutritional status of children with CP [2,10,12].

Feeding difficulties among children with CP increase their risk of growth failure [3]. A populationbased study in Norway reported significantly higher longitudinal growth faltering in the first five years of age among children with CP who had feeding difficulties compared to their peers without such difficulties [11]. In Ghana, children with CP and feeding difficulties had 3 to 10 times higher odds of being underweight than children with CP without feeding difficulties [13]. Similar patterns were observed in Uganda [14]. Moreover, children with CP are at high risk of poor nutritional outcome due to metabolic alterations and changes in body composition for example, decreased fat free mass, changes in body fat level and altered calorie demand, associated with CP [15]. In LMICs, the significance of these risk factors is likely to be intensified due to their complex interaction with socio-economic factors.

Adverse consequences of malnutrition among children with CP are widespread. Evidence from different countries reveals that malnutrition among children with CP results in poor health-related outcomes, poor quality of life and premature mortality [16,17]. The vicious cycle of malnutrition and infection has also been illustrated in several studies [18]. In a recent population-based study in Bangladesh, 86% of deaths among children with CP were attributed to infections and nearly all of the children who died were undernourished [16]. Studies conducted in Bangladesh and Ghana also reported poor quality of life among the caregivers of children with CP and feeding difficulties [13,19]. Existing evidence suggests that the scenario is likely to be similar for children with CP in the resource constraint settings of other LMICs such as Vietnam.

Nutritional interventions such as gastrostomy and tube feeding are beneficial for weight gain among children with CP [20]. However, such interventions are not commonly available or used in low resource settings and are reportedly poorly understood by clinicians and families. Furthermore, in LMICs there is limited access to dieticians and speech pathologists, who play a crucial role in the management of feeding problems among children with CP. Better understanding of the nutritional status of children with CP is essential to guide the development of evidence-based programs suitable for these settings. In this study we aimed to assess the burden of malnutrition and the underlying risk factors among children with CP in Vietnam.

2. Materials and Methods

The study utilized data from an active hospital-based surveillance of children with CP in the National Children's Hospital in Hanoi, Vietnam [21]. The diagnosis of CP was made using the standard definition adopted from Surveillance of Cerebral Palsy in Europe and the Australian CP Register [5,22].

2.1. Study Setting and Participants

Study participants included children aged ≤18 years who visited the National Children's Hospital in Hanoi, Vietnam between June and November 2017 and were diagnosed with CP.

2.2. Data Collection and Clinical Assessment

Data were recorded by trained clinicians using a standardized case record form in the Vietnamese language and responses were translated into English by two clinicians on the study team. Sociodemographic factors (e.g., age, gender, income, educational level of the parents), clinical factors (e.g., GMFCS level, Manual Ability Classification System (MACS) level, predominant CP motor type, associated impairments) and anthropometric measurements (e.g., weight, height) were collected as part of the registration and clinical assessment process. A detailed study protocol including the methods and measures used in the study has been published previously [21].

2.3. Anthropometric Measurements

We measured height and weight using a standard World Health Organization (WHO) guideline [23].

Weight: Weight was measured in kilograms using a digital weighing scale with a precision of 100 g following standard guidelines (i.e., removing extra clothing, standing still). Tared weight was measured for young children aged less than 2 years and for children who could not stand alone. Clinicians were trained before taking anthropometric measurements. Three repeated measures were taken and the average was noted in the questionnaire.

Height: Recumbent length was measured in cm for children aged less than 2 years using a length board and standing height was measured for children aged ≥ 2 years using a standing frame. The mean of three repeated measures were taken to minimize measurement error. For children who could not stand alone, segmental height (i.e., knee height) was measured and full height was derived using the following formula; height = (2.69 × knee height) + 24.2 cm [24].

2.4. Data Management and Analysis

Data entry and analyses were carried out using SPSS version 23.0 (IBM Corporation, Chicago, IL, USA). All data were entered into a database by investigators in Sydney and Vietnam. Three indices were used to describe the nutritional status of children with CP—weight for age (WA), height for age (HA) and weight for height (WH). The z scores for these three indices (i.e., WAZ, HAZ and WHZ) were calculated using WHO Anthro (version 3.2) and WHO AnthroPlus software. WAZ was calculated for children aged less than 121 months and WHZ was calculated for children aged less than 61 months. Nutritional status of children was categorized using the WHO cutoffs for the z scores (i.e., –2SD to +2SD: Normal; <–2SD to –3SD: Moderately undernourished and <–3SD: Severely undernourished).

A normality test was conducted for all continuous variables for example, WAZ, HAZ and WHZ using Shapiro Wilk test. In case of variables not normally distributed [WAZ, HAZ, WHZ] we used median [IQR]). Descriptive analyses were done to present socio-demographic characteristics, clinical factors and nutritional status of the study participants. Bivariate analyses were done to identify potential risk factors for malnutrition among children with CP. Chi-square and Fisher's exact tests were used to examine the statistical differences in nutritional status between groups of children according to their socio-demographic characteristics and clinical factors. The nutritional status of children with CP aged less than five years was compared with the general population of the same age group in Vietnam using World Bank data. Binomial test were done to identify statistically significant differences between the proportion of malnutrition among our study participants and a general population of the same age group. Factors that were found to be related to underweight and/or stunting among children in bivariate analyses for example, chi-square test, Fisher's exact tests and unadjusted logistic

regression were included in an adjusted logistics regression model. A p value < 0.05 was considered statistically significant.

2.5. Ethical Consideration

Ethics approval was obtained from the University of Sydney Human Research Ethics Committee (HREC) (2016/456), Hanoi Medical University (HMU) (1722/QD-QHYHN) and NCH (812/QD-BVNTU) in Hanoi, Vietnam. Informed consent was obtained from parents or caregivers of all the study participants.

3. Results

A total of 765 children with CP and aged less than 18 years was identified between June and November 2017. The mean (SD) age was 2.6 (2.5) years and 35.8% were female.

3.1. Overall Nutritional Status

Of the 765 children 71.1% (n = 524) had a normal WAZ, 70.1% (n = 502) had a normal HAZ and 75.1% (n = 435) children had a normal WHZ. The distribution of WAZ, HAZ and WHZ was positively skewed, hence the median [IQR] for these three indicators was reported. The median [IQR] z score for WAZ was -1.3 [-2.2, -0.2], the median z score for HAZ was -1.1 [-2.3, 0.2] and the median z score for WHZ was -0.86 [-2.0, 0.3]. Among the children, 28.9% (n = 213) were underweight, 29.9% (n = 214) were stunted and 24.9% (n = 144) were wasted. When compared to children aged under five years in the general population in Vietnam, children aged less than five years with CP had a significantly higher prevalence of underweight, stunting and wasting (p < 0.001 for all three indices). Moreover, 14.4% (n = 103) of children with CP had severe chronic malnutrition (i.e., severe stunting) and 13.8% (n = 80) had severe acute malnutrition (SAM) (i.e., severe wasting). (Table 1)

	Children	Children with CP in Our Study					
Indicators	Overall		Aged <5 Years	Section (Aged <5 Years) in 2015 ¹	<i>p</i> Value ²		
	Median [IQR]	n (%)	n (%)	(%)			
	Weight-for-age z score ($n = 737$) ³						
Normal		524 (71.1)	_	_			
Underweight	-1.25 [-2.2,	. ,			-0.001		
Moderate	-0.2]	139 (18.9)	176 (27 7)	141	<0.001		
Severe		74 (10.0)	170 (27.7)	14.1			
]	Height-for-age	$z \text{ score } (n = 716)^4$	L			
Normal		502 (70.1)	_	_			
Stunted	_11[_2302]				<0.001		
Moderate	-1.1 [-2.3, 0.2]	111 (15.5)	184 (30.6)	24.6	<0.001		
Severe		103 (14.4)	104 (00.0)	24.0			
	Weight-for-height z score ($n = 579$) ⁵						
Normal		435 (75.1)	_	_			
Wasted	_0.86 [_2.0.0.3]				<0.001		
Moderate	-0.00 [-2.0, 0.3]	64 (11.1)	142 (24 7)	6.4	\0.001		
Severe		80 (13.8)	172 (27.7)	0.4			

Table 1. Nutritional status of the study participants.

¹ The World Bank data, 2015. Available from: https://data.worldbank.org, ² Binomial test. ³ Weight-for-age z score was calculated for children aged ≤ 121 months. ⁴ Missing data (n = 49). ⁵ Weight-for-height z score was calculated for children aged ≤ 61 months. The *p* values for statistically significant differences are shown in bold.

3.2. Socio-Demographic Characteristics and Nutritional Status of Children with CP

Table 2 summarizes the nutritional status of study participants according to their sociodemographic characteristics. A significant inverse relationship was observed between age group

Ę

and proportion of underweight among children with CP. Among children aged less than two years, 25.3% (n = 106) were underweight. The proportion of underweight was significantly higher among children aged 2–4 years (32.3%, n = 70) and ≥ 5 years (36.6%, n = 37) (p = 0.03). Conversely, stunting was less prevalent among children aged ≥ 5 years.

	Total	Weight-	for-Age Z Score	$(n = 737)^{1}$	Height-fo	r-Age Z Scor	re ($n = 716$) ²
Indicators	<i>n</i> = 765	Normal $n = 524$	Underweight n = 213	<i>p</i> Value ³	Normal $n = 502$	Stunted $n = 214$	<i>p</i> Value ³
		Age g	roups (years), n :	= 765			
less than 2	424 (55.4)	313 (74.7)	106 (25.3)	0.03	282 (72.3)	108 (27.7)	0.07
2–4	220 (28.8)	147 (67.7)	70 (32.3)		135 (64.0)	76 (36.0)	0.07
5 and more	121 (15.8)	64 (63.4)	37 (36.6)		85 (73.9)	30 (26.1)	
			Sex, $n = 765$				
Female	274 (35.8)	200 (75.2)	66 (24.8)	0.07	184 (71.6)	73 (24.8)	0.52
Male	491 (64.2)	324 (68.8)	147 (31.2)		318 (69.3)	141 (30.7)	
		Source of	f drinking water	, <i>n</i> = 761			
Piped water	423 (55.6)	308 (75.3)	101 (24.7)	0.01	287 (72.8)	107 (27.2)	0 17
Well water	284 (37.3)	182 (66.9)	90 (33.1)	(6	179 (66.8)	89 (33.2)	0.17
Other sources (ponds/river/stream/lake)	54 (7.1)	31 (58.5)	22 (41.5)		33 (64.7)	18 (35.3)	
		Sanita	ation practice, <i>n</i> :	= 759			
Flush toilet	633 (83.4)	439 (72.1)	170 (27.9)	0.18	425 (71.7)	168 (28.3)	0.04
Pit toilet	126 (16.6)	82 (66.1)	42 (33.9)		74 (62.2)	45 (37.8)	
		Monthly	/ family income,	<i>n</i> = 728			
below 50 USD	180 (24.7)	104 (60.5)	68 (39.5)		110 (65.1)	59 (34.9)	
51-100 USD	231 (31.7)	158 (71.8)	62 (28.2)	0.002	154 (70.0)	66 (30.0)	0.23
101-150 USD	153 (21.0)	109 (73.6)	39 (26.4)		103 (71.5)	41 (28.5)	
Above 150 USD	164 (22.5)	127 (78.9)	34 (21.1)		112 (75.7)	36 (24.3)	

Table 2. Socio-demographic characteristics of children and their nutritional status.

¹ Weight-for-age z score was calculated for children aged \leq 121 months. ² Missing data (n = 49). ³ Chi square test. The *p* values for statistically significant differences are shown in bold.

The burden of underweight was significantly higher among children who did not have access to safe drinking water compared to those who had access (24.7%, n = 101 vs. 41.5%, n = 22; p = 0.01). Although not statistically significant, a similar finding was observed for stunting among children with CP in our cohort.

Rates of both underweight (39.5%, n = 68) and stunting (34.9%, n = 59) were highest among children whose monthly family income was less than 50 USD. When compared with the proportion of underweight children in families in our cohort with a monthly family income of >150 USD (21.1%, n = 34) the difference was significant (p < 0.001). Although stunting was more prevalent among children with a monthly family income of <50 USD compared to those with higher income, this difference was not statistically significant (p = 0.23).

3.3. Clinical Characteristics and Nutritional Status of Children With CP

3.3.1. Age of CP Diagnosis and Timing of CP

Although not statistically significant, both underweight and stunting were slightly higher among children who had a confirmed diagnosis of CP at or over five years of age than others in the cohort. Similarly, children with pre and perinatally acquired CP had higher rates of underweight and stunting than children with postnatally acquired CP. (Table 3)

To Proton	TotalWeight-for-Age Z Score, $(n = 737)^{1}$			Height-for	-Age Z Score, $(n = 716)^2$			
Indicators	<i>n</i> = 765	Normal <i>n</i> = 524 n (%)	Underweight <i>n</i> = 213 n (%)	<i>p</i> Value ³	Normal <i>n</i> = 502 n (%)	Stunted <i>n</i> = 214 n (%)	<i>p</i> Value ³	
Age of CP diagnosis (years), $n = 754$								
less than 2	559 (74.1)	395 (72.2)	152 (27.8)		370 (70.7)	153 (29.3)		
2–4	143 (19.0)	95 (68.3)	44 (31.7)	0.45	94 (68.6)	43 (31.4)	0.87	
5 and more	52 (6.9)	28 (65.1)	15 (34.9)		33 (68.8)	15 (31.3)		
			Timing of CP, <i>n</i> =	753				
Postnatal	98 (13.0)	72 (75.0)	24 (25.0)		72 (76.6)	22 (23.4)		
Pre and perinatal	416 (55.2)	279 (69.4)	123 (30.6)	0.44	259 (66.4)	131 (33.6)	0.05	
Unknown	239 (31.7)	167 (72.9)	62 (27.1)		165 (74.0)	58 (26.0)		
			GMFCS level, <i>n</i> =	: 754				
Ι	179 (23.7)	125 (76.7)	38 (23.3)		134 (78.8)	36 (21.2)		
II	121 (16.0)	89 (76.7)	27 (23.3)		80 (69.6)	35 (30.4)		
III	103 (13.7)	76 (73.8)	27 (26.2)	0.04	71 (75.5)	23 (24.5)	0.01	
IV	131 (17.4)	88 (68.8)	40 (31.3)		81 (65.9)	42 (34.1)		
V	220 (29.2)	140 (64.2)	78 (35.8)		129 (63.2)	75 (36.8)		
			MACS level, <i>n</i> =	188				
Ι	71 (37.8)	44 (72.1)	17 (27.9)		48 (70.6)	20 (29.4)		
II	73 (38.8)	47 (69.1)	21 (30.9)		48 (69.6)	21 (30.4)		
III	26 (13.8)	17 (70.8)	7 (29.2)	0.77	18 (75.0)	6 (25.0)	$0.58^{\ 4}$	
IV	17 (9.0)	10 (58.8)	7 (41.1)		11 (64.7)	6 (35.3)		
V	1 (0.5)	-	-		0 (0.0)	1 (100.0)		
			Neurological type of C	P, <i>n</i> = 765				
Mono/hemi	157 (20.5)	118 (79.2)	31 (20.8)		119 (79.9)	30 (20.1)		
Diplegia	40 (5.2)	30 (76.9)	9 (23.1)		29 (76.3)	9 (23.7)		
Triplegia	0 (0.0)	0 (0.0)	0 (0.0)	0.104	0 (0.0)	0 (0.0)	0.01.4	
Quadriplegia	532 (69.5)	352 (68.2)	164 (31.8)	0.10*	328 (66.4)	166 (33.6)	0.01 *	
Ataxia	1 (0.1)	1 (100.0)	0 (0.0)		0 (0.0)	1 (100.0)		
Dyskinesia	35 (4.6)	23 (71.9)	9 (28.1)		26 (76.5)	8 (23.5)		
Number of associated impairments, $n = 765$								

Table 3. Clinical characteristics and nutritional status of the children with Cerebral Palsy (CP).

T. 1	Total	Weight-	for-Age Z Score, ($n = 737$) ¹		Height-for	-Age Z Score, $(n = 716)^2$	
indicators	<i>n</i> = 765	Normal <i>n</i> = 524 n (%)	Underweight <i>n</i> = 213 n (%)	<i>p</i> Value ³	Normal <i>n</i> = 502 n (%)	Stunted <i>n</i> = 214 n (%)	<i>p</i> Value ³
None	196 (25.6)	143 (78.1)	40 (21.9)		146 (78.5)	40 (21.5)	
At least one	206 (26.9)	145 (72.1)	56 (27.9)	0.02	131 (70.1)	56 (29.9)	0.01
Multiple	363 (47.5)	236 (66.9)	117 (33.1)		225 (65.6)	118 (34.4)	
			Type of associated imp	airments			
Epilepsy	82 (10.8)	54 (68.4)	25 (31.6)	0.41	56 (71.8)	22 (28.2)	0.71
Intellectual	439 (65.6)	283 (66.6)	142 (33.4)	0.001	273 (65.5)	144 (34.5)	< 0.001
Visual	40 (5.4)	24 (61.5)	15 (38.5)	0.11	21 (53.8)	18 (46.2)	0.02
Hearing	13 (1.7)	6 (46.2)	7 (53.8)	0.04	9 (69.2)	4 (30.8)	$0.60^{\ 4}$
Speech	449 (63.7)	305 (69.2)	136 (30.8)	0.07	281 (67.9)	133 (32.1)	0.04
			Swallowing difficultie	s, <i>n</i> = 740			
Present	60 (8.1)	39 (7.7)	20 (9.6)		38 (7.8)	18 (8.8)	
Absent	680 (91.9)	469 (92.3)	188 (90.4)	0.39	451 (92.2)	186 (91.2)	0.64
			Received rehabilitation ser	vices, $n = 75$	9		
Ever received rehabilitation services	703 (92.6)	483 (70.9)	198 (29.1)	0.56	461 (69.1)	206 (30.9)	0.03

Table 3. Cont.

¹ Weight-for-age z score was calculated for children aged ≤ 121 months. ² Missing data (n = 49). ³ Chi square test. ⁴ Fisher's exact test. The *p* values for statistically significant differences are shown in bold.

3.3.2. Motor Severity and Neurological Type of CP

The proportion of children with underweight was significantly higher among those with GMFCS level V than in children with GMFCS level I (35.8%, n = 78 vs. 23.3%, n = 38; p = 0.04), as was the proportion with stunting (36.8%, n = 75 vs. 21.2%, n = 36; p = 0.01). A similar pattern was observed for MACS level among children aged 4 years and above. Among children with MACS level IV, 41.1% (n = 7) were underweight and 35.5% (n = 6) were stunted, whereas these percentages were 27.9% (n = 17) and 29.4% (n = 20) respectively for children with MACS level I (p = 0.77 and p = 0.58 respectively).

The majority of children in our study had spastic CP, however none had triplegia. Both underweight (31.8%, n = 164) and stunting (33.6%, n = 166) were more commonly observed among children with quadriplegia than the children without quadriplegia (p = 0.01). (Table 3)

3.3.3. Associated Impairments

In our study, 74.4% (n = 569) children had one or multiple associated impairments. Among the children who had one or more associated impairments, 33.1% (n = 117) were underweight and 34.4% (n = 118) were stunted. The proportion with underweight or stunting was significantly higher among children with one or multiple associated impairments (p = 0.02 and p = 0.01 respectively) than with no associated impairments. Both underweight and stunting were significantly higher among children reported to have intellectual impairment (p < 0.001) or hearing impairment (p < 0.001) compared to children without these impairments. (Table 3)

3.3.4. Factors Associated with Underweight and Stunting among Children with CP

Tables 4 and 5 present findings from unadjusted and adjusted analyses. In the adjusted analysis, the odds of being underweight was statistically associated with the age of the children, monthly family income and GMFCS level. Children aged more than five years had 3.2 times higher odds of being underweight than children aged less than two years. When adjusted for other covariates, children from the poorest families (monthly family income <50 USD) had 3.0 times higher odds of being underweight than children whose monthly family income was >150 USD.

Monthly family income was not significantly associated with stunting. However, children between 2 and 4 years of age had higher odds of being stunted than younger children. When adjusted for covariates, the presence of intellectual impairment and motor function at GMFCS levels IV and V significantly increased the odds of stunting. Similarly, underweight was associated with GMFCS levels IV and V but there was no association between intellectual impairment and the level of underweight.

	Underweig	;ht	Stunting	Stunting	
Covariates	Odds Ratio (95% CI) (Unadjusted) <i>p</i> Value		Odds Ratio (95% CI) (Unadjusted)	p Value	
	Age group	(years)			
less than 2	Reference				
2–4	1.4 (1.0, 2.0)	0.63	1.5 (1.2, 2.1)	0.03	
5 and more	1.7 (1.1, 2.7)	0.02	0.9 (0.6, 1.5)	0.73	
	Source of drin	king water			
Well water		Refe	rence		
Piped water	0.7 (0.5, 0.9)	0.02	0.7 (0.5, 1.0)	0.09	
Other sources (ponds/river/stream/lake)	1.4 (0.8, 2.6)	0.24	1.1 (0.6, 2.5)	0.77	

Table 4. Factors associated with underweight and stunting among children with CP in Vietnam (unadjusted analyses).

	Underweig	;ht	Stunting	5
Covariates	Odds Ratio (95% CI) (Unadjusted)	p Value	Odds Ratio (95% CI) (Unadjusted)	p Value
	Monthly family	income (USI))	
Above 150		Refe	erence	
below 50	2.4 (1.5, 4.0)	<0.001	1.7 (1.0, 2.7)	0.04
51-100	1.5 (0.9, 2.4)	0.12	1.3 (0.8, 2.1)	0.23
101–150	1.3 (0.8, 2.3)	0.28	1.2 (0.7, 2.1)	0.42
	GMFCS	level		
Ι		Refe	erence	
II	1.0 (0.6, 1.7)	0.99	1.6 (0.9, 2.8)	0.08
III	1.2 (0.7, 2.1)	0.59	1.2 (0.7, 2.2)	0.54
IV	1.5 (0.9, 2.5)	0.13	1.9 (1.1, 3.2)	0.01
V	1.8 (1.2, 2.9)	0.01	2.2 (1.3, 3.4)	0.001
	Intellectual in	npairment		
No		Refe	erence	
Yes	0.5 (0.4, 0.8)	0.001	0.5 (0.3, 0.7)	<0.001
	Hearing im	pairment		
No		Refe	erence	
Yes	0.3 (0.1, 1.0)	0.05	0.9 (0.3, 3.1)	0.93
	Visual imp	airment		
No		Refe	erence	
Yes	0.6 (0.3, 1.2)	0.17	0.5 (0.2, 0.9)	0.02
	Speech Imp	pairment		
No		Refe	erence	
Yes	0.7 (0.5, 1.1)	0.12	0.7 (0.5, 1.0)	0.04
	Neurological t	ype of CP ¹		
Mono/hemiplegia		Refe	erence	
Diplegia	1.1 (0.5, 2.6)	0.76	1.2 (0.5, 2.9)	0.63
Quadriplegia	1.8 (1.1, 2.7)	0.01	2.0 (1.3, 3.1)	0.002
Dyskinesia	1.5 (0.6, 3.5)	0.37	1.2 (0.5, 3.0)	0.66

Table 4. Cont.

¹ There were no triplegic child and one Ataxic child in our cohort. Due to the smaller size, we have excluded these two categories from this unadjusted analysis. The p values for statistically significant differences are shown in bold.

Table 5. Factors associated with underweight and stunting among children with CP in Vietnam (unadjusted analysis).

Covariates	Underweig	ht	Stunting	Stunting	
Covariances	Odds Ratio (95% CI) (Adjusted)	p Value	Odds Ratio (95% CI) (Adjusted)	p Value	
	Age group	(years)			
less than 2	Reference				
2–4	2.1 (1.3, 3.5)	0.002	2.0 (1.3, 3.2)	0.003	
5 and more	3.1 (1.7, 6.1)	0.001	1.2 (0.6, 2.4)	0.58	
	Source of drin	king water			
Well water		Refe	rence		
Piped water	0.9 (0.5, 1.3)	0.52	0.8 (0.5, 1.3)	0.47	
Other sources (ponds/river/stream/lake)	1.4 (0.7, 2.9)	0.38	1.2 (0.5, 2.5)	0.65	

Covariates	Underweight		Stunting	
	Odds Ratio (95% CI) (Unadjusted)	p Value	Odds Ratio (95% CI) (Unadjusted)	p Value
Monthly family income (USD)				
Above 150		Refe	rence	
below 50	3.0 (1.5, 5.7)	0.001	1.3 (0.7, 2.4)	0.43
51-100	1.8 (1.0, 3.4)	0.06	1.2 (0.7, 2.2)	0.50
101–150	1.4 (0.7, 2.8)	0.34	1.1 (0.5, 2.0)	0.85
GMFCS level				
Ι	Reference			
II	1.3 (0.7, 2.6)	0.40	1.7 (0.9, 3.2)	0.11
III	1.6 (0.7, 3.4)	0.23	1.0 (0.4, 2.1)	0.93
IV	2.9 (1.4, 6.0)	0.005	1.9 (0.9, 4.0)	0.08
V	2.4 (1.1, 4.9)	0.02	2.1 (1.0, 4.3)	0.04
Intellectual impairment				
No		Reference		
Yes	1.3 (0.7, 2.3)	0.35	2.0 (1.1, 3.5)	0.02
Hearing impairment				
No		Reference		
Yes	1.9 (0.5, 7.3)	0.35	1.3 (0.3, 4.9)	0.72
Visual impairment				
No		Refe	rence	
Yes	1.5 (0.6, 3.4)	0.35	1.6 (0.7, 3.6)	0.26
Speech Impairment				
No		Refe	rence	
Yes	0.8 (0.5, 1.4)	0.48	0.7 (0.4, 1.1)	0.13
Neurological type of CP ¹				
Mono/hemiplegia	Reference			
Diplegia	1.2 (0.4, 3.2)	0.70	1.2 (0.4, 3.3)	0.69
Quadriplegia	1.2 (0.7, 2.1)	0.01	1.4 (0.8, 2.5)	0.23
Dyskinesia	1.3 (0.4, 3.9)	0.68	1.8 (0.6, 5.4)	0.30

Table 5. Cont.

¹ There were no triplegic child and one Ataxic child in our cohort. Due to the smaller size, we have excluded these two categories from this unadjusted analysis. The p values for statistically significant differences are shown in bold.

4. Discussion

Our novel study findings derive from an active hospital-based surveillance system and provide valuable insights on the burden and risk factors for malnutrition among children with CP in Vietnam. The proportion of children with malnutrition among our study cohort (i.e., 29% underweight and 30% stunted) was significantly higher than that reported in the general child population in Vietnam but lower than in other LMICs [10,14,25]. The reported burden of underweight from recent studies conducted in Bangladesh, Uganda and Ghana was 70%, 42% and 65% respectively [10,13,14] Our study utilized data from a hospital-based surveillance system. Considering the potential risk of representation bias of such a recruitment strategy, our findings may be an underestimate of the true burden of malnutrition among children with CP in Vietnam.

Vietnam has observed a steady improvement in the nutritional status of children over the past decade. The health policy and strategy framework of Vietnam demonstrates a commitment to reducing the burden of malnutrition among young children [26] The National Nutrition Strategy (NNS) of Vietnam (2011–2020) envisions a substantial reduction in malnutrition among children by the year 2020, aligning with the sustainable development goals (SDGs) [27] The cardinal role of nutrition in the

maintenance of health and wellbeing is well recognized and is demonstrated further by the emphasis on nutrition in the SDGs, particularly goals 1, 2 and 3. Following the launch of the NNS the Vietnamese government introduced several multisectoral initiatives including nutrition specific and nutrition sensitive intervention programs. The comparatively better nutritional status of children with CP in our study compared to those in other LMICs may be a result of these government initiatives to improve the nutritional outcomes of the population as a whole.

The characteristics of the malnourished children with CP identified through our active surveillance were similar to those from several other LMICs [10,13,14,25]. The majority of children in our study were from younger age groups. The significant association between age and underweight demonstrates the increased vulnerability of these young children to malnutrition as they get older. Similar findings were reported elsewhere [10]. Studies from other countries have also reported the relationship between malnutrition and age and motor severity of CP. Motor severity is likely to be associated with increased severity of oromotor dysfunction among children with CP, which acts as a barrier to self-feeding and/or adequate feeding. Evidence from HICs suggests that children with severe motor impairment for example, GMFCS level IV-V, often face difficulties with oromotor function [28,29]. These children also frequently have feeding difficulties such as choking, drooling, biting and gastro-esophageal reflux [2,3]. All these factors directly hamper regular food intake among these children. Studies also reported that children with severe motor impairment have altered energy requirements and body composition for example, reduced fat free mass, changes in total body fat [30]. This is consistent with our study findings, which revealed a significant association between the nutritional status of children with CP and motor severity and associated impairments.

It is evident that in the absence of appropriate interventions, children with CP are at high risk of developing malnutrition. In HICs, nasogastric tube feeding has been found to be beneficial for weight gain in children with CP [31]. However, this intervention is not widely available or utilized in LMICs, where there is aversion to increased risk of morbidity through adverse consequences for example, overfeeding, infection or aspiration [32]. Studies conducted in Bangladesh and Ghana demonstrate that improved nutritional status among children with CP can be achieved through training of caregivers with a specific focus on recommended feeding behavior [13,19].

Poor feeding practices such as incorrect positioning and nutritional inadequacy of the diet are often closely intertwined with the poor socio-economic status of families of children with CP. In our study, the burden of malnutrition among children with CP is higher among poorer families. This finding remained unchanged when adjustment was made for other socioeconomic and clinical characteristics. A similar pattern was observed in another LMICs [10]. The conceptual framework of malnutrition clearly shows that economic vulnerability directly and indirectly enhances the risk of malnutrition, comorbidity and mortality among children [9].

Study Limitations

Despite our considerable efforts, this study had limitations. First, it was a hospital-based study which imposed a potential risk for biased representation of children with CP in Vietnam. Thus, the findings must be interpreted with caution and consideration that the true scenario might differ from the reported findings. Second, we focused on selective anthropometric measurements to assess nutritional status. Although, utilization of other inexpensive anthropometric measurements for example, subscapular skinfold thickness, triceps skinfold thickness, mid upper arm circumference could provide more information of the overall nutritional status of children with CP, considering the availability of resources and to minimize errors in the data we only included the height and weight in this study. A detailed assessment, such as measurement of body fat, fat free mass or bone mineral density requires advanced technology for example, biometric impedance analysis, dual-energy X-ray absorptiometry. Assessment of the dietary intake pattern including caregiver's knowledge, capacity and practice regarding adequate feeding practices to the children with CP and detailed assessment of the level of oromotor dysfunction could potentially yield a better understanding of nutritional status

and establish possible causal relationships between risk factors identified in this study for example, socio-economic status, motor severity and malnutrition among children with CP. Third, we assessed the nutritional status of our study participants by comparing their anthropometric data to the WHO reference population. Recent evidence from HICs indicates that the growth curve of children with CP differs from the general population [33]. However, such CP specific and/or ethnicity specific data are not available for LMICs like Vietnam. Nevertheless, the method utilized in our study for the assessment of nutritional status is widely accepted in these settings.

5. Conclusions

Global evidence shows that malnutrition increases the risk of morbidity and mortality among children. This is intensified when malnutrition is accompanied by a disability such as CP. Although evidence from different countries has identified few common risk factors for malnutrition among children with CP, the burden and intensity is highly influenced by contextual factors. Considering the dearth of available evidence from LMICs, our study has generated new data that add substantially to current knowledge about the nutritional status of children with CP in Vietnam. These findings will guide the development of evidence-based training for health professionals and interventions for children with CP in Vietnam.

Author Contributions: Conceptualization, N.V.B.; G.K. and E.E.; Data curation, T.K.; R.D. and N.T.V.A.; Formal analysis, T.K. and I.J.; Funding acquisition, N.B.; N.V.B.; G.K. and E.E.; Investigation, N.T.H.G.; N.T.V.A.; T.Q.D. and N.V.B.; Methodology, N.V.B.; G.K.; E.E. and C.M.C.; Project administration, R.D.; Supervision, N.T.H.G.; T.Q.D.; N.B.; G.K. and E.E.; Writing—original draft, T.K. and I.J.; Writing—review & editing, T.K.; I.J.; R.D.; N.T.H.G.; N.T.V.A.; T.Q.D.; N.T.H.G.; N.T.V.A.; T.Q.D.; N.B.; G.K. and E.E.; Writing—original draft, T.K. and I.J.; Writing—review & editing, T.K.; I.J.; R.D.; N.T.H.G.; N.T.V.A.; T.Q.D.; C.M.C.; N.B.; N.V.B.; G.K. and E.E.

Funding: This study was funded by the Cerebral Palsy Alliance Research Foundation, Australia (PG03317 and PG6115). The study sponsor played no role in study design or the collection, analysis, and interpretation of data; in the writing of the report; and in the decision to submit the paper for publication.

Acknowledgments: EE is supported by a National Health and Medical Research Council Practitioner Fellowship (1021480). TK is supported by a Cerebral Palsy Alliance Research Foundation Career Development Grant (CDG04617).

Conflicts of Interest: The authors declare no conflict of interest.

References

- Rosenbaum, P.; Paneth, N.; Leviton, A.; Goldstein, M.; Bax, M.; Damiano, D.; Dan, B.; Jacobsson, B. A report: The definition and classification of cerebral palsy April 2006. *Dev. Med. Child Neurol. Suppl.* 2007, 109 (Suppl. 109), 8–14.
- 2. Aggarwal, S.; Chadha, R.; Pathak, R. Nutritional status and growth in children with cerebral palsy: A review. *Int. J. Med. Sci. Public Health* **2015**, *4*, 737–744. [CrossRef]
- 3. Aggarwal, S.; Chadha, R.; Pathak, R. Feeding Difficulties among Children with Cerebral Palsy: A Review. *Int. J. Health Sci. Res.* **2015**, *1*, 9–12.
- 4. Oskoui, M.; Coutinho, F.; Dykeman, J.; Jetté, N.; Pringsheim, T. An update on the prevalence of cerebral palsy: A systematic review and meta-analysis. *Dev. Med. Child Neurol.* **2013**, *55*, 509–519. [CrossRef]
- 5. ACPR. Report of the Australian Cerebral Palsy Register, Birth Years 1995–2012. Australia. Available online: https://www.ausacpdm.org.au/resources/australian-cerebral-palsy-register/ (accessed on 12 December 2018).
- Khandaker, G.; Muhit, M.; Karim, T.; Smithers-Sheedy, H.; Novak, I.; Jones, C.; Badawi, N. Epidemiology of Cerebral Palsy (CP) in Bangladesh: Findings from the first population-based CP register for children in a low-and middle-income country. *Dev. Med. Child Neurol.* 2019, 61, 601–609. [CrossRef]
- Forthun, I.; Strandberg-Larsen, K.; Wilcox, A.J.; Moster, D.; Petersen, T.G.; Vik, T.; Lie, R.T.; Uldall, P.; Tollånes, M.C. Parental socioeconomic status and risk of cerebral palsy in the child: Evidence from two Nordic population-based cohorts. *Int. J. Epidemiol.* 2018, 47, 1298–1306. [CrossRef]
- 8. Durkin, M.S. Socio-economic disparities and functional limitations of children with cerebral palsy. *Dev. Med. Child Neurol.* **2016**, *58*, 115. [CrossRef]

- 9. UNICEF. The State of World's Children. 1998. Available online: https://www.unicef.org/sowc/archive/ ENGLISH/The%20State%200f%20the%20World%27s%20Children%201998.pdf (accessed on 7 January 2019).
- Jahan, I.; Muhit, M.; Karim, T.; Smithers-Sheedy, H.; Novak, I.; Jones, C.; Badawi, N.; Khandaker, G. What makes children with cerebral palsy vulnerable to malnutrition? Findings from the Bangladesh cerebral palsy register (BCPR). *Disabil. Rehabil.* 2019, 41, 2247–2254. [CrossRef]
- Strand, K.M.; Dahlseng, M.O.; Lydersen, S.; Ro, T.B.; Finbraten, A.K.; Jahnsen, R.B.; Andersen, G.L.; Vik, T. Growth during infancy and early childhood in children with cerebral palsy: A population-based study. *Dev. Med. Child Neurol.* 2016, *58*, 924–930. [CrossRef]
- 12. Perenc, L.; Przysada, G.; Trzeciak, J. Cerebral palsy in children as a risk factor for malnutrition. *Ann. Nutr. Metab.* **2015**, *66*, 224–232. [CrossRef]
- Polack, S.; Adams, M.; O'banion, D.; Baltussen, M.; Asante, S.; Kerac, M.; Gladstone, M.; Zuurmond, M. Children with cerebral palsy in Ghana: Malnutrition, feeding challenges and caregiver quality of life. *Dev. Med. Child Neurol.* 2018, 60, 914–921. [CrossRef]
- 14. Kakooza-Mwesige, A.; Tumwine, J.K.; Eliasson, A.C.; Namusoke, H.K.; Forssberg, H. Malnutrition is common in Ugandan children with cerebral palsy, particularly those over the age of five and those who had neonatal complications. *Acta Paediatr.* **2015**, *104*, 1259–1268. [CrossRef]
- Sung, K.H.; Chung, C.Y.; Lee, K.M.; Cho, B.C.; Moon, S.J.; Kim, J.; Park, M.S. Differences in Body Composition According to Gross Motor Function in Children with Cerebral Palsy. *Arch. Phys. Med. Rehabil.* 2017, 98, 2295–2300. [CrossRef]
- 16. Power, R.; King, C.; Muhit, M.; Heanoy, E.; Galea, C.; Jones, C.; Badawi, N.; Khandaker, G. Health-related quality of life of children and adolescents with cerebral palsy in low-and middle-income countries: A systematic review. *Dev. Med. Child Neurol.* **2018**, *60*, 469–479. [CrossRef]
- Jahan, I.; Karim, T.; Das, M.C.; Muhit, M.; Mcintyre, S.; Smithers-Sheedy, H.; Badawi, N.; Khandaker, G. Mortality in children with cerebral palsy in rural Bangladesh: A population-based surveillance study. *Dev. Med. Child Neurol.* 2019. [CrossRef]
- 18. Karim, T.; Muhit, M.; Khandaker, G. Interventions to prevent respiratory diseases-Nutrition and the developing world. *Paediatr. Respir. Rev.* 2017, 22, 31–37. [CrossRef]
- Adams, M.S.; Khan, N.Z.; Begum, S.A.; Wirz, S.L.; Hesketh, T.; Pring, T.R. Feeding difficulties in children with cerebral palsy: Low-cost caregiver training in Dhaka, Bangladesh. *Child Care Health Dev.* 2012, *38*, 878–888. [CrossRef]
- 20. Ferluga, E.D.; Sathe, N.A.; Krishnaswami, S.; Mcpheeters, M.L. Surgical intervention for feeding and nutrition difficulties in cerebral palsy: A systematic review. *Dev. Med. Child Neurol.* **2014**, *56*, 31–43. [CrossRef]
- 21. Khandaker, G.; Van Bang, N.; Dũng, T.Q.; Giang, N.T.; Chau, C.M.; Van Anh, N.T.; Van Thuong, N.; Badawi, N.; Elliott, E.J. Protocol for hospital based-surveillance of cerebral palsy (CP) in Hanoi using the Paediatric Active Enhanced Disease Surveillance mechanism (PAEDS-Vietnam): A study towards developing hospital-based disease surveillance in Vietnam. *BMJ Open* **2017**, *7*, e017742. [CrossRef]
- 22. Cans, C. Surveillance of cerebral palsy in Europe: A collaboration of cerebral palsy surveys and registers. *Dev. Med. Child Neurol.* **2000**, *42*, 816–824. [CrossRef]
- 23. World Health Organization. WHO Global Database on Child Growth and Malnutrition; World Health Organization: Geneva, Switzerland, 1997; Available online: http://apps.who.int/iris/bitstream/10665/63750/1/WHO_NUT_97.4.pdf (accessed on 7 January 2019).
- 24. Stevenson, R.D. Use of segmental measures to estimate stature in children with cerebral palsy. *Arch. Pediatr. Adolesc. Med.* **1995**, *149*, 658–662. [CrossRef]
- 25. Caram, A.L.A.; Morcillo, A.M.; Costa-Pinto, E.A.L. Nutritional status of children with cerebral palsy in a Brazilian tertiary-care teaching hospital. *Dev. Med. Child Neurol.* **2008**, *50*, 956–960. [CrossRef]
- 26. World Bank Data. 2017. Available online: https://data.worldbank.org/indicator/SH.STA.STNT.ZS?view=chart (accessed on 7 January 2019).
- 27. The National Nutrition Strategy for 2011–2020, with a Vision Toward 2030. Available online: https://extranet.who.int/nutrition/gina/sites/default/files/VNM%202011%202.%20National%20Nutrition% 20%20Strategy%202011-2020.pdf (accessed on 7 January 2019).
- 28. Herrera-Anaya, E.; Angarita-Fonseca, A.; Herrera-Galindo, V.M.; Martinez-Marin, R.D.P.; Rodriguez-Bayona, C.N. Association between gross motor function and nutritional status in children with cerebral palsy: A cross-sectional study from Colombia. *Dev. Med. Child Neurol.* **2016**, *58*, 936–941. [CrossRef]

- 29. Dahlseng, M.O.; Finbråten, A.K.; Júlíusson, P.B.; Skranes, J.; Andersen, G.; Vik, T. Feeding problems, growth and nutritional status in children with cerebral palsy. *Acta Paediatr.* **2012**, *101*, 92. [CrossRef]
- Benfer, K.A.; Weir, K.A.; Bell, K.L.; Ware, R.S.; Davies, P.S.; Boyd, R.N. Food and fluid texture consumption in a population-based cohort of preschool children with cerebral palsy: Relationship to dietary intake. *Dev. Med. Child Neurol.* 2015, 57, 1056–1063. [CrossRef]
- 31. Brooks, J.; Day, S.; Shavelle, R.; Strauss, D. Low weight, morbidity and mortality in children with cerebral palsy: New clinical growth charts. *Pediatr. Eng. Ed.* **2011**, *128*, e299. [CrossRef]
- 32. Bell, K.L.; Samson-Fang, L. Nutritional management of children with cerebral palsy. *Eur. J. Clin. Nutr.* 2013, 67, S13. [CrossRef]
- Henderson, R.C.; Grossberg, R.I.; Matuszewski, J.; Menon, N.; Johnson, J.; Kecskemethy, H.H.; Vogel, L.; Ravas, R.; Wyatt, M.; Bachrach, S.J.; et al. Growth and nutritional status in residential center versus home-living children and adolescents with quadriplegic cerebral palsy. *J. Pediatr.* 2007, 151, 161–166. [CrossRef]



© 2019 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).

5.4 CHAPTER SYNOPSIS

In this chapter, findings regarding children with CP from another LMIC, Vietnam, have been described. These studies used a different epidemiological tool, hospital-based surveillance, to collect data from Vietnam. Although not population representative, these studies have generated valuable data on children with CP.

The children with CP in Vietnam predominantly had spastic CP (95.2%). Most (69.6%) were quadriplegic and 60.3% were GMFCS level III-V. 28.9% underweight and 29.0% stunted; children with quadriplegia, GMFCS level IV to V and/or aged >5 years were more vulnerable to malnutrition. These findings will inform clinical practice and policy in low resource settings, particularly in hospital-based settings.

Detailed discussion on the implications of these findings and the overall thesis are described in the following section.

DISCUSSION AND RECOMMENDATIONS FOR PRACTICE, POLICY AND RESEARCH

In this section, the findings from the doctoral research and their bearing on practice, policy and research are discussed.

THESIS SUMMARY: BRIDGING THE GAPS

The overall aim of this thesis was to describe the clinical characteristics of CP, with concurrent exploration of measures for early identification and intervention for children with CP in low resource settings. The areas covered in the thesis in relation to each of the specific aims of the thesis are outlined below,

Aim (i) better understand the clinical characteristics of children with CP in Bangladesh with a specific focus on hip dysplasia and epilepsy

Hip dysplasia and epilepsy among children with CP in Bangladesh have been described in *Chapters 1 and 2*. The BCPR served as the sampling frame for both the studies.

Aim (ii) explore pathways to improve age at diagnosis of CP in Bangladesh

Chapter 3 delineates pathways for early diagnosis of CP in Bangladesh. It includes a publication outlining current practice and eludes into the future of early diagnosis of CP in LMICs.

Aim (iii) enhancing interventions for children with CP and their primary caregivers in Bangladesh through a quasi-experimental study assessing the outcome of an existing program in Bangladesh

Chapter 4 describes the outcome of an existing intervention program for children from BCPR. In additional to the child outcomes i.e. motor, speech and communication, this study reports the wellbeing of the primary caregivers.

Aim (iv) describe CP in another low resource setting using an alternative epidemiological tool through a hospital-based surveillance in Vietnam

Chapter 5 describes that state of evidence on CP in another LMIC, Vietnam. This chapter includes description of clinical characteristics, aetiologies, rehabilitation, education, and detailed nutritional status of children with CP through a hospital-based surveillance modelled on Paediatric Active Enhanced Disease Surveillance (PAEDS).

CEREBRAL PALSY REGISTERS IN LOW RESOURCE SETTINGS

CP registers are emerging in LMICs and the ensuing increased availability of data from lowresource settings has advanced our knowledge on CP substantially.(4, 33, 48, 49). The findings from recent studies in Nepal, Indonesia and Uganda confirm a marked contrast to the data from HICs.(22, 50, 51) In LMICs, the prevalence of CP is higher, diagnosis is delayed, there is an overwhelming number of children with greater motor severity of impairment, the burden of associated impairments is high, few have access to education and rehabilitation.(4, 14, 51) The rates of the prevailing aetiologies and risk factors in LMICs (i.e. intrapartum related neonatal respiratory depression, neonatal encephalopathy, neonatal jaundice, infections) differ considerably from the HICs that have realized tremendous success in reducing the rates of CP by addressing such factors.(50, 52) Thus, reaffirming that prioritization of equitable access to quality perinatal care for effective prevention of CP is imperative.(15)

BANGLADESH CERERBAL PALSY REGISTER: ONWARDS AND UPWARDS

The BCPR study and the translational research facilitated by it demonstrates the tremendous potential CP registers hold, even in low resource settings. Since the publication of the BCPR study findings in 2019 described previously in this thesis,(33) the study activities have been expanded across six districts of Bangladesh to include 3135 children with CP to date with view of developing a national CP register; updated findings from the ongoing surveillance have been reported in two recent publications.(4, 53). The BCPR has served as the sampling frame for three studies described in *Chapters 1,2 and 4* of this thesis.(54). Since its inception in 2015, it has additionally enabled the conduct of a range of studies reporting immunization, nutrition, rehabilitation, health related quality of life and sexual wellbeing of children and adolescents with CP and their primary caregivers in Bangladesh.(40, 55-59). Furthermore, the BCPR has facilitated local capacity building and continues to foster emerging researchers in an important area. In addition to this thesis, the expansive work related to the BCPR study has resulted in the following doctoral research,

 Adolescents with cerebral palsy in rural Bangladesh: Health-related quality of life, mental health and sexual and reproductive health |*Dr Rosalie Power, University of Sydney (PhD completed)*

- The epidemiology, prevention and control of malnutrition among children with Cerebral Palsy in low- and middle-income countries | *Israt Jahan, final year PhD student,* CQUniversity
- Developing a sustainable model of early intervention and rehabilitation service for children with CP in LMICs: translational research using the Global LMIC CP Register | Mahmudul Hassan Al Imam, final year PhD student, CQUniversity

The BCPR has enabled generation of high quality population-based data which is the one of the best sources of data for epidemiological exploration.(24) and enabled estimation of prevalence and extrapolation at national level.(33) Such information is invaluable to program planning and policy development through an evidence-based approach.

HOSPITAL-BASED SURVEILLANCE IN VIETNAM: NEXT STEPS

Methodological differences need to be accounted for in interpretation and comparison of study findings from different parts of the world.(6) In contrast to Bangladesh, the findings from the hospital-based surveillance show a glimpse of common risk factors and clinical characteristics among children with CP in Vietnam. Although not population representative, these findings provide the necessary baseline for further exploration in hospital-based settings which is often the ideal setting for certain strategies for prevention, diagnosis, and intervention for CP. The PAEDs system on which this surveillance is modelled enabled data collection as part of routine services through an opportunist purposive recruitment strategy and can be easily scaled up to generate best available data in absence of population data.(60)

The findings and the ongoing collaboration through the studies outline in *Chapter 5* have stimulated further research in Vietnam. This has resulted in an upcoming project titled, 'Evidence based approach to surveillance, early diagnosis and intervention of children with cerebral palsy in Vietnam' [Research Foundation of Cerebral Palsy Alliance: PRG08419, Grant amount: AU\$200,230.00 over 3 years]. As one of the chief investigators on this study, I aim to build on my doctoral research in Vietnam through this project as part of my postdoctoral research. The broader vision is to replicate the successes in early detection of CP reported in HICs. This will complement the ongoing national efforts in Vietnam, including the

implementation of the National Rehabilitation Guideline for the Management of Children with CP (<u>https://hivndrive.wixsite.com/hivn/rehabguidelines</u>).

ASSOCIATED IMPAIRMENTS

Although CP is defined as a motor disability, it is commonly associated with a range of impairments.(61) A systematic review on the clinical prognostic messages for CP outlines, "Among children with cerebral palsy, 3 in 4 were in pain; 1 in 2 had an intellectual disability; 1 in 3 could not walk; 1 in 3 had a hip displacement; 1 in 4 could not talk; 1 in 4 had epilepsy; 1 in 4 had a behavior disorder; 1 in 4 had bladder control problems; 1 in 5 had a sleep disorder; 1 in 5 dribbled; 1 in 10 were blind; 1 in 15 were tube-fed; and 1 in 25 were deaf".(62) While it is important to address each of these concerns for children with CP and their families, the following discussion and recommendations focus on the associated impairments described in detail in this thesis i.e. hip dysplasia (*Chapter 1*), epilepsy (*Chapter 2*), and malnutrition (*Chapter 5*).

Hip dysplasia

Hip displacement often results in progressive functional limitation.(63) However, with timely detection and intervention, this can be prevented and the secondary complications, such as hip dislocation, pain, functional limitations, and poor quality of life, can be reduced;(64) through period clinical examinations and radiography i.e. routine hip surveillance among children with CP.(65) There are now established clinical guidelines for hip surveillance among children with CP.(64) The hip surveillance program in Sweden demonstrated the impact of hip surveillance i.e. improved wellbeing of children with CP and their families, and has virtually eradicated hip dislocation among children with CP.(66)

The Australian hip surveillance guideline has been rationalized recently informed by temporal data. It informs the frequency of radiographic examination among lower risk groups and the continuation of surveillance into adulthood for adolescents with identified risk factors.(64, 67, 68) This reaffirms the importance of monitoring temporal trends in hip dysplasia to inform clinical practice and interventions for CP. Chapter 1 of this thesis provides baseline data that will inform the establishment of a hip surveillance program in Bangladesh and other low

resources settings where similar patterns of motor severity of CP and hip dysplasia are observed.(4, 54)

Epilepsy

Poor epilepsy control results in poor health outcomes for children with CP and greater caregiver burden.(69) As outlined in *Chapter 2* epilepsy control among children with CP is challenging. This is particularly challenging in remote settings where retention of qualified/trained health professionals in poor.(70) Therefore, as an extension of the work described in Chapter 2, a simple guideline for epilepsy management among children with CP has been developed and translated to local language (Appendix B). The aim is to make the guideline available to physicians and supporting health workers in remote settings in Bangladesh to ensure better epilepsy control even in absence of highly skilled professionals.

Furthermore, intellectual impairment is common in children with epilepsy and CP. Antecedents of epilepsy and intellectual impairment overlap and often reflect greater severity of brain injury and motor impairment.(71) Both epilepsy and intellectual impairment have significant bearing on long-term prognosis, education, employment, social integration and overall wellbeing of individuals with CP.(72, 73) Further research into management of these related conditions through a comprehensive life-course health development approach in low resource settings is essential.

Malnutrition

Malnutrition among children with CP is compounded by a range of risk factors and associated impairments.(74) Poor nutritional status is often both a cause and consequence of more severe outcomes of CP. Malnourished children with CP are also at greater risk of mortality.(75) Nutritional interventions need to be integrated into regular medical check-ups and rehabilitation programs for children with CP and tailored to address the impact of the co-occurring conditions and heterogeneity of CP. Development of caregiver knowledge and capacity is instrumental to the maintenance of optimum nutritional status among children with CP, particularly in LMIC settings.(76, 77) Due to prevailing health inequities and their

impact on access to services, (78) such equity focused intervention approaches are warranted in low resource settings.

EARLY DETECTION

The 2017 clinical guideline on the early diagnosis of CP outlines best practice tools to support diagnosis.(79, 80) It states that it is now possible to accurately diagnose CP before six months' corrected age through a combination of evidence-based tools, i.e., General Movements Assessment and Hammersmith Infant Neurological Examination, and neuroimaging. Although there have been remarkable strides made in the sphere of early diagnosis of CP in the past decade, there is an increasing shift towards ensuring intervention as soon as a child is identified to be at risk of CP.(47) An interim high risk of CP clinical diagnosis is recommended when confirmation of diagnosis is delayed.(46) With earlier diagnosis and the availability of the latest international clinical practice guideline for intervention for children younger than two years with or at risk of CP, there is tremendous potential to harness neuroplasticity to enhance function and participation of children with CP.(81) There is a dire need for local capacity building to integrate the use of these tools within the existing structures in LMICs.

Finding from BCPR show that diagnosis of CP is substantially delayed in Bangladesh. Similar patterns have been reported in studies from other LMICs.(48, 82) This is unsurprising in absence of the use of GMA, HINE and neuroimaging in these settings. In contrast to HICs, diagnosis of CP in LMICs is delayed beyond the period when children with CP show early signs of activity limitations and associated delay in developmental milestones from six months of age.(83) In HICs waiting for delayed motor developmental milestones may be too late. This however is often the first sign of concern for parents of children with CP. The DMC described in *Chapter 3* can therefore serve as a valuable screening tool to ensure referral for detailed clinical assessment by a qualified clinician for an interim high risk of CP clinical diagnosis based on risk factors and clinical presentation. With the increasing shift towards identification of "at risk" of CP to ensure early intervention within the period of optimum neuroplasticity, further research to explore the feasibility of the widely used criteria for the interim high risk of CP clinical diagnosis in low resource settings is needed.

These efforts however are not a substitute to the best practice clinical guidelines.(46) Contemporaneous efforts to support the implementation of the use of GMA, HINE, and neuroimaging for early accurate diagnosis of CP must be made in LMICs. My postdoctoral research will explore the feasibility of the use of these tools in low resource settings. This work will utilise data from an ongoing study titled, *Early Detection and Early Natural History of Cerebral Palsy in Bangladesh*, and an upcoming study titled, *Evidence based approach to surveillance, early diagnosis and intervention of children with cerebral palsy in Vietnam*. Both studies are an extension of the ongoing collaborations in Bangladesh and Vietnam and are funded by the Research Foundation of Cerebral Palsy Alliance. These studies are expected to support the development of a cost-effective scalable model for early detection of children with CP in low resource settings.

EARLY INTERVENTION

As outlined previously, delayed diagnosis is one of the major barriers to early intervention in LMICs. This further contributes to poor outcomes for children with CP and the caregiving burden of the families. In *Chapter 4*, outcomes of a community-based intervention program for children identified with CP have been assessed and described through a quasi-experimental study. Although, the findings from this quasi-experimental study are promising, further exploration through a randomized controlled trial is needed. This work underpins the need for such caregiver led community-based rehabilitation program for children with CP in LMICs such as Bangladesh, particularly in the rural settings, where services and trained health workers are scarce.(70)

LIMITATIONS AND STRENGTHS

Specific limitations and strengths of each study have been discussed in the publications included within the thesis in *Chapter 1,2,3,4* and 5. Here, the limitations and strengths of the overall thesis is described.

Firstly, one of the major challenges of a population-based surveillance is attainment of complete case ascertainment. When conducted conventionally over a short period of time, KIM has a 77.6% case ascertainment rate compared to door-to-door survey for identification
of children with physical impairments such as CP in LMICs.(37) However, it is likely that majority of children living with CP in the BCPR surveillance area during the collection of baseline information for the BCPR have been identified due to the continuing engagement of the CSF Global team on site for a range of research projects and services. Secondly, diagnosis of CP is substantially delayed, it is likely that complete ascertainment has not been possible for children with CP below the average age of diagnosis of CP. Furthermore, mortality is high among children with CP in LMICs.(75) A substantial proportion of children with CP, particularly those with more severe CP, would have had premature deaths before being enrolled in the CP register which limits accurate identification of etiological factors. Thirdly, confirmation of precise diagnosis and description of CP is challenging among children under the age of five years due to the changes in the clinical presentation and CP subtype during this period. Fourthly, a proportion of children with mild CP may not have been identified for inclusion in both the methods described; In Bangladesh, KIM may have missed children with mild CP and in Vietnam, children with mild CP may not have presented at the hospital-based services.

Important strengths of this doctoral research are that it describes high quality data from two robust methods. Studies from Bangladesh use the BCPR, a population-based surveillance as the sampling frame, therefore generating best available epidemiological data. The BCPR has served as sampling frame for several observational studies and RCTs. These have been an indelible contribution to the knowledge base on CP in LMICs. In contrast, the hospital-based surveillance in Vietnam is modelled on a highly successful method in Australia, the PAEDS mechanism. This method can potentially be utilized for the surveillance of other conditions and is therefore a valuable epidemiological resource for studies with aims from the BCPR studies. Existing hospital settings are often ideal sites for piloting approaches novel to different parts of the world. This can be achieved through integration of evidence-based approaches to early detection and intervention for children with CP in regular medical check-ups and available health services in hospitals. Therefore, baseline hospital-based data is the necessary first step towards this vision.

OVERALL RECOMMENDATIONS FOR PRACTICE, POLICY AND FUTURE RESEARCH

Global Low-and Middle-income country CP Register (GLM-CPR) is the first multi-country epidemiological exploration of data on children with CP in LMICs.(4) Findings from the GLM-CPR reaffirm the urgent need for identification and implementation of appropriate strategies for prevention, early detection, and intervention for children with CP in LMICs. Greater motor severity of CP has been reported from LMICs across the world. This is linked to a range of factors including delayed diagnosis, delayed intervention, poor access to rehabilitation services, unmet equipment needs and the interplay of a myriad of socioeconomic factors with the severity of impairments.(4, 53)

With the advent of understanding of the etiopathogenesis of CP, there are now clear pathways for prevention of CP.(84-86) There is ample evidence on evidence-based approaches to early detection and intervention for children with CP.(46, 81) There is documented success from HICs with long standing CP registers and related programs e.g. perinatal care for prevention of CP to reduce incidence(15) and hip surveillance programs to avert complications.(64) There is an urgent call for action to extend the implementation of the numerous proven strategies to LMICs. International multidisciplinary efforts are warranted to address the global burden of CP with the shared vision to prevent CP and support individuals living with CP and their families with an equity focused approach.

There is an increasing focus on neurological interventions for CP with the recent advances in the knowledge on neuroplasticity of the developing brain.(3) However, as a lifelong condition, the approach to interventions for CP need to be through a life-course perspective.(2) As described by Chabrier et al, "This new paradigm should guide individual, familial and collective care interventions and research strategies beyond the scope of CP".(87) Further research is needed in the spheres of school readiness, employment, and social integration to ensure that children with CP can grow up to be adults with future opportunities, dignity, and meaningful relationships. Supporting the often-overlooked families and caregivers of individuals with CP, who bear the majority of the caregiver burden, particularly in low resource settings, will also be instrumental to this vision.

Summary of recommendations of practice, policy and future research

- Surveillance of CP in LMICs to identify gaps, monitor temporal trends, facilitate further research, and rationalize guidelines for low resource settings
- Equity focused approach to prevention, early diagnosis and intervention for CP and common associated impairments such as hip dysplasia and epilepsy
- Promotion of true disability inclusive development through a life course approach to ensure best outcomes and wellbeing of individuals with CP into adulthood
- Interventions to support families and caregivers of individuals living with CP who bear majority of the burden, particularly in low resource settings
- Foster international collaborations and knowledge transfer to address the global burden of CP as a shared goal.

CONCLUSION

This thesis provides valuable data for a better understanding of CP and maps the path towards the establishment and use of national CP registers in low resource settings. Strategies for early detection and a promising early intervention program for children with CP and their families in Bangladesh, a typical LMIC, have also been described. Furthermore, using hospitalbased surveillance, the clinical characteristics of children with CP have been reported in Vietnam. This doctoral research identified important knowledge and service gaps for children with CP and their families in LMICs and addressed some of them.

The gradually increasing shift of focus towards countries where the burden of CP is higher and greater risk prevails; the reported decline of prevalence, severity, and co-occurring conditions in HICs; and the emerging CP registers and research in LMICs is encouraging. There are reasons to be optimistic that these changes will be coupled by the incorporation of preventive, diagnostic and rehabilitation efforts in LMICs while the search for a cure for CP continues. With concerted efforts the successes observed in HICs can be replicated in low resource settings with a shared vision to address the global burden of CP.

REFERENCES

1. Rosenbaum P, Paneth N, Leviton A, Goldstein M, Bax M, Damiano D, et al. A report: the definition and classification of cerebral palsy April 2006. Developmental Medicine & Child Neurology Suppl 2007;49(109):8-14.

2. Rosenbaum PL, Dan B. The continuing evolution of "Cerebral Palsy". Annals of physical and rehabilitation medicine. 2019.

3. Cieza A, Causey K, Kamenov K, Hanson SW, Chatterji S, Vos T. Global estimates of the need for rehabilitation based on the Global Burden of Disease study 2019: a systematic analysis for the Global Burden of Disease Study 2019. The Lancet. 2020;396(10267):2006-17.

4. Jahan I, Muhit M, Hardianto D, Laryea F, Chhetri AB, Smithers-Sheedy H, et al. Epidemiology of cerebral palsy in low-and middle-income countries: preliminary findings from an international multi-centre cerebral palsy register. Developmental Medicine & Child Neurology. 2021.

5. Kakooza-Mwesige A, Andrews C, Peterson S, Mangen FW, Eliasson AC, Forssberg H. Prevalence of cerebral palsy in Uganda: a population-based study. The Lancet Global Health. 2017;5(12):e1275-e82.

6. Dan B, Paneth N. Making sense of cerebral palsy prevalence in low-income countries. The Lancet Global Health. 2017;5(12):e1174-e5.

7. Gladstone M. A review of the incidence and prevalence, types and aetiology of childhood cerebral palsy in resource-poor settings. Ann Trop Paediatr. 2010;30(3):181-96.

8. Rosenbaum PL, Palisano RJ, Bartlett DJ, Galuppi BE, Russell DJ. Development of the Gross Motor Function Classification System for cerebral palsy. Dev Med Child Neurol. 2008;50(4):249-53.

9. Korzeniewski SJ, Slaughter J, Lenski M, Haak P, Paneth N. The complex aetiology of cerebral palsy. Nat Rev Neurol. 2018;14(9):528-43.

10. McIntyre S. Causal pathways to cerebral palsy in term and near-term singletons: Analysis of a total population case control study 2013.

11. McIntyre S, Taitz D, Keogh J, Goldsmith S, Badawi N, Blair E. A systematic review of risk factors for cerebral palsy in children born at term in developed countries. Dev Med Child Neurol. 2013;55(6):499-508.

12. Duke R, Torty C, Nwachukwu K, Ameh S, Kim M, Eneli N, et al. Clinical features and aetiology of cerebral palsy in children from Cross River State, Nigeria. Arch Dis Child. 2020;105(7):625-30.

13. Namaganda LH, Almeida R, Kajungu D, Wabwire-Mangen F, Peterson S, Andrews C, et al. Excessive premature mortality among children with cerebral palsy in rural Uganda: A longitudinal, population-based study. PLOS ONE. 2021;15(12):e0243948.

14. Kakooza-Mwesige A, Andrews C, Peterson S, Wabwire Mangen F, Eliasson AC, Forssberg H. Prevalence of cerebral palsy in Uganda: a population-based study. The Lancet Global Health. 2017;5(12):e1275-e82.

15. Badawi N, McIntyre S, Hunt RW. Perinatal care with a view to preventing cerebral palsy. Dev Med Child Neurol. 2021;63(2):156-61.

16. Hollung SJ, Vik T, Lydersen S, Bakken IJ, Andersen GL. Decreasing prevalence and severity of cerebral palsy in Norway among children born 1999 to 2010 concomitant with improvements in perinatal health. European Journal of Paediatric Neurology. 2018;22(5):814-21.

17. Durkin MS, Maenner MJ, Benedict RE, Van Naarden Braun K, Christensen D, Kirby RS, et al. The role of socio-economic status and perinatal factors in racial disparities in the risk of cerebral palsy. Dev Med Child Neurol. 2015;57(9):835-43.

18. Durkin MS. Socio-economic disparities and functional limitations of children with cerebral palsy. Dev Med Child Neurol. 2016;58(2):115.

19. LeBrun DG, Banskota B, Banskota AK, Rajbhandari T, Baldwin KD, Spiegel DA. Socioeconomic Status Influences Functional Severity of Untreated Cerebral Palsy in Nepal: A Prospective Analysis and Systematic Review. Clin Orthop Relat Res. 2019;477(1):10-21.

20. Woolfenden S, Galea C, Smithers-Sheedy H, Blair E, McIntyre S, Reid S, et al. Impact of social disadvantage on cerebral palsy severity. Dev Med Child Neurol. 2019;61(5):586-92.

21. Group A. Report of the Australian Cerebral Palsy Register, Birth Years 1993-2003. Annual report. Australia: ACPR; 2009 Dec 2009.

22. Smithers-Sheedy H, McIntyre S, Gibson C, Meehan E, Scott H, Goldsmith S, et al. A special supplement: findings from the Australian Cerebral Palsy Register, birth years 1993 to 2006. Dev Med Child Neurol. 2016;58 Suppl 2:5-10.

23. Sellier E, Platt MJ, Andersen GL, Krägeloh-Mann I, De La Cruz J, Cans C, et al. Decreasing prevalence in cerebral palsy: a multi-site European population-based study, 1980 to 2003. Developmental Medicine & Child Neurology. 2016;58(1):85-92.

24. Porta M. A dictionary of epidemiology: Oxford university press; 2014.

25. Thygesen LC, Ersbøll AK. When the entire population is the sample: strengths and limitations in register-based epidemiology. European journal of epidemiology. 2014;29(8):551-8.

26. Goldsmith S, McIntyre S, Smithers-Sheedy H, Blair E, Cans C, Watson L, et al. An international survey of cerebral palsy registers and surveillance systems. Developmental Medicine & Child Neurology. 2016;58:11-7.

27. BBS. POVERTY MAPS OF BANGLADESH 2016. 2019.

28. The World Bank Group. The World Bank In Bangladesh 2021 [Available from: <u>https://www.worldbank.org/en/country/bangladesh/overview</u>.

 Ministry of Health and Family Welfare. Health Bulletin 2017. Dhaka, Bangladesh2017.
 Chowdhury AM, Bhuiya A, Chowdhury ME, Rasheed S, Hussain Z, Chen LC. The Bangladesh paradox: exceptional health achievement despite economic poverty. Lancet.

2013;382(9906):1734-45.

31. World Health Organization. Bangladesh health system review. Manila: WHO Regional Office for the Western Pacific; 2015. Report No.: 9290617055.

32. Khandaker G, Smithers-Sheedy H, Islam J, Alam M, Jung J, Novak I, et al. Bangladesh Cerebral Palsy Register (BCPR): a pilot study to develop a national cerebral palsy (CP) register with surveillance of children for CP. BMC neurology. 2015;15(1):173.

33. Khandaker G, Muhit M, Karim T, Smithers-Sheedy H, Novak I, Jones C, et al. Epidemiology of cerebral palsy in Bangladesh: a population-based surveillance study. Dev Med Child Neurol. 2019;61(5):601-9.

34. BBS. Bangladesh Population and Housing Census 2011, National Report Volume 2: Union Statistics. Dhaka, Bangladesh: BBS; 2014.

35. Cans C. Surveillance of cerebral palsy in Europe: a collaboration of cerebral palsy surveys and registers. Developmental Medicine & Child Neurology. 2000;42(12):816-24.

36. Group ACPR, Badawi N, Balde I, Goldsmith S, Karlsson P, McIntyre S, et al. Australia and the Australian Cerebral Palsy Register for the birth cohort 1993 to 2006. Developmental Medicine & Child Neurology. 2016;58:3-4.

37. Muhit MA, Shah SP, Gilbert CE, Hartley SD, Foster A. The key informant method: a novel means of ascertaining blind children in Bangladesh. Br J Ophthalmol. 2007;91(8):995-9.

38. Rosenbaum PL, Palisano RJ, Bartlett DJ, Galuppi BE, Russell DJ. Development of the Gross Motor Function Classification System for cerebral palsy. Developmental Medicine & Child Neurology. 2008;50(4):249-53.

39. Eliasson A-C, Krumlinde-Sundholm L, Rösblad B, Beckung E, Arner M, Öhrvall A-M, et al. The Manual Ability Classification System (MACS) for children with cerebral palsy: scale development and evidence of validity and reliability. Developmental Medicine & Child Neurology. 2006;48:549–54.

40. Jahan I, Muhit M, Karim T, Smithers-Sheedy H, Novak I, Jones C, et al. What makes children with cerebral palsy vulnerable to malnutrition? Findings from the Bangladesh cerebral palsy register (BCPR). Disabil Rehabil. 2019;41(19):2247-54.

41. Hadders-Algra M. Early diagnosis and early intervention in cerebral palsy. Front Neurol. 2014;5:185.

42. Khandaker G, Van Bang N, Dũng TQ, Giang NTH, Chau CM, Van Anh NT, et al. Protocol for hospital based-surveillance of cerebral palsy (CP) in Hanoi using the Paediatric Active Enhanced Disease Surveillance mechanism (PAEDS-Vietnam): a study towards developing hospital-based disease surveillance in Vietnam. BMJ Open. 2017;7(11):e017742.

43. Karim T, Dossetor R, Huong Giang NT, Dung TQ, Son TV, Hoa NX, et al. Data on cerebral palsy in Vietnam will inform clinical practice and policy in low and middle-income countries. Disability and Rehabilitation. 2020:1-8.

44. Karim T, Jahan I, Dossetor R, Giang NTH, Van Anh NT, Dung TQ, et al. Nutritional Status of Children with Cerebral Palsy-Findings from Prospective Hospital-Based Surveillance in Vietnam Indicate a Need for Action. Nutrients. 2019;11(9).

45. Robin J, Graham HK, Baker R, Selber P, Simpson P, Symons S, et al. A classification system for hip disease in cerebral palsy. Dev Med Child Neurol. 2009;51(3):183-92.

46. Novak I, Morgan C, Adde L, Blackman J, Boyd RN, Brunstrom-Hernandez J, et al. Early, accurate diagnosis and early intervention in cerebral palsy: advances in diagnosis and treatment. JAMA pediatrics. 2017;171(9):897-907.

47. McIntyre S, Morgan C, Walker K, Novak I. Cerebral palsy—don't delay. Developmental disabilities research reviews. 2011;17(2):114-29.

Jahan I, Al Imam MH, Karim T, Muhit M, Hardianto D, Das MC, et al. Epidemiology of cerebral palsy in Sumba Island, Indonesia. Developmental Medicine & Child Neurology. 2020.
Karim T, Dossetor R, Huong Giang NT, Dung TQ, Son TV, Hoa NX, et al. Data on cerebral

palsy in Vietnam will inform clinical practice and policy in low and middle-income countries. Disabil Rehabil. 2021:1-8.

50. Himmelmann K, Uvebrant P. The panorama of cerebral palsy in Sweden part XII shows that patterns changed in the birth years 2007-2010. Acta Paediatr. 2018;107(3):462-8.

51. Kakooza-Mwesige A, Forssberg H, Eliasson AC, Tumwine JK. Cerebral palsy in children in Kampala, Uganda: clinical subtypes, motor function and co-morbidities. BMC Res Notes. 2015;8:166.

52. Galea C, McIntyre S, Smithers-Sheedy H, Reid SM, Gibson C, Delacy M, et al. Cerebral palsy trends in Australia (1995-2009): a population-based observational study. Dev Med Child Neurol. 2019;61(2):186-93.

53. Imam MHA, Jahan I, Das MC, Muhit M, Smithers-Sheedy H, McIntyre S, et al. Rehabilitation status of children with cerebral palsy in Bangladesh: Findings from the Bangladesh Cerebral Palsy Register. PLOS ONE. 2021;16(5).

54. Karim T, Al Imam MH, Golland P, Khan AI, Hossain I, Smithers-Sheedy H, et al. Hip dysplasia among children with spastic cerebral palsy in rural Bangladesh. BMC Musculoskelet Disord. 2019;20(1):494.

55. Khan A, Ashher F, Karim T, Fatema A, Jahan I, Muhit M, et al. Immunization of Mothers of Children with Cerebral Palsy in Rural Bangladesh. Infect Disord Drug Targets. 2020;20(3):303-8.

56. Power R, Galea C, Muhit M, Heanoy E, Karim T, Badawi N, et al. What predicts the proxy-reported health-related quality of life of adolescents with cerebral palsy in Bangladesh? BMC Public Health. 2020;20(1):18.

57. Power R, Akhter R, Muhit M, Wadud S, Heanoy E, Karim T, et al. A quality of life questionnaire for adolescents with cerebral palsy: psychometric properties of the Bengali CPQoL-teens. Health Qual Life Outcomes. 2019;17(1):135.

58. Power R, King C, Muhit M, Heanoy E, Galea C, Jones C, et al. Health-related quality of life of children and adolescents with cerebral palsy in low- and middle-income countries: a systematic review. Dev Med Child Neurol. 2018;60(5):469-79.

59. Power R, Akhter R, Muhit M, Wadud S, Heanoy E, Karim T, et al. Cross-cultural validation of the Bengali version KIDSCREEN-27 quality of life questionnaire. BMC Pediatr. 2019;19(1):19.

60. Khandaker G VBN, Dũng TQ, Giang NT, Chau CM, Van Anh NT, Van Thuong N, Badawi N, Elliott EJ. Protocol for hospital based-surveillance of cerebral palsy (CP) in Hanoi using the Paediatric Active Enhanced Disease Surveillance mechanism (PAEDS-Vietnam): a study towards developing hospital-based disease surveillance in Vietnam. BMJ open. 2017 Nov;7(11):e017742.

61. Rosenbaum P PN, Leviton A, Goldstein M, Bax M, Damiano D, Dan B, Jacobsson B. A report: the definition and classification of cerebral palsy April 2006. Dev Med Child Neurol Suppl 2007 Feb;109:8-14.

62. Novak I, Hines M, Goldsmith S, Barclay R. Clinical prognostic messages from a systematic review on cerebral palsy. Pediatrics. 2012;130(5):e1285-312.

63. Robin J, Graham HK, Baker R, Selber P, Simpson P, Symons S, et al. A classification system for hip disease in cerebral palsy. Developmental Medicine & Child Neurology. 2009;51(3):183-92.

Baker F, Burnett H, Gibson N, Graham K, Kentish M, Love S, et al. Australian Hip Surveillance Guidelines for Children with Cerebral Palsy 2020. In: Group ANHSW, editor. 2020.
Vitrikas K, Dalton H, Breish D. Cerebral Palsy: An Overview. Am Fam Physician. 2020;101(4):213-20.

66. Hägglund G, Alriksson-Schmidt A, Lauge-Pedersen H, Rodby-Bousquet E, Wagner P, Westbom L. Prevention of dislocation of the hip in children with cerebral palsy: 20-year results of a population-based prevention programme. The bone & joint journal. 2014;96(11):1546-52.

67. Hashan MR, Smoll N, King C, Ockenden-Muldoon H, Walker J, Wattiaux A, et al. Epidemiology and clinical features of COVID-19 outbreaks in aged care facilities: A systematic review and meta-analysis. EClinicalMedicine. 2021:100771.

68. Wynter M, Gibson N, Willoughby KL, Love S, Kentish M, Thomason P, et al. Australian hip surveillance guidelines for children with cerebral palsy: 5-year review. Dev Med Child Neurol. 2015;57(9):808-20.

69. Boldyreva U, Streiner DL, Rosenbaum PL, Ronen GM. Quality of life in adolescents with epilepsy, cerebral palsy, and population norms. Dev Med Child Neurol. 2020;62(5):609-14.

70. Ahmed SM, Hossain MA, Rajachowdhury AM, Bhuiya AU. The health workforce crisis in Bangladesh: shortage, inappropriate skill-mix and inequitable distribution. Hum Resour Health. 2011;9(1):3.

71. Ahlin K, Jacobsson B, Nilsson S, Himmelmann K. Antecedents and neuroimaging patterns in cerebral palsy with epilepsy and cognitive impairment: a population-based study in children born at term. Acta obstetricia et gynecologica Scandinavica. 2017;96(7):828-36.

72. Hanes JE, Hlyva O, Rosenbaum P, Freeman M, Nguyen T, Palisano RJ, et al. Beyond stereotypes of cerebral palsy: Exploring the lived experiences of young Canadians. Child: care, health and development. 2019;45(5):613-22.

73. Tsubouchi Y, Tanabe A, Saito Y, Noma H, Maegaki Y. Long-term prognosis of epilepsy in patients with cerebral palsy. Developmental Medicine & Child Neurology. 2019;61(9):1067-73.

74. Aggarwal S, Chadha R, Pathak R. Nutritional status and growth in children with cerebral palsy: a review. International Journal of Medical Science and Public Health. 2015;4(6).

75. Jahan I, Karim T, Das MC, Muhit M, Mcintyre S, Smithers-Sheedy H, et al. Mortality in children with cerebral palsy in rural Bangladesh: a population-based surveillance study. Developmental Medicine & Child Neurology. 2019;61(11):1336-43.

76. Mlinda SJ, Leyna GH, Massawe A. The effect of a practical nutrition education programme on feeding skills of caregivers of children with cerebral palsy at Muhimbili National Hospital, in Tanzania. Child Care Health Dev. 2018;44(3):452-61.

77. Donkor CM, Lee J, Lelijveld N, Adams M, Baltussen MM, Nyante GG, et al. Improving nutritional status of children with Cerebral palsy: a qualitative study of caregiver experiences and community-based training in Ghana. Food Sci Nutr. 2019;7(1):35-43.

78. Woolfenden S, Goldfeld S, Raman S, Eapen V, Kemp L, Williams K. Inequity in child health: the importance of early childhood development. J Paediatr Child Health. 2013;49(9):E365-9.

79. Novak I, Morgan C, Adde L, Blackman J, Boyd RN, Brunstrom-Hernandez J, et al. Early, Accurate Diagnosis and Early Intervention in Cerebral Palsy: Advances in Diagnosis and Treatment. JAMA Pediatr. 2017;171(9):897-907.

80. te Velde A, Morgan C, Novak I, Tantsis E, Badawi N. Early Diagnosis and Classification of Cerebral Palsy: An Historical Perspective and Barriers to an Early Diagnosis. Journal of clinical medicine. 2019;8(10):1599.

81. Morgan C, Fetters L, Adde L, Badawi N, Bancale A, Boyd RN, et al. Early Intervention for Children Aged 0 to 2 Years With or at High Risk of Cerebral Palsy: International Clinical Practice Guideline Based on Systematic Reviews. JAMA pediatrics. 2021.

82. Kakooza-Mwesige A, Andrews C, Peterson S, Wabwire Mangen F, Eliasson AC, Forssberg H. Prevalence of cerebral palsy in Uganda: a population-based study. Lancet Glob Health. 2017;5(12):e1275-e82.

83. te Velde A, Morgan C, Novak I, Tantsis E, Badawi N. Early Diagnosis and Classification of Cerebral Palsy: An Historical Perspective and Barriers to an Early Diagnosis. J Clin Med. 2019;8(10):1599.

84. Shepherd E, Salam RA, Middleton P, Han S, Makrides M, McIntyre S, et al. Neonatal interventions for preventing cerebral palsy: an overview of Cochrane Systematic Reviews. Cochrane Database Syst Rev. 2018;6(6):CD012409.

85. Shepherd E, Salam RA, Middleton P, Makrides M, McIntyre S, Badawi N, et al. Antenatal and intrapartum interventions for preventing cerebral palsy: an overview of Cochrane systematic reviews. Cochrane Database of Systematic Reviews. 2017(8).

86. Novak I, Morgan C, Fahey M, Finch-Edmondson M, Galea C, Hines A, et al. State of the Evidence Traffic Lights 2019: Systematic Review of Interventions for Preventing and Treating Children with Cerebral Palsy. Curr Neurol Neurosci Rep. 2020;20(2):3.

87. Chabrier S, Pouyfaucon M, Chatelin A, Bleyenheuft Y, Fluss J, Gautheron V, et al. From congenial paralysis to post-early brain injury developmental condition: Where does cerebral palsy actually stand? Annals of physical and rehabilitation medicine. 2020;63(5):431-8.

Epidemiology of cerebral palsy in Bangladesh: a population-based surveillance study

GULAM KHANDAKER^{1,2,3,4} (D) | MOHAMMAD MUHIT^{2,3} | TASNEEM KARIM^{1,2,3} | HAYLEY SMITHERS-SHEEDY^{1,5} (D) | IONA NOVAK⁵ | CHERYL JONES^{1,6,7} | NADIA BADAWI^{1,5}

1 Discipline of Child and Adolescent Health, Sydney Medical School, University of Sydney, Sydney, New South Wales, Australia. 2 CSF Global, Dhaka; 3 Asian Institute of Disability and Development (AIDD), University of South Asia, Dhaka, Bangladesh. 4 Public Health Unit, Central Queensland Hospital and Health Service, Rockhampton, Queensland; 5 Cerebral Palsy Alliance Research Institute, University of Sydney, Sydney, New South Wales; 6 Marie Bashir Institute for Infectious Diseases and Biosecurity (MBI), University of Sydney, Sydney, New South Wales; 7 Department of Paediatrics, The University of Melbourne, Parkville, Victoria, Australia.

Correspondence to Gulam Khandaker at The Children's Hospital at Westmead (Clinical School), Cnr Hawkesbury Rd and Hainsworth St, Locked Bag 4001, Westmead, NSW 2145, Australia. E-mail: gulam.khandaker@health.nsw.gov.au

PUBLICATION DATA

Accepted for publication 13th July 2018. Published online

ABBREVIATIONS

BCPR	Bangladesh Cerebral Palsy
	Register
HIC	High-income country
LMIC	Low- and middle-income
	country
MACS	Manual Ability Classification
	System

AIM To examine the prevalence, clinical characteristics, and risk factors of cerebral palsy (CP) in children in Bangladesh.

METHOD The Bangladesh CP Register is an ongoing population-based surveillance database of children with CP from a geographically defined area in Bangladesh. Cases were defined based on Surveillance of CP in Europe and Australian CP Register criteria after clinical assessments and identification by the key informant's method.

RESULTS In total, 726 children with CP were identified between January 2015 and December 2016. Mean age was 7 years 7 months (standard deviation [SD] 4y 6mo; range: 4.8mo–18y; median 7y 1.2mo; 61.8% male, 38.2% female). Mean age at CP diagnosis was 5 years 2 months (SD 3.8). Observed prevalence was 3.4 per 1000 children (95% confidence interval [CI]: 3.2–3.7), resulting in an estimated 233 514 children (95% CI: 219 778–254 118) with CP in Bangladesh. The majority (79.6%) had spastic CP. Altogether, 79.6% of the children with CP had at least one associated impairment (speech 67.6%, intellectual 39.0%, epilepsy 23.7%, visual 10.2%, and hearing 10.2%). In total, 78.2% never received rehabilitation.

INTERPRETATION In Bangladesh, the burden of CP is high, and diagnosis is substantially delayed, limiting opportunities for early intervention. There is a lack of available services and the majority of the children had preventable risk factors.

Cerebral palsy (CP) comprises a heterogeneous group of early-onset, non-progressive, neurodevelopmental disorders caused by insult to the developing brain.¹ CP is a leading cause of disability among children, with a prevalence of approximately 2 per 1000 live births. Globally, there are an estimated 17 million people living with CP.² The epidemiology of CP has been studied extensively in high-income countries (HICs). However, in low- and middle-income countries (LMICs), there is paucity of evidence based on large, general population-representative samples, which can provide reliable, unbiased data.

A recent systematic review and meta-analysis of prevalence of CP included 49 studies, of which 46 were conducted in HICs.³ Out of the three studies from LMICs, only one (from China) used a population-based sample of children aged younger than 7 years and reported a prevalence of 1.6 per 1000 children.⁴ Most studies of children with CP in LMICs are based on high-risk groups from hospital or specialist clinic samples.⁵ A recent populationbased study of CP in Uganda, using a health and demographic surveillance system, showed a prevalence of CP of 2.7 (95% confidence interval [CI] 2.2–3.3) per 1000 children.⁶ General population-based studies are therefore needed to accurately quantify the prevalence of CP in LMICs, which is crucial for health service planning.

The clinical characteristics of and risk factors for CP in LMICs may differ from those in HICs.^{5,7,8} Preterm birth is a leading risk factor in HICs.⁹ However, a study based on rehabilitation clinic attendees suggested the proportion of children with CP in Bangladesh born preterm is much lower than that in Australia (18.5% vs 42.9%).¹⁰

Moreover, CP continues to be commonly attributed to a range of perinatal and birth-related factors, such as birth asphyxia, in LMICs.^{11,12} Additionally, the role of infections in the aetiology of CP and relevant preventive strategies are being increasingly explored, and are pertinent in low-resource settings where they continue to be more prevalent.¹³

In a recent study, signs of neonatal encephalopathy were more common among children with CP in Bangladesh than in Australia.¹⁰ Similarly, a hospital-based study from Nigeria reported that birth asphyxia, bilirubin encephalopathy, and postinfectious brain damage were the main causes of CP.¹⁴ As access to specialist care is limited for children with CP in LMICs, studies based on clinical samples may not be representative of all children with CP. There may be systematic differences between those who can and cannot afford rehabilitative services. Therefore, large population-based studies are needed to investigate the clinical characteristics and risk factors for children with CP in LMICs.

In the last 30 years, general population-based registers and surveillance programmes have facilitated comprehensive studies into the epidemiology of CP.^{15–18} The main aims of these initiatives are to monitor incidence and prevalence of CP, understand comorbidities and risk factors, design and evaluate preventative strategies, and monitor service delivery.^{15,17} To the best of our knowledge, there are currently 38 established CP registers/surveillance programmes worldwide, none of which are in LMICs.¹⁹

We examined the prevalence, clinical characteristics, associated impairments, risk factors, and service utilization in children with CP in rural Bangladesh through the first general population-based CP register in a LMIC, the Bangladesh Cerebral Palsy Register (BCPR).

METHOD

The BCPR is an ongoing population-based surveillance programme that includes children (recruited before 18y of age) with a confirmed clinical diagnosis of CP, living in a geographically defined area in a northern subdistrict of Bangladesh (i.e. Shahjadpur) with a known denominator population. There are 296 villages and an estimated 70 998 households with a total population of 561 076 (child population 226 114),²⁰ and 12 117 live births per annum.²¹ Baseline case identification for the register took place between January 2015 and December 2016, which forms the BCPR cohort.

Participants

Cerebral palsy was defined based on Surveillance of CP in Europe²² and Australian CP Register¹⁵ criteria. Children were identified by the community-based key informant's method^{23,24} and then assessed in a subsequent clinic. A detailed account of the BCPR study has been described previously.²⁰

Screening and identification of children with suspected CP

The key informant's method involves training of local volunteers, referred to as key informants, who reside in the community.²⁰ It is a novel method that was developed to identify children with blindness in LMICs.²³ Subsequently, the key informant's method was successfully used to identify children with other impairments in LMICs and was found to be sensitive and highly cost-effective.²⁴ The study implementation team (CSF Global; www.csf-global.org) has an extensive network of trained key informants in the surveillance area. For this study, two trained community

What this paper adds

- Prevalence of cerebral palsy (CP) is 3.4 per 1000 children in rural Bangladesh.
- There are an estimated 233 514 children with CP in Bangladesh.
- The majority have potentially preventable risk factors.
- Diagnosis of CP is delayed, limiting opportunities for early intervention.
 There is a lack of available services for children with CP in rural Bangla-
- desh.

mobilizers were engaged for identification of the key informants in the study area. The key informants received daylong structured training on the identification of children with suspected CP. After initial training, key informants identified and listed potential cases in their communities and informed the families of the date and venue of the medical assessment camps, which were conducted regularly in the study area. The children underwent clinical assessment for confirmation of the diagnosis of CP before registration into the BCPR.²⁰

Multidisciplinary medical assessment

A detailed neurological assessment was performed by a multidisciplinary medical assessment team comprised of a paediatrician, a physiotherapist, and a counsellor. The BCPR registration form, a modified version of the Australian CP Register record form (Appendix S1, online supporting information), was used to collect information on pregnancy, birth history, and sociodemographic factors. Gestational age was estimated based on last menstrual period, patient's history, and limited available medical records. Neurological types of CP, along with Gross Motor Function Classification System (GMFCS)²⁵ and Manual Ability Classification System (MACS)²⁶ levels, were assessed. MACS was determined for children over 4 years of age. Anthropometric measurements - weight (kg) and height (cm) - were also measured using standard measuring instruments and following standard guidelines. Z-scores for weight for age, height for age, and weight for height were calculated using the standard formula. Detailed methods and the nutritional status of our cohort has been reported previously.²⁷

The presence and severity of comorbidities (i.e. epilepsy, visual, hearing, speech, and intellectual impairments) was documented based on review of any previous medical records, report by the parents or primary caregivers of the children with CP, and clinical assessment. Parents or primary caregivers were asked specific questions regarding the presence of each of the associated impairments and subsequent clinical history was taken, and detailed assessment was conducted by clinicians to determine the type and severity of associated impairments.

Diagnosis of epilepsy was based on the history of one or more unprovoked seizure in the previous 3 months beyond the neonatal period. Assessment was based on clinical history taken by medical practitioner and review of any available medical records of relevance.

Assessment of intellectual impairment was based on history taken by the medical practitioner from the primary caregiver and review of any available medical records. Visual impairment was determined through history, available medical records, and clinical assessment of visual acuity and functional vision, including counting fingers, hand motion, and light perception. A monocular cover–uncover test was conducted for determination of strabismus.

Assessment of hearing impairment was based on hearing-related medical history, including child's response to name call, clap and vehicle horns, previous ear infections, symptoms of ear pain or drainage, family history of hearing loss, prescription and over-the-counter medication history, and previous use of hearing loss interventions. Otoscopic examination was performed to evaluate the pinna (outer ear), external auditory canal, and tympanic membrane for any conditions that could be contributing to hearing loss or that may require further evaluation and treatment, such as cerumen impaction and abnormality of the tympanic membrane.

Assessment of speech was based on history taken from primary caregiver and detailed speech and language assessment by medical practitioner. Expressive and respective language, naming quality, and quality of conversational speech were also observed.

Intrapartum-related neonatal respiratory depression was defined as a newborn failing to cry at the time of birth, experiencing delayed onset of breathing (>1min), or requiring assistance to initiate breathing (ranging from drying, stimulation, and milking of the umbilical cord to mouth-to-mouth breaths). Probable intrapartum-related neonatal respiratory depression was defined as neonatal respiratory depression among infants born at term without congenital malformations.²⁸

Early feeding difficulty was considered when a neonate was unable to breastfeed in the first month of life owing to poor sucking ability.

Neonatal encephalopathy was defined as a 'disturbance of neurological function in the earliest days of life in the term infant manifested by difficulty initiating and maintaining respiration, depression of tone or reflexes, abnormal level of consciousness and often by seizures'.²⁹ Possible neonatal encephalopathy was defined based on neurological abnormalities observed in the first 7 days of life (seizures or lethargy and early feeding difficulty).²⁸

Neonatal sepsis was defined as 'a clinical syndrome characterized by signs and symptoms of infection with or without accompanying bacteremia in the first month of life'.³⁰

Comparison between the BCPR cohort and the general population of the surveillance area was conducted using National Population and Housing Census, which collects data on demographic and health indicators nationally up to subdistrict level.^{21,31} The demographics and known maternal and perinatal risk factors for CP were also compared with local and national data for the general population.

Ethical considerations

Ethical approval was obtained from the Cerebral Palsy Alliance Human Research Ethics Committee (EC00402; ref no. 2015-03-02) in Australia; the Asian Institute of Disability and Development Human Research Ethics Committee (southasia-irb-2014-l-01); and the Bangladesh Medical Research Council Human Research Ethics Committee (BMRC/NREC/2013-2016/1267) in Bangladesh. Written informed consent was obtained from parents or primary caregivers of all the study participants.

Statistical analysis

Descriptive (mean, median, and proportion) and bivariate $(\chi^2$, Fisher's exact test, and independent-sample *t*-test) analyses were carried out. A *p* value of less than 0.05 was considered significant. Data were analysed using SPSS Statistics software version 23 (IBM, Armonk, NY, USA).

The observed prevalence of CP was estimated by dividing the total number of children with CP by all live births in the surveillance area between 1998 and 2016.

Prevalence of CP by gestational age was estimated only in children aged 5 years of age or older. Children aged younger than 5 years (i.e. born after 2011) were excluded from the analysis, as diagnosis of CP in young children is difficult and less precise than that of older cohorts,³² and could give an underestimate because of incomplete case ascertainment. The total number of live births between 1998 and 2011 in the surveillance area was derived from the Bangladesh Population and Housing Census data.²¹ The proportion of births across gestational age groups was used to determine the total number of live births across the gestational age groups.³¹ The prevalence of CP across each gestational age group was then determined by dividing the number of children living with CP by the total number of live births in respective gestational age group in the surveillance area between 1998 and 2011.

RESULTS

During the study period, 847 children with suspected CP were assessed, of whom 726 (85.7%) had a clinical diagnosis of CP and were recruited into the BCPR. Herein, we describe all cases of CP identified in the surveillance area.

The mean age of study participants was 7 years 7 months (standard deviation [SD] 4y 6mo; range: 4.8mo–18y; median 7y 1.2mo; interquartile range 3y 8.4mo); 61.8% (n=449) were male, 38.2% (n=277) were female. The mean age at CP diagnosis was 5 years 2 months (SD 3y 9.6mo); this varied significantly between semi-urban (4y 4mo [SD 3y 2.4mo]) and rural (5y 5mo [SD 4y]) areas (p<0.005). In total, 17.6% (n=128) children were diagnosed within the first 1000 days of life, of whom 95.3% (n=122) were diagnosed during the assessment camps as part of this study. Table I demonstrates the baseline characteristics, predominant CP subtypes, functional motor severity classifications, and associated impairments of the BCPR cohort.

The mean monthly income of the families was 103.9 US dollars (USD) (SD 86.2USD; median 72.0USD, interquartile range 72.0–120.0USD). In total, 97.2% of the families included in the CP register were living below the poverty line (i.e. daily income per capita <1.90USD),³³ which is

 Table I: Baseline characteristics, predominant cerebral palsy (CP) subtype, functional motor severity classifications, and associated impairments of the children with CP recruited into the Bangladesh Cerebral Palsy Register (n=726)

Characteristics	BCPR
Sex	
Male	449 (61.8)
Female	277 (38.2)
Age group (y)	
<5	248 (34.2)
5–9	263 (36.2)
10–14	155 (21.3)
15–18	60 (8.3)
Predominant CP motor type and subtype	
Spastic	578 (79.6)
Monoplegia/hemiplegia	198 (27.3)
Diplegia	124 (17.1)
Triplegia	70 (9.6)
Quadriplegia	186 (25.6)
Hypotonic	87 (12.0)
Dyskinetic	60 (8.3)
Ataxic	1 (0.1)
GMFCS level	
	98 (13.5)
	133 (18.3)
	157 (21.6)
IV	172 (23.7)
V MAGO I IR	166 (22.9)
MACS level	110 (01 5)
	116 (21.5)
II.	92 (17.0)
	97 (18.0)
	111 (20.6)
	124 (23.0)
Associated impairments	407 (07 1)
Speech	487 (07.1)
Intellectual Strahiomus	201 (30.7)
Subjectures	207 (28.5) 170 (22.4)
Epilepsy	74 (10.2)
Vieuel	74 (10.2) 72 (10.1)
visual	/3 (10.1)

Data are *n* (%). ^aManual Ability Classification System (MACS) was applicable for 540 children \geq 4y. ^bMultiple impairments were reported for several children. GMFCS, Gross Motor Function Classification System.

significantly higher than the national average (18.5%; p < 0.001).

Prevalence of CP

In the surveillance area the observed prevalence of CP was 3.4 per 1000 live births (95% CI: 3.2–3.7) from 1998 to 2016. Based on this prevalence we estimate that there are 233 514 children with CP in Bangladesh (95% CI: 219 778–254 118), based on 68 680 562 live births in Bangladesh from 1998 to 2016.

The observed prevalence of CP varied by sex and area of residence; prevalence was higher in males than females (χ^2 =29.4; *p*<0.001); 3.7 per 1000 males (95% CI: 3.4–4.1) and 2.5 per 1000 females (95% CI: 2.2–2.8). The observed prevalence was 4.7 per 1000 children in semi-urban (95% CI: 3.9–5.6), and 2.9 per 1000 children in rural (95% CI: 2.7–3.2) parts of the surveillance area (χ^2 =16.738; *p*<0.001).

Birth prevalence of CP by gestational age in children aged 5 years or older

The prevalence of CP varied by gestational age; the highest prevalence was observed to be 7.8 per 1000 live births (95% CI: 5.6–10.9) for those born very preterm (28–31wks' gestation). The prevalence of CP was 1.1 per 1000 live birth (95% CI: 0.5–2.4) and 1.6 per 1000 live births (95% CI: 1.2–2.2) for those born at 32 weeks to 33 weeks and at 34 weeks to 36 weeks respectively (Table II).

CP motor type and functional motor severity classifications

The majority of the children (79.6%; n=578) had a spastic motor type; 27.3% (n=198) had monoplegia/hemiplegia, 17.1% (n=124) had diplegia, 9.6% (n=70) had triplegia, and 25.6% (n=186) had quadriplegia. In total, 68.2% (n=495) were described as GMFCS level III to V; 61.5% (n=332) of the children were MACS level III to V (Table I).

Associated impairments

In total, 79.6% (n=578) of the children had at least one associated impairment. Altogether, 67.1% (n=487) had speech impairments, including 35.1% (n=253) who were non-verbal. Two hundred and eighty-one (38.7%) had an intellectual impairment, 28.5% (n=207) had strabismus, 23.4% (n=170) had epilepsy, 10.2% (n=74) had hearing impairment, and 10.1% (n=73) had visual impairment, including 3.9% (n=28) who were functionally blind. Children with CP had a significantly higher rate of associated impairments compared with the general population (p<0.001; Table III).

Risk factors

Prenatal and perinatal risk factors were reported for 61.6% (*n*=447) of children, including intrapartum-related neonatal respiratory depression, neonatal encephalopathy, and infections (e.g. neonatal sepsis, pneumonia, central nervous system infection, such as meningitis and encephalitis). Post-neonatal factors were reported for 6.1% (*n*=44), which included infection (e.g. pneumonia and sepsis), drowning, and trauma.

Mean maternal age at birth was 22 years and 2 months (SD 6y 6mo; median 20y 10.8mo; range 13–42y). In total, 40.1% (n=291) of mothers were illiterate versus the national average of 25.0% (p<0.001). Only 30.3% (n=220) of mothers received regular antenatal care. The majority (72.7%; n=528) of the children were delivered at home, and over two-thirds of the births (67.1%; n=487) were attended by traditional birth attendants.

The mean gestational age at birth of the children with CP was 38.2 weeks (SD 3.1 wks). Altogether, 16.3% (n=119) were born preterm (i.e. <37wks). Birthweight was known for only 19.4% of children (n=141; mean 2648.9g). In total, 61.8% children had probable intrapartum-related neonatal respiratory depression, 42.8% had early feeding

Table II: Prevalence of cerebral palsy (CP) by gestational age in children aged \geq 5y (n=478)

		Percentage of live births in a	Estimated number of live births in	Prevalence of CP per
	BCPR, <i>n</i> (%)	Bangladeshi cohort	Shahjadpur ^a (1998–2011) ^{21,81}	1000 live births (95% CI)
 Term (≥37wks)	402 (84.1)	80.6	134 672	3.0 (2.7–3.3)
Preterm (28–36wks)	76 (15.9)	19.4	32 415	2.3 (1.9–2.9)
Very preterm (28–31wks)	34 (7.1)	2.6	4344	7.8 (5.6–10.9)
Moderate preterm (32–33wks)	6 (1.3)	3.3	5514	1.1 (0.5–2.4)
Late preterm (34–36wks)	36 (7.5)	13.5	22 557	1.6 (1.2–2.2)

^aThe estimated number of live births in Shahjadpur has been calculated from the percentage of live births reported in a Bangladeshi cohort³¹ from the surveillance area of the Bangladesh Population and Housing Census.²¹ BCPR, Bangladesh Cerebral Palsy Register; CI, confidence interval.

Table III: Associated impairments in children with cerebral palsy in com- parison to national data						
Associated impairments ^a	BCPR, <i>n</i> (%)	Bangladesh national data (%)	р			
Epilepsy	170 (23.4)	1.2 ⁵²	<0.001			
Intellectual	281 (38.7)	0.11 ⁴⁷	< 0.001			
Visual	73 (10.1)	0.09 ⁴⁷	< 0.001			
Hearing	74 (10.2)	0.0547	< 0.001			
Speech	487 (67.1)	0.18 ⁴⁷	< 0.001			

^aMultiple impairments were reported for several children. BCPR, Bangladesh Cerebral Palsy Register.

difficulty, and 30.9% had possible neonatal encephalopathy. The known maternal and perinatal risk factors are given in Table IV along with a comparison of Bangladesh national data and local data percentages for the general population. Neonatal jaundice was reported in 4.1% (n=30) of the children. Altogether, 15.8% (n=115) of the cohort had a history of consanguinity and 8.3% (n=5) of the children with dyskinesia had history of consanguinity.

The families of the BCPR cohort had a slightly better housing, water, and sanitation status than the general population. However, the proportion of families living below the poverty line was significantly (five times) greater in the BCPR cohort than in the general population in the surveillance area. Moreover, antenatal care was relatively poor in mothers of children in the BCPR cohort compared with the general population; a greater proportion of mothers from the BCPR cohort did not receive any antenatal check-up and delivered their child at home. Furthermore, there were significantly higher levels of preterm births, probable intrapartum-related neonatal respiratory depression, and feeding difficulties among the BCPR cohort compared with the general population.

Access to rehabilitation

The majority (n=568; 78.2%) of the children never received any rehabilitation services. Of these families, 64.8% (n=368) were unaware of the need for rehabilitation, 14.3% (n=81) could not afford the required services, and 4.2% (n=24) could not access the services owing to lack of disability accessible transport.

Only one-fifth (21.8%) of the children had access to rehabilitation services at non-governmental organization centres and/or hospitals. The mean age at which these children first received rehabilitation services was 4 years and 2 months (SD 3y 2.4mo). Among them, 28.5% received physical therapy and of the 495 children in GMFCS level III to V who could have benefited from assistive mobility equipment, only 5.5% (n=27) had any assistive devices.

Those who received rehabilitation had a higher family income (113.6USD vs 100.0USD; p=0.447). Mean age at assessment and mean age at diagnosis of children with CP were significantly different between those who received rehabilitation and those who did not (6y 10.8mo vs 7y 9.6mo [p=0.03] and 4y 2.4mo vs 5y 6mo [p<0.001] respectively). Table SI, online supporting information, shows the differences in sociodemographic, perinatal, and clinical characteristics between the two groups.

Mortality

During the study period, 26 children with CP died (19.8 deaths per 1000 person-years of observation). Mean age at death was 7 years 5 months (SD 5y 2.4mo); 65.4% were male, 34.6% were female. Spasticity was the most common type of CP (n=14 [53.9%]), while the remaining children had dyskinetic (n=3 [11.5%]) or hypotonic (n=9 [34.6%]) CP. Most children who died had severe functional mobility limitations (GMFCS level III–V, n=22 [84.6%]). Almost two-thirds (n=16 [61.5%]) of these children never received any rehabilitation services.

Nutritional status of the deceased children was particularly poor. The mean z-scores for weight for age (calculated for 20/26 children aged 0–121mo), height for age, and weight for height (calculated for 14/26 children aged ≤ 60 mo) were -3.3 (SD 1.4), -4.6 (SD 2.9), and -1.4 (SD 3.4) respectively. In total, 75.0% (*n*=15) were severely underweight, 76.9% (*n*=20) were severely stunted, and 50.0% (*n*=7) had severe acute malnutrition (severe underweight: weight for age less than -3SD, severe stunting: height for age less than -3SD, severe wasting: weight for height less than -3SD). Furthermore, five of the deceased children had both severe acute malnutrition and severe chronic malnutrition.

Table IV:	Comparison of	demographics	and know	n maternal	and	perinatal	risk fa	actors fo	r cerebral	palsy w	ith loca	l and nationa	l data fo	r general	l popu-
lation															

Local data Area of residence $(n-726)$ Semi-urban $(n-726)$ Type of accommodation $(n-726)$ Temporary shelter (jhupri) $5(0.7)$ 2.1^{271} 0.000 Mud houses (kutcha house) 551 (81.4) 87.0^{271} 0.001 Semi-pucca house 150 (81.4) 87.0^{271} 0.001 Semi-pucca house 108 (14.9) 8.3^{271} 0.001 Semi-pucca house 108 (14.9) 8.3^{271} 0.001 Semi-pucca house 22 (3.0) 2.6^{271} 0.263 Source of drinking water $(n-726)$ Tap water $6(0.8)$ 1.0^{271} 0.411 Tube-well 720 (99.2) 95.4^{271} 0.000 Semitary latrine 265 (36.5) 51.5^{471} 0.000 Sanitary latrine 265 (36.5) 51.5^{471} 0.000 Not oblet facility 9 (12) 2.6^{271} 0.000 National data Families living below poverty line (daily income per capita <1.90USD) 706 (97.2) 18.50^{531} <0.001 Antenatal care $(n-726)$ 220 (30.3) 31.2^{293} 0.031 I regular (>2 visits) 122 (19.6) 29.4^{43} <0.001 No Regular (>2 visits) 122 (19.6) 29.4^{43} <0.001 I regular (>2 visits) 124 (19.6) 29.4^{43} <0.001 I regular (>2 visits) 124 (19.6) 29.4^{43} <0.001 Mode of delivery (n-726) 660 (90.9) 77.0^{53} <0.001 Mode of delivery (n-726) 660 (90.9) 77.0^{53} <0.001 Mode of delivery (n-726) 660 (90.9) 77.0^{53} <0.001 I regular (>2 visits) 124 (19.6) 29.4^{43} <0.001 Vaginal birth $(n-726)$ 660 (90.9) 77.0^{53} <0.001 Mode of delivery (n-726) 660 (90.9) 77.0^{53} <0.001 Mode of delivery (n-726) 660 (90.9) 77.0^{53} <0.001 Vaginal birth $(n-726)$ 660 (90.9) 77.0^{53} <0.001 Vaginal birth $(n-726)$ 677 (63.6^{51} <0.001 Mode rate pretern ($32-33$ 9 (112		BCPR, <i>n</i> (%)	Local/national data (%)	р
Area of residence (n=726) 911,5 ²¹ <0.001	Local data			
	Area of residence (<i>n</i> =726)			
Rural 605 (63.3) 88.5 ²¹ <0.001 Type of accommodation (n=726) 5 0.77 2.1 ²¹ 0.002 Mud houses (kutcha house) 591 (81.4) 87.0 ²⁷¹ <0.001	Semi-urban	121 (16.7)	11.5 ²¹	< 0.001
$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$	Rural	605 (83.3)	88.5 ²¹	< 0.001
$\begin{array}{cccc} \begin{tabular}{l lllllllllllllllllllllllllllllllllll$	Type of accommodation (<i>n</i> =726)			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Temporary shelter (jhupri)	5 (0.7)	2.1 ²¹	0.002
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Mud houses (kutcha house)	591 (81.4)	87.0 ²¹	< 0.001
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Semi-pucca house	108 (14.9)	8.3 ²¹	< 0.001
Source of drinking water (<i>n</i> =726) Tap water (<i>n</i> =726) Tap water (<i>n</i> =726) 0 3.6 ²¹ - 0 3.6 ²¹ - Sanitation (<i>n</i> =725) No toilet facility 9 (1.2) 2.6 ²¹ 0.009 Non-sanitary latrine 9 (1.2) 2.6 ²¹ 0.009 Non-sanitary latrine 9 (1.2) 2.6 ²¹ 0.009 Non-sanitary latrine 451 (62.1) 45.9 ²¹ -0.001 Sanitary latrine 451 (62.1) 45.9 ²¹ -0.001 National dat Family income Family income	Permanent brick (pucca) house	22 (3.0)	2.6 ²¹	0.263
Tap water 6 (0.8) 1.0^{21} 0.411 Tube-well 720 (99.2) 95.4 ²¹ <0.001	Source of drinking water (n=726)			
Tube-well720 (99.2)95.421<0.001Other sources (well, ponds, river)03.621-Sanitation (n =725)9 (1.2)2.6210.009Non-sanitary latrine265 (36.5)51.521<0.001	Tap water	6 (0.8)	1.0 ²¹	0.411
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Tube-well	720 (99.2)	95.4 ²¹	< 0.001
$\begin{array}{llllllllllllllllllllllllllllllllllll$	Other sources (well, ponds, river)	0	3.6 ²¹	_
No toilet facility 9 (1.2) 2.6^{21} 0.009 Non-sanitary latrine 265 (36.5) 51.5^{21} <0.001	Sanitation (n=725)			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	No toilet facility	9 (1.2)	2.6 ²¹	0.009
Sanitary latrine 451 (62.1) 45.9 ²¹ <0.001	Non-sanitary latrine	265 (36.5)	51.5 ²¹	< 0.001
National data Familis living below poverty line (daily income per capita <1.90USD)	Sanitary latrine	451 (62.1)	45.9 ²¹	< 0.001
Family income Family income proventy line (daily income per capita <1.90USD)706 (97.2) 18.50^{53} <0.001Antenatal care (n=726)262 (36.1) 21.4^{53} <0.001	National data			
Families living below poverty line (daily income per capita <1.90USD)706 (97.2) 18.50^{53} <0.001Antenatal care (n =726)262 (36.1) 21.4^{53} <0.001	Family income			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Families living below poverty line (daily income per capita <1.90USD)	706 (97.2)	18.50 ⁵³	<0.001
No262 (36.1)21.4 53<0.001Regular220 (30.3)31.2 530.316Irregular (≥ visits)142 (19.6)29.4 530.001Irregular (≥ visits)102 (14.0)17.9 530.003Place of birth (n=726)000Home528 (72.7)63.0 53<0.001	Antenatal care (n=726)			
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	No	262 (36.1)	21.4 ⁵³	<0.001
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Regular	220 (30.3)	31.2 ⁵³	0.316
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Irregular (>2 visits)	142 (19.6)	29 .4 ⁵³	<0.001
Place of birth (<i>n</i> =726) 63.0 ⁵³ <0.001	Irregular (≤ 2 visits)	102 (14.0)	17.9 ⁵³	0.003
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Place of birth (n=726)			
Health facility198 (27.3) 37.0^{53} <0.001Mode of delivery ($n=726$)66 (9.1) 23.0^{53} <0.001	Home	528 (72.7)	63.0 ⁵³	< 0.001
Mode of delivery ($n=726$)66 (9.1) 23.0^{53} <0.001Vaginal birth660 (90.9) 77.0^{53} <0.001	Health facility	198 (27.3)	37.0 ⁵³	< 0.001
Caesarean section $66 (9.1)$ 23.0^{53} <0.001 Vaginal birth $660 (90.9)$ 77.0^{53} <0.001 Delivery attended ($n=726$) $214 (29.5)$ 42.1^{53} <0.001 Traditional birth attendant $487 (67.1)$ 47.6^{53} <0.001 Other family member $25 (3.4)$ 6.3^{53} <0.001 Gestational age (wks) ($n=726$) $Term (\geq 37)$ $607 (83.6)$ 80.5^{31} 0.018 Very preterm ($28-31$) $59 (8.1)$ 2.6^{31} <0.001 Moderate preterm ($32-33$) $9 (1.2)$ 3.3^{31} <0.001 Late preterm ($34-36$) $51 (7.0)$ 13.5^{31} <0.001 Low birthweight (<2500 g) ^a ($n=141$) $53 (37.6)$ 36.7^{54} 0.445 Probable IPR NRD ($n=607$) ^b $380 (62.7)$ 22.0^{55} <0.001 Early feeding difficulty ($n=724$) $311 (42.9)$ 10.9^{55} <0.001 Maternal complications during child birth ($n=726$) $344 (47.4)$ $ -$ Maternal fever during pregnancy ($n=726$) $147 (20.2)$ $ -$	Mode of delivery (n=726)			
Vaginal birth $660 (90.9)$ 77.0^{53} <0.001 Delivery attended ($n=726$) $214 (29.5)$ 42.1^{53} <0.001 Traditional birth attendant $487 (67.1)$ 47.6^{53} <0.001 Other family member $25 (3.4)$ 6.5^{53} <0.001 Gestational age (wks) ($n=726$) $Term (\geq 37)$ $607 (83.6)$ 80.5^{31} 0.001 Moderate preterm ($28-31$) $59 (8.1)$ 2.6^{31} <0.001 Moderate preterm ($32-33$) $9 (1.2)$ 3.3^{31} <0.001 Late preterm ($34-36$) $51 (7.0)$ 13.5^{31} <0.001 Low birthweight (<2500 g) ^a ($n=141$) $53 (37.6)$ 36.7^{54} 0.445 Probable IPR NRD ($n=607$) ^b $380 (62.7)$ 22.0^{55} <0.001 Early feeding difficulty ($n=724$) $311 (42.9)$ 10.9^{55} <0.001 Maternal complications during child birth ($n=726$) $344 (47.4)$ $ -$ Maternal fever during pregnancy ($n=726$) $147 (20.2)$ $ -$	Caesarean section	66 (9.1)	23.0 ⁵³	<0.001
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Vaginal birth	660 (90.9)	77.0 ⁵³	<0.001
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Delivery attended (n=726)			
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	Doctor/midwife	214 (29.5)	42.1 ⁵³	<0.001
Other family member 25 (3.4) 6.3^{53} <0.001 Gestational age (wks) (n=726) 607 (83.6) 80.5^{31} 0.018 Very preterm (28–31) 59 (8.1) 2.6^{31} <0.001	Traditional birth attendant	487 (67.1)	47.6 ⁵³	<0.001
Gestational age (wks) (n=726) Term (≥37) 607 (83.6) 80.5^{31} 0.018 Very preterm (28–31) 59 (8.1) 2.6^{31} <0.001	Other family member	25 (3.4)	6.3 ⁵³	< 0.001
Term (≥37) 607 (83.6) 80.5^{31} 0.018 Very preterm (28–31) 59 (8.1) 2.6^{31} <0.001	Gestational age (wks) (n=726)			
Very preterm (28–31)59 (8.1) 2.6^{31} <0.001Moderate preterm (32–33)9 (1.2) 3.3^{31} <0.001	Term (>37)	607 (83.6)	80.5 ³¹	0.018
Moderate preterm (32–33) 9 (1.2) 3.3^{31} <0.001	Very preterm (28–31)	59 (8.1)	2.6 ³¹	< 0.001
Late preterm (34–36)51 (7.0) 13.5^{31} <0.001Low birthweight (<2500g) ^a (n=141)53 (37.6) 36.7^{54} 0.445 Probable IPR NRD (n=607) ^b 380 (62.7) 22.0^{55} <0.001	Moderate preterm (32–33)	9 (1.2)	3.3 ³¹	<0.001
Low birthweight (<2500g)a (n=141)53 (37.6) 36.7^{54} 0.445 Probable IPR NRD (n=607)b $380 (62.7)$ 22.0^{55} <0.001	Late preterm (34-36)	51 (7.0)	13.5 ³¹	< 0.001
Probable IPR NRD $(n=607)^{b}$ 380 (62.7) 22.0 ⁵⁵ <0.001	Low birthweight (<2500g) ^a (<i>n</i> =141)	53 (37.6)	36.7 ⁵⁴	0.445
Early feeding difficulty (n=724) 311 (42.9) 10.9 ⁵⁵ <0.001	Probable IPR NRD (n=607) ^b	380 (62.7)	22.0 ⁵⁵	< 0.001
Maternal complications during child birth (n=726)344 (47.4)Maternal fever during pregnancy (n=726)147 (20.2)	Early feeding difficulty (n=724)	311 (42.9)	10.9 ⁵⁵	< 0.001
Maternal fever during pregnancy (n=726) 147 (20.2) – –	Maternal complications during child birth (n=726)	344 (47.4)	_	_
	Maternal fever during pregnancy (n=726)	147 (20.2)	_	_

^aBirthweight unknown for 585 cases. ^bProbable IPR NRD was defined as neonatal respiratory depression among infants born at term (*n*=607) without congenital malformations. USD, US dollars; IPR NRD, intrapartum-related neonatal respiratory depression.

DISCUSSION

To the best of our knowledge, this is one of the first population-based surveillance studies of children with CP in a LMIC. We adopted an easily replicable and cost-effective model for collection of population-level data, which will serve as a tool for coordination of services for children with CP, and for future research on the epidemiology of CP in LMICs across the world.²⁰

Our findings suggest that the burden of CP is high in Bangladesh compared with HICs and other LMICs.^{6,13,15,22} However, our reported prevalence is an underestimate owing to delayed diagnosis and strong survival bias. In our study, the mean age at diagnosis of CP was 5 years and 2 months, which is considerably delayed compared with HICs (e.g. 1y 5mo in Australia).³⁴ Existing knowledge supports the importance of early diagnosis in ensuring referral to early intervention for optimization of infant motor and cognitive plasticity, prevention of secondary complications, and enhancement of caregivers' well-being.³⁴ Although systematic reviews of randomized controlled trials identified limited, heterogenous, and weak evidence on the long-term effect of early intervention,³⁵ several interventions appear promising in yielding a better outcome in children with CP.³⁶ Less than a fifth of our cohort were diagnosed within the first 1000 days of life;

among them, the majority were diagnosed during the assessment camps as part of this study. Therefore, it is evident that early intervention is largely limited by failure in early diagnosis. Furthermore, owing to delayed diagnosis, it is likely that we have underestimated the number of younger children (i.e. those aged <5y 2mo), resulting in an underestimate of prevalence.

Our findings reflect the high mortality rate in children with CP. We speculate that an additional number of children died before being identified as having CP, particularly considering the delayed diagnosis. Therefore, the observed prevalence is almost certainly affected by survival bias and should be considered an underestimate of the true prevalence in Bangladesh.

Preterm birth, a common cause of CP in HICs, is relatively underrepresented in our study. Moreover, the prevalence of CP was higher in children born at 34 weeks to 36 weeks' gestation, than in those born at 32 weeks to 33 weeks' gestation. This is possibly owing to the high mortality of infants born very preterm in low-resource settings, and there is a strong survival bias at play in lowresource settings such as the study site, where facilities are inadequate for neonatal resuscitation.37 However, we found that the prevalence of CP is nearly fourfold higher in children born very preterm. Although Bangladesh is among the top-10 countries with the greatest number of preterm births,³⁸ it has made considerable progress over the past decades.³⁹ However, successes in neonatal survival are counteracted by concerns of resultant increases in adverse neurodevelopmental outcomes, particularly CP.40

Of the known risk factors for CP, we found an alarmingly high proportion of probable intrapartum-related neonatal respiratory depression in children with CP. Many of those children had early feeding difficulty and possible neonatal encephalopathy. Neonatal jaundice was also reported in our cohort. Measures should be taken to address the prevalent risk factors that are potentially modifiable by quality perinatal care.

Associated impairments among children with CP is high and intellectual disability is a common accompanying impairment. We observed a significantly higher prevalence of intellectual impairment among our cohort compared to the national data. UNICEF reported that 'Estimates of the proportion of children with disabilities in Bangladesh are even more varied, ranging from less than 1.4 per cent to 17.5 per cent'.⁴¹ Hence, we speculate that the national data for the prevalence of intellectual impairment is possibly not representative of the true burden. Furthermore, a comparable rate of 45.0% was observed in another populationbased study conducted in Australia.⁴²

Compared with the Australian CP Register, our cohort had fewer children with hemiplegia and monoplegia (34.3% vs 39.1%) and more severe GMFCS classifications were observed (GMFCS levels III–V: 68.2% vs 40.4%).¹⁵ These reflect a greater need for rehabilitation services and assistive devices among these children.

Our study showed significant differences among children who had received rehabilitation services compared with those who had not. A significantly higher proportion of parents of children who had access to rehabilitation services were literate (49.1% vs 33.1%; p<0.001), and the diagnosis of CP in those children was established earlier (mean age at diagnosis 5y 6mo vs 4y 2.4mo; p<0.001) but was still late compared with developed countries. This suggests that reports from studies recruiting children from rehabilitation centres or hospitals are not representative of the general population, particularly in LMICs, where most children with CP are deprived of basic health care and rehabilitation services. Owing to the unique method of case ascertainment and population-level surveillance, we believe our study reflects the true picture of CP.

Children with CP and their families are disadvantaged compared with their counterparts in the general population in Bangladesh with significant disparities in basic demographic, antenatal, and perinatal indicators. Disability and poverty perpetuate one another.43 The majority of our cohort were living below the poverty line. Low socio-economic status was reported to be associated with increased risk of CP in several countries.44 A population-based study in the USA reported that black infants were 29% more likely to have CP than white infants.⁴⁵ The increased risk of CP was found to be primarily related to their higher risk of low birthweight and prematurity, owing largely to suboptimal prenatal care as a consequence of socio-economic factors. Socio-economic deprivation is a risk factor for more severe functional limitations in children with CP.46 Disability can. in turn, lead to poorer living standards through limited access to education, employment, and health services.⁴⁷

In Bangladesh, special education/schools are limited, with an alarmingly low number of seats (~2000 seats) available for disability inclusive education.⁴² Less than half (40.3%) of disabled children receive any education versus 69.7% of typically developing children.⁴⁷ Primary and secondary education providers in the study area do not have the facility to support children with special needs and none of them are equipped with trained teachers. Moreover, special education training of teachers is largely lacking in Bangladesh.

Findings from a study conducted in rural settings in a LMIC (i.e. Pakistan) highlight the cardinal role of preventive and promotive maternal and newborn interventions through community health workers.⁴⁸ Hence, ensuring equitable access to quality perinatal care through a combination of facility-based and community-based approaches can be a key preventive strategy in countries such as Bangladesh.

Limitations

One of the major challenges in establishing a populationbased register is completeness of case ascertainment. The key informant's method has a 77.6% case ascertainment rate compared with a door-to-door survey for identifying children with physical impairments in LMICs when it is conventionally conducted over a short period.^{23,24} It is likely that we have accounted for most of the children with CP alive during the collection of baseline information for the BCPR owing to our ongoing engagement in the surveillance area. However, as diagnosis of CP is delayed, it is likely that complete ascertainment has not been possible in young children (i.e. aged <5y). Furthermore, owing to a high mortality rate in children with CP in LMICs,⁴⁸ many children born between 1998 and 2014, especially those with more severe CP, would have died long before the formation of the registry. Hence, our reported prevalence is likely to be an underestimate of the actual burden.

We followed a strict clinical definition of CP adopted from established register networks in Europe and Australia.^{15,21} In our setting (i.e. remote and rural) clinical records were limited mainly to hospital discharge summaries, doctors' consultation reports, prescriptions, and immunization cards. Therefore, for some information (including risk factor data), we had to rely predominantly on parental interview. Moreover, estimation of gestational age based on last menstrual period, patient's history, and limited available medical records were the only possible option. Although there is existing evidence that mothers can accurately report the majority of prenatal and perinatal events,⁴⁹ and that the last menstrual period provides an accurate estimate of gestational age,^{50,51} this information could have been affected by recall bias.

The establishment of a precise diagnosis/description of CP was challenging, particularly for children aged younger than 5 years, as the clinical presentation and CP subtype can change over time. There is also the possibility that some children with mild CP may not have been identified for inclusion.

CONCLUSION

In Bangladesh, a typical LMIC, the burden of CP is high and diagnosis is frequently delayed, resulting in limited opportunities for early intervention during a time of high brain plasticity. Additionally, rehabilitation services are limited and not easily accessible. There are numerous potentially modifiable risk factors for CP in Bangladesh. Therefore, equitable access to quality perinatal care should be prioritized for effective prevention.

ACKNOWLEDGEMENTS

We would like to acknowledge the CSF Global team in Bangladesh for their support in both implementing this project and in supporting the families of children with cerebral palsy in their referrals and access to services. We would like to thank Mr Paul Novak from Compots Pty Ltd for developing the Bangladesh Cerebral Palsy Register online data repository. The authors have no interests which might be perceived as posing a conflict or bias. This study was supported by the Cerebral Palsy Alliance Research Institute (PG4314 – Bangladesh CP Register), Australia, and CSF Global, Bangladesh. GK is supported by the Cerebral Palsy Alliance Research Foundation Career Development Fellowship (CDF 0116). The study sponsor played no role in study design; in the collection, analysis, and interpretation of data; in the writing of the report; and in the decision to submit the paper for publication.

SUPPORTING INFORMATION

The following additional material may be found online:

Appendix S1: Bangladesh Cerebral Palsy Register registration form.

Table SI: Comparison of children with cerebral palsy who received rehabilitative services with those who never received any rehabilitation

REFERENCES

- Rosenbaum P, Paneth N, Leviton A, et al. A report: the definition and classification of cerebral palsy April 2006. Dev Med Child Neurol Suppl 2007; 109: 8–14.
- Graham HK, Rosenbaum P, Paneth N, et al. Cerebral palsy. Nat Rev Dis Primers 2016; 2: 15082.
- Oskoui M, Coutinho F, Dykeman J, Jetté N, Pringsheim T. An update on the prevalence of cerebral palsy: a systematic review and meta-analysis. *Dev Med Child Neurol* 2013; 55: 509–19.
- Liu JM, Li S, Lin Q, Li Z. Prevalence of cerebral palsy in China. Int *J Epidemiol* 1999; 28: 949–54.
- Singhi P, Saini AG. Changes in the clinical spectrum of cerebral palsy over two decades in North India – an analysis of 1212 cases. *J Trop Pediatr* 2013; 59: 434–40.
- Kakooza-Mwesige A, Andrews C, Peterson S, Wabwire Mangen F, Eliasson AC, Forssberg H. Prevalence of cerebral palsy in Uganda: a population-based study. *Lancet Glob Health* 2017; 5: e1275–82.
- Lawson RD, Badawi N. Etiology of cerebral palsy. *Hand* Clin 2003; 19: 547–56.
- Odding E, Roebroeck ME, Stam HJ. The epidemiology of cerebral palsy: incidence, impairments and risk factors. *Disabil Rebabil* 2006; 28: 183–91.

- Korzeniewski SJ, Slaughter J, Lenski M, Haak P, Paneth N. The complex aetiology of cerebral palsy. *Nat Rev Neural* 2018; 14: 528–43.
- 10. Benfer KA, Jordan R, Bandaranayake S, Finn C, Ware RS, Boyd RN. Motor severity in children with cerebral palsy studied in a high-resource and low-resource country. *Pediatrics* 2014; 134: e1594–602.
- Donald KA, Samia P, Kakooza-Mwesige A, Bearden D. Pediatric cerebral palsy in Africa: a systematic review. *Semin Pediatr Neurol* 2014; 21: 30–5.
- Gladstone M. A review of the incidence and prevalence, types and aetiology of childhood cerebral palsy in resource-poor settings. *Ann Trop Paediatr* 2010; 30: 181–96.
- Dan B, Paneth N. Making sense of cerebral palsy prevalence in low-income countries. *Lancet Glob Health* 2017; 5: e1174–5.
- Lagunju IA, Fatunde OJ. The child with cerebral palsy in a developing country – diagnosis and beyond. *J Pedi*atr Neurol 2009; 7: 375–9.
- Australian Cerebral Palsy Register Report, Birth Years 1993–2009. https://www.cpregister.com/pubs/pdf/ ACPR-Report_Web_2016.pdf (accessed 1 July 2017).

- Stanley FJ, Blair E, Alberman E. Cerebral Palsies: Epidemiology and Causal Pathways. Cambridge: Cambridge University Press, 2000.
- Hurley DS, Sukal-Moulton T, Msall ME, Gaebler-Spira D, Krosschell KJ, Dewald JP. The Cerebral Palsy Research Registry: development and progress toward national collaboration in the United States. *J Child Neu*rol 2011; 26: 1534–41.
- Blair E, Watson L. Epidemiology of cerebral palsy. Semin Fetal Neonatal Med 2006; 11: 117–25.
- Goldsmith S, McIntyre S, Smithers-Sheedy H, et al. An international survey of cerebral palsy registers and surveillance systems. *Dev Med Child Neurol* 2016; 58 (Suppl. 2): 11–17.
- 20. Khandaker G, Smithers-Sheedy H, Islam J, et al. Bangladesh Cerebral Palsy Register (BCPR): a pilot study to develop a national cerebral palsy (CP) register with surveillance of children for CP. *BMC Neurol* 2015; 15: 173.
- Bangladesh Bureau of Statistics. Bangladesh Population and Housing Census 2011. http://www.bbs.gov.bd/ site/page/47856ad0-7e1c-4aab-bd78-892733bc06eb/ Population-&-Housing (accessed 3 July 2017).

- 22. Surveillance of Cerebral Palsy in Europe. Surveillance of Cerebral Palsy in Europe: a collaboration of cerebral palsy surveys and registers. Surveillance of Cerebral Palsy in Europe (SCPE). Dev Med Child Neurol 2000; 42: 816–24.
- Muhit MA, Shah SP, Gilbert CE, Hartley SD, Foster A. The key informant method: a novel means of ascertaining blind children in Bangladesh. *Br J Ophtbalmol* 2007; 91: 995–9.
- 24. Murthy GV, Mactaggart I, Mohammad M, et al. Assessing the prevalence of sensory and motor impairments in childhood in Bangladesh using key informants. *Arch Dis Childhood* 2014; **99**: 1103–8.
- Rosenbaum PL, Palisano RJ, Bartlett DJ, Galuppi BE, Russell DJ. Development of the gross motor function classification system for cerebral palsy. *Dev Med Child Neurol* 2008; 50: 249–53.
- 26. Eliasson A-C, Krumlinde-Sundholm L, Rösblad B, et al. The Manual Ability Classification System (MACS) for children with cerebral palsy: scale development and evidence of validity and reliability. *Dev Med Child Neurol* 2006; 48: 549–54.
- 27. Jahan I, Muhit M, Karim T, et al. What makes children with cerebral palsy vulnerable to malnutrition? Findings from the Bangladesh cerebral palsy register (BCPR). *Disabil Rehabil* 2018; 13: 1–8.
- Lee AC, Mullany LC, Tielsch JM, et al. Incidence of and risk factors for neonatal respiratory depression and encephalopathy in rural Sarlahi. *Nepal. Pediatrics* 2011; 128: e915–24.
- Leviton A, Nelson KB. Problems with definitions and classifications of newborn encephalopathy. *Pediatr Neurol* 1992; 8: 85–90.
- 30. Aggarwal R, Sarkar N, Deorari AK, Paul VK. AIIMS protocols in Neonatology 2014: Sepsis in the Newborn. http://www.newbornwhocc.org/2014_pdf/Neonatal% 20sepsis%202014.pdf (accessed 15 May 2018).
- Baqui AH, Rosen HE, Lee AC, et al. Preterm birth and neonatal mortality in a rural Bangladeshi cohort: implications for health programs. *J Perimatol* 2013; 33: 977–81.
- McIntyre S, Morgan C, Walker K, Novak I. Cerebral palsy – don't delay. Dev Disabil Res Rev 2011; 17: 114– 29.

- The World Bank. Poverty & Equity. Bangladesh 2015. http://povertydata.worldbank.org/poverty/country/BGD (accessed 4 June 2017).
- 34. Novak I, Morgan C, Adde L, et al. Early, accurate diagnosis and early intervention in cerebral palsy: advances in diagnosis and treatment. *JAMA Pediatr* 2017; 171: 897–907.
- 35. Morgan C, Darrah J, Gordon AM, et al. Effectiveness of motor interventions in infants with cerebral palsy: a systematic review. *Dev Med Child Neurol* 2016; 58: 900–9.
- 36. Herskind A, Greisen G, Nielsen JB. Early identification and intervention in cerebral palsy. *Dev Med Child Neurol* 2015; 57: 29–36.
- 37. Shah R, Mullany LC, Darmstadt GL, et al. Incidence and risk factors of preterm birth in a rural Bangladeshi cohort. *BMC Pediatr* 2014; 14: 112.
- 38. Blencowe H, Cousens S, Oestergaard MZ, et al. National, regional, and worldwide estimates of preterm birth rates in the year 2010 with time trends since 1990 for selected countries: a systematic analysis and implications. *Lancet* 2012; 379: 2162–72.
- 39. UNICEF. World Prematurity Day: Saving "Every Newborn" born too soon. https://blogs.unicef.org/blog/ world-prematurity-day-saving-every-newborn-born-toosoon/ (accessed 14 May 2017).
- 40. Wilson-Costello D, Friedman H, Minich N, Fanaroff AA, Hack M. Improved survival rates with increased neurodevelopmental disability for extremely low birth weight infants in the 1990s. *Pediatrics* 2005; 115: 997–1003.
- 41. UNICEF Bangladesh. Situation Analysis on Children with Disabilities in Bangladesh. Dhaka, Bangladesh. UNICEF Bangladesh. June 2014. https://www.unicef.org/banglade sh/knowledgecentre_9023.html (accessed 4 June 2017).
- Reid SM, Meehan EM, Arnup SJ, Reddihough DS. Intellectual disability in cerebral palsy: a population-based retrospective study. *Dev Med Child Neurol* 2018; 60: 687–94.
- 43. Khandaker G, Muhit M, Rashid H, et al. Infectious causes of childhood disability: results from a pilot study in rural Bangladesh. *J Trop Pediatr* 2014; 60: 363–9.
- 44. McIntyre S, Taitz D, Keogh J, Goldsmith S, Badawi N, Blair EV. A systematic review of risk factors for cerebral palsy in children born at term in developed countries. *Dev Med Child Neurol* 2013; 55: 499–508.

- 45. Wu YW, Xing G, Fuentes-Afflick E, Danielson B, Smith LH, Gilbert WM. Racial, ethnic, and socioeconomic disparities in the prevalence of cerebral palsy. *Pediatrics* 2011; 127: e674–81.
- 46. Durkin MS. Socio-economic disparities and functional limitations of children with cerebral palsy. *Dev Med Child Neurol* 2016; 58: 115.
- Bangladesh Bureau of Statistics. Disability in Bangladesh: prevalence and pattern. http://203.112.218.65: 8008/WebTestApplication/userfiles/Image/PopMonogra phs/disabilityFinal.pdf (accessed 4 June 2017).
- Khan NZ, Ferdous S, Munir S, Huq S, McConachie H. Mortality of urban and rural young children with cerebral palsy in Bangladesh. *Dev Med Child Neurol* 1998; 40: 749–53.
- 49. Bhutta ZA, Soofi S, Cousens S, et al. Improvement of perinatal and newborn care in rural Pakistan through community-based strategies: a cluster-randomised effectiveness trial. *Lancet* 2011; 377: 403–12.
- 50. Deputy NP, Nguyen PH, Pham H, et al. Validity of gestational age estimates by last menstrual period and neonatal examination compared to ultrasound in Vietnam. BMC Pregnancy Childbirth 2017; 17: 25.
- Rosenberg RE, Ahmed AS, Ahmed S, et al. Determining gestational age in a low-resource setting: validity of last menstrual period. *J Health Popul Nutr* 2009; 27: 332–8.
- Epilepsy in the WHO South-East Asian Region Bridging the Gap. http://www.who.int/mental_health/neurol ogy/epilepsy/searo_report.pdf (accessed 4 June 2017).
- 53. National Institute of Population Research and Training (NIPORT), Mitra and Associates, and ICF International. Bangladesh demographic and health survey 2014. Dhaka, Bangladesh: NIPORT, Mitra and Associates and Rockville, MD: ICF International, 2016.
- 54. Bangladesh Bureau of Statistics, UNICEF. National Low Birth Weight Survey of Bangladesh, 2003–2004. BBS Dhaka Sep 2005. ISBN: 984-508-602-0. https:// www.unicef.org/bangladesh/Low_Birth_Weight_report. pdf (accessed 4 June 2017).
- 55. Parvez A, Rahman MM, Billah M, Akter A. Demographic approach to the empirical study of maternal and child health profile and policy in Bangladesh. *J Environ Sci Nat Res* 2014; 7: 197–206.

RESUMEN

EPIDEMIOLOGÍA DE LA PARÁLISIS CEREBRAL EN BANGLADESH: UN ESTUDIO DE VIGILANCIA BASADO EN LA POBLACIÓN

OBJETIVO Examinar la prevalencia, las características clínicas y los factores de riesgo de la parálisis cerebral (PC) en niños en Bangladesh.

MÉTODO El Registro de PC (BCPR siglas en ingles) de Bangladesh es una base de datos de vigilancia basada en la población de niños con PC, de un área geográficamente definida, en Bangladesh. Los casos se definieron siguiendo los criterios de vigilancia de PC en Europa y los criterios del registro de PC australiano, teniendo en cuenta las evaluaciones clínicas y la identificación por el método del informante clave.

RESULTADOS En total, se identificaron 726 niños con PC entre enero del 2015 y diciembre del 2016. La edad media fue de 7 años y 7 meses (desviación estándar [DE] 4ª 6m; rango: 4,8m-18a; mediana 7a 1,2m; 61,8% masculino, 38,2 % femenino). La edad promedio en el diagnóstico de PC fue de 5 años y 2 meses (DE 3.8). La prevalencia observada fue de 3,4 por cada 1.000 niños (intervalo de confianza [IC] del 95%: 3,2-3,7), resultando en un número estimado de 233.514 niños con PC (IC del 95% 219 778-254 118) en Bangladesh. La mayoría (79.6%) tenía PC espástica. En total, el 79.6% de los niños con PC tenían al menos un impedimento asociado (habla 67.6%, intelectual 39.0%, epilepsia 23.7%, visual 10.2% y audición 10.2%). En total, el 78.2% nunca recibió rehabilitación.

INTERPRETACIÓN En Bangladesh, el impacto de PC es alto y el diagnóstico se retrasa sustancialmente, lo que limita las oportunidades de intervención temprana. Hay una falta de servicios disponibles y la mayoría de los niños tenían factores de riesgo prevenibles.

RESUMO

EPIDEMIOLOGIA DA PARALISIA CEREBRAL EM BANGLADESH: UM ESTUDO DE VIGILÂNCIA COM BASE POPULACIONAL

OBJETIVO Examinar a prevalência, características clínicas, e fatores de risco para paralisia cerebral (PC) em crianças de Bangladesh.

MÉTODO O Registro de PC de Bangladesh (RPCB) é uma base de dados populacional em andamento para vigilância de crianças com PC de uma área geograficamente definida de Bangladesh. Os casos foram definidos com base nos critérios do Vigilância da PC na Europa e do Registro Australiano de PC após avaliação clínica e identificação pelo método pelo informante principal

RESULTADOS No total, 726 crianças com PC foram identificadas entre Janeiro 2015 e Dezembro 2016. A média de idade foi 7 anos e 7 meses (desvio padrão [DP] 4a 6m; variação: 4,8m–18a; mediana 7a 1,2m; 61,8% do sexo masculino, 38,2% do sexo feminino). A média de idade no momento do diagnóstic de PC foi de 5 anos e 2 meses (DP 3,8). A prevalência observada foi 3,4 a cada 1000 crianças (intervalo de confiança [IC] a 95% 3,2–3,7), resultando em uma estimativa de 233.514 crianças (IC 95% 219.778–254.118) com PC em Bangladesh. A maioria (79,6%) tinha PC espástica. No total, 79,6% das crianças com PC tinha pelo menos uma deficiência associada (linguagem 67,6%, intelectual 39,0%, epilepsia 23,7%, visual 10,2%, e auditiva 10,2%). No total, 78,2% nunca recebeu reabilitação.

INTERPRETAÇAO Em Bangladesh, o peso da PC é alto, e o diagnóstico é substancialmente tardio, limitando as oportunidades de intervenção precoce. Há carência de serviços disponíveis, e a maioria das crianças apresentou fatores de risco preveníveis.

APPENDIX B: BENGALI TRANSLATION OF THE EPILEPSY MANAGEMENT GUIDELINE

<u>খিঁচুনির চিকিৎসায় ঔষধ ব্যবহারের নির্দেশিকা</u>

<u>সূচীপত্র</u> খিঁচুনি ব্যবস্থাপনার ঔষধ সমূহঃ

खेयथ	পৃষ্ঠা
ফেনোবারবিটোন	৩
 ঔষধের প্রয়োগ মাত্রা 	
 যে ধরনের খিঁচুনির জন্যে উপকারী 	
 পার্শ্ব প্রতিক্রিয়া 	
 মন্তব্য 	
সোডিয়াম ভ্যালপ্রোয়েট	8
 ঔষধের প্রয়োগ মাত্রা 	
 যে ধরনের খিঁচুনির জন্যে উপকারী 	
 পার্শ্ব প্রতিক্রিয়া 	
 মন্তব্য 	
ক্লোবাজাম	¢
 ঔষধের প্রয়োগ মাত্রা 	
 যে ধরনের খিঁচুনির জন্যে উপকারী 	
 পার্শ্ব প্রতিক্রিয়া 	
 মন্তব্য 	
নাইট্রাজেপাম	৬
 ঔষধের প্রয়োগ মাত্রা 	
 যে ধরনের খিঁচুনির জন্যে উপকারী 	
 পার্শ্ব প্রতিক্রিয়া 	
 মন্তব্য 	
কার্বামাজেপিন	٩
 ঔষধের প্রয়োগ মাত্রা 	
 যে ধরনের খিঁচুনির জন্যে উপকারী 	
 পার্শ্ব প্রতিক্রিয়া 	
• মন্তব্য	

<u>ডিস্টোনিয়া/স্পাস্টিসিটি ব্যবস্থাপনার ঔষধ সমূহঃ</u>

ট্রাইহেক্সিফেনিডিল/বেনজহেক্সোল • ঔষধের প্রয়োগ মাত্রা • যে ধরনের লক্ষনের ক্ষেত্রে উপকারী	ዮ
 পার্শ প্রতিক্রিয়া 	
• মন্তব্য	
বেক্রোফেন	8
• ঔষধের প্রয়োগ মাত্রা	
 যে ধরনের লক্ষনের ক্ষেত্রে উপকারী 	
 পার্শ্ব প্রতিক্রিয়া 	
• মন্তব্য	
লেভাডোপা/কার্বিডোপা	20
• ঔষধের মাত্রা	
 যে ধরনের লক্ষনের ক্ষেত্রে উপকারী 	
 পার্শ্ব প্রতিক্রিয়া 	
 মন্তব্য 	
শিশুর অস্বাভাবিক আচরন ব্যবস্থাপনার ঔষধ সমূহঃ	
(Allelion	22
• ওধবের প্রয়োগ মাত্রা সংগ্রহনর উপস্থি দেশা পেরের শুরু করের	
 বে বরনের ভপ্রপা পেখা গেলে ওরু করতে 	
ere alter	
 ମାସ ଆତାଙ୍କରା ସେମ୍ବର 	
• শগুব্য কিন্দুর প্রিদান	
ার্বদোর্ভন - এম্বার্ধর প্রস্থার মার্চা	22
 ওববের এরে। প নাও। সে প্রবনের উপস্থের্ছ রবেরার উপস্থোগী 	
 বে বরণের ওপগণে ব্যবহার ওপবে। প। পোর্ব প্রতিক্রিয়া 	
 শার বাতাজন্য। সমরদ 	
 নতব্য <li< td=""><td></td></li<>	
<u> এরখরা অবহার রোশা ব্যাবহাশনার ওবব সমূহঃ</u>	
ডায়াজিপাম	১৩
 ঔষধের প্রয়োগ মাত্রা 	
 যে ধরনের উপসর্গে ব্যবহার উপযোগী 	
 পার্শ্ব প্রতিক্রিয়া 	
• মন্তব্য	
তথ্যসূত্র	১৩

<u>খিঁচুনি ব্যবস্থাপনার ঔষধ সমূহঃ</u>

ফেনোবারবিটোন

- ঔষধের মাত্রা
 - ১ থেকে ৬ মিলিগ্রাম/কেজি হিসেবে প্রতিদিন এক বা দুইবার দেয়া যাবে; ঔষধের সর্বনিম্ন মাত্রা থেকে ব্যবহার শুরু করে প্রয়োজনে ঔষধের মাত্রা ধীরে ধীরে বাড়াতে হবে।
- যে ধরনের খিঁচুনির জন্যে উপকারী
 - সারা শরীর জুড়ে খিঁচুনি এবং শরীরের একাংশের খিঁচুনির জন্যে দেয়া হয়।
 - বারবার শরীরে জ্বরের সাথে সাথে খিঁচুনি হলে দেয়া হয়।
 - জন্মের পর প্রথম ২৮ দিনে বা নবজাতক বয়সে খিঁচুনি হলে (Neonatal Seizure)
 - "অ্যাবসেন্স" খিঁচুনি (Absence Seizure) বা ফাঁকা দৃষ্টিতে চোখ খোলা রেখে যে খিঁচুনি হয় তাতে এই ঔষধ প্রয়োগে কোন উপকার পাওয়া যায়না।
- পার্শ্ব প্রতিক্রিয়া
 - অ্যালার্জি জনিত র্যাশ হতে পারে। হাইপারসেনসিটিভিটি রিঅ্যাকশন হতে পারে, যদিও তা এখন পর্যন্ত খুব কম সংখ্যক রোগীর ক্ষেত্রে দেখা গিয়েছে।
 - অতিক্রিয়াশীল বা শিশু অতি চঞ্চল হতে পারে।
 - তন্দ্রাচ্ছন্নতা বা ঘুম ঘুম ভাব।
- মন্তব্য
 - সাধারনত কোন ভালো ঔষধ নিয়মিত রোগীকে গ্রহন করতে হয় যদি না ঔষধ রোগীর কোন পার্শ্ব প্রতিক্রিয়া না হয় বা রোগ সারিয়ে না তোলে।
 - যদি রোগী এই ঔষধ তিন মাসের বেশি ব্যবহার করে থাকে সেক্ষেত্রে হঠাত করে ঔষধ বন্ধ করে দেয়া যাবেনা। কয়েক সপ্তাহ ধরে ধীরে ধীরে মাত্রা কমিয়ে একসময় ঔষধ পুরোপুরি বন্ধ করতে হবে।

সোডিয়াম ভ্যালপ্রোয়েট

- ঔষধের মাত্রা
 - রোগীর শরীরের প্রতি কেজি ভরে ২০ থেকে ৪০ মিলিগ্রাম হারে হিসেব করে প্রতিদিন দুইভাগে ভাগ করে ঔষধ দিতে হবে। শুরুতে শরীরের প্রতি কেজি ভরে ৫ থেকে ১০ মিলিগ্রাম/দিন হারে শুরু করে প্রতি কেজি ভরে ধীরে ধীরে ২০ মিলিগ্রাম পর্যন্ত ঔষধ বাড়াতে হয়। এরপর প্রয়োজনে মাত্রা বাড়ানো যায়।
- যে ধরনের খিঁচুনির জন্যে উপকারী
 - সকল ধরনের খিঁচুনির জন্যে প্রযোজ্য।
- পার্শ্ব প্রতিক্রিয়া
 - অতিক্রিয়াশীল বা শিশু অতি চঞ্চল হতে পারে।
 - যকৃত বা লিভার এর কার্যক্ষমতা কমে যেতে পারে।
 - দুই বছর বয়সের কম বা বেশি যে সকল শিশু সঠিক ভাবে বেড়ে উঠেনি বা দেরিতে বৃদ্ধিপ্রাপ্ত হয়েছে বা যেসকল শিশুর সত্যি সত্যি সেরেব্রাল পালসি হয়নি তাদের ক্ষেত্রে সোডিয়াম ভ্যালপ্রোয়েট সাবধানে ব্যবহার করতে হবে। শিশুর মাইটোকন্ড্রিয়াল ডিসফাংশন এর ইতিহাস থাকলে ভ্যালপ্রোয়েট যকৃত অকার্যকর করে মৃত্যু ঘটাতে পারে। মাইটোকন্ড্রিয়াল ডিসফাংশন শিশুর ঋণাত্মক প্রবৃদ্ধি বা আগের চাইতে খারাপ শারিরীক অবস্থা দিয়ে বোঝা যায়ঃ যেমন যে শিশু আগে বসতে সক্ষম ছিলো সে এখন বসতে পারেনা, এটিই ঋণাত্মক প্রবৃদ্ধি।
- মন্তব্য
 - যে কোন ধরনের খিঁচুনিতে ভ্যালপ্রোয়েট ভালো কাজ করে এবং ফেনোবারবিটোন এর মতো রোগীর উপর তন্দ্রাচ্ছন্নতার মতো কোন প্রতিক্রিয়া তৈরী করেনা।
 - সোডিয়াম ভ্যালপ্রোয়েট, ক্লোবাজাম এবং ল্যামোট্রিজিন এর সাথেও একসাথে কাজ করতে পারে।

4

 চিকিৎসা হিসেবে সোডিয়াম ভ্যালপ্রোয়েট ঔষধ গ্রহন করছে, এমন রোগীর ক্ষেত্রে ল্যামোট্রিজিন খুব সাবধানে ব্যবহার করতে হবে।

ক্লোবাজাম

- ঔষধের মাত্রা
 - শুরুতে ১- ২ মিলিগ্রাম ডোজে দৈনিক একবার প্রয়োগ করতে হবে। এরপর নিম্নবর্নিত উপায়ে ঔষধের মাত্রা বাড়ানো যায়ঃ
 - বয়স ২ বছরের কম হলেঃ ২.৫- ৫ মিলিগ্রাম দৈনিক দুই বেলা,
 - বয়স ২- ১০ বছর হলে ৫- ১০ মিলিগ্রাম দৈনিক দুই বেলা,
 - বয়স ১০ বছরের উর্ধ্বে হলে ১০ মিলিগ্রাম দৈনিক তিনবেলা,
 - মেইন্টেনেন্স এর ক্ষেত্রে ০.৩- ১ মিলিগ্রাম/কেজি/দিন হারে মোট ঔষধ দুই বেলায় ভাগ করে দিতে হবে।
- উপরে বর্নিত ঔষধের মাত্রা হচ্ছে মূল গাইডলাইন, ছোট শিশুদের ক্ষেত্রে এর চাইতে বেশী মাত্রাতেও ঔষধ ব্যবহার করা যায় যদি শিশুর শরীরে কোন পার্শ্বপ্রতিক্রিয়া দেখা না দেয় এবং খিঁচুনি নিয়ন্ত্রনে সহায়ক হয়।
- যে ধরনের খিঁচুনির জন্যে উপকারী
 - সকল ধরনের খিঁচুনির জন্যে প্রযোজ্য,
 - ডিসটোনিয়া ব্যবস্থাপনার ক্ষেত্রেও কিছু কিছু ক্ষেত্রে এই ঔষধ কাজ করে।
- পার্শ্ব প্রতিক্রিয়া
 - তন্দ্রাচ্ছরতা,
 - অতিক্রিয়াশীল বা শিশু অতি চঞ্চল হতে পারে।
 - হ্যালুসিনেশন।
 - মুখ থেকে ক্রমাগত লালা ঝরতে পারে।
- মন্তব্য
 - মনোথেরাপি হিসেবে কার্যকর এবং খিঁচুনির অন্য ঔষধের সাথেও ব্যবহার উপযোগী।
 - স্বল্পমাত্রার ডোজ যেমন ১ মিলিগ্রাম দৈনিক দুইবার হারে ব্যবহার শুরু করে মাত্রা বাড়ানো যায়।
 - স্বল্পমাত্রায় পাওয়া না গেলে ঔষধ গুড়ো করে পানির সাথে মিশিয়ে স্বল্পমাত্রায় খাওয়ানো যায়।
 - অভিভাবক যদি পানিতে গুলিয়ে খাওয়ানোর মাত্রা না বোঝেন, সেক্ষেত্রে ট্যাবলেটের চারভাগের একভাগ বা অর্ধেক করেও রোগীকে খাওয়ানো যায়।
 - কোন কোন ক্ষেত্রে শিশুদের শরীর বেনজোডায়াজেপিন এ অভ্যস্ত হয়ে পরে। যদি বেশ কয়েকমাস খিঁচুনির সম্পূর্ন নিয়ন্ত্রনের পর হঠাত আবার দেখা দেয়, সেক্ষেত্রে অন্য বেনজোডায়াজেপিন যেমন নাইট্রাজেপাম চিকিৎসায় ব্যবহার করতে হয়।



নাইট্রাজেপাম

- ঔষধের মাত্রা
 - ১ মাস থেকে ২ বছর বয়সের মাঝে এই ঔষধ শুরু করতে হবেঃ শরীরের প্রতি কেজি ভরে ০.২৫ মিলিগ্রাম হারে দৈনিক দুইবার থেকে শুরু করে, প্রতি কেজি ভরে ০.৫ মিলিগ্রাম হারে দৈনিক দুইবার দেয়া যায়।
 - উপরে বর্নিত ঔষধের মাত্রা হচ্ছে মূল গাইডলাইন, ছোট শিশুদের ক্ষেত্রে এর চাইতে বেশী মাত্রাতেও ঔষধ ব্যবহার করা যায় যদি শিশুর শরীরে কোন পার্শ্বপ্রতিক্রিয়া দেখা না দেয় এবং খিঁচুনি নিয়ন্ত্রনে উপকারী হয়।
- যে ধরনের খিঁচুনির জন্যে উপকারী
 - সকল ধরনের খিঁচুনির জন্যে প্রযোজ্য,
 - ইনফ্যান্টাইল স্পাজম (Infantile Spasm) এর ক্ষেত্রেও প্রযোজ্য।
- পার্শ্ব প্রতিক্রিয়া
 - তন্দ্রাচ্ছন্নতা,
 - মুখ থেকে ক্রমাগত লালা ঝরতে পারে।
- মন্তব্য
 - মনোথেরাপি হিসেবে কার্যকর এবং খিঁচুনির অন্য ঔষধের সাথেও ব্যবহার উপযোগী।
 - স্বল্পমাত্রার ডোজ থেকে ব্যবহার শুরু করে এর মাত্রা বাড়ানো যায়।
 - যদি রোগী খিঁচুনির চিকিৎসায় তিন মাসের বেশি নাইট্রাজেপাম পেয়ে থাকে এবং ঔষধ বন্ধ করার প্রয়োজন হয়, সেক্ষেত্রে ফেনোবারবিটোন এর মতো ধীরে ধীরে মাত্রা কমিয়ে কমিয়ে বন্ধ করতে হয়।
 - কোন কোন ক্ষেত্রে শিশুদের শরীর বেনজোডায়াজেপিন এ অভ্যস্ত হয়ে পরে। যদি বেশ কয়েকমাস খিঁচুনির পুরোপুরি নিয়ন্ত্রনের পর আবার হঠাত দেখা দেয়, সেক্ষেত্রে অন্য বেনজোডায়াজেপিন যেমন ক্লোবাজাম চিকিৎসায় ব্যবহার করতে হয়।

কার্বামাজেপাইন

- ঔষধের মাত্রা
 - শরীরের প্রতি কেজি ভরে ১০- ২০ মিলিগ্রাম হারে হিসেব করে যতোটুকু ঔষধ হয় তা ২৪ ঘন্টায় ২ বা ৩ ভাগে ভাগ করে ডোজ হিসেবে দিতে হয়; শুরুতে শরীরের প্রতি কেজি ভরে ২.৫- ৫ মিলিগ্রাম প্রতিদিন হারে দেয়া শুরু করে পরবর্তীতে ধীরে ধীরে শরীরের প্রতি কেজি ভরে দৈনিক ১০ মিলিগ্রাম পর্যন্ত বাড়ানো যায় দিনে ২ বা ৩ ভাগে দিতে হয়। এরপর ঔষধের মাত্রা প্রয়োজন অনুযায়ী বাড়ানো যায়। কোন কোন রোগী স্বল্প মাত্রায় যেমন ৫- ১০ মিলিগ্রাম/কেজি/দিন হারেই সুস্থ্য থাকে; সেক্ষেত্রে পরবর্তী প্রয়োজন ছাড়া ঔষধের একই মাত্রা বজায় রাখতে হয়।
- যে ধরনের খিঁচুনির জন্যে উপকারী
 - শরীরের একাংশ বা কোন বিশেষ অংশের খিঁচুনির (Focal Seizure) জন্যে প্রযোজ্য।
 - সারা শরীরের খিঁচুনি বা অ্যাবসেন্স সিজার (Absence Seizure) এবং ফ্রোইল সিজার (Febrile Seizure) এর জন্যে এই ঔষধ উপযোগী নয়।
- পার্শ্ব প্রতিক্রিয়া
 - শিশু অতিক্রিয়াশীল বা অতি চঞ্চল হতে পারে,
 - যকৃত বা লিভারের কার্যক্ষমতা লোপ পেতে পারে বা কমে যেতে পারে।
- মন্তব্য
 - শরীরের একাংশ বা কোন বিশেষ অংশের খিঁচুনির জন্যে উপকারী ঔষধ, যেমন হেমিপ্লেজিক সেরেব্রাল পালসির সাথে থাকা খিঁচুনি।

ডিসটোনিয়া/স্পাসটিসিটি ব্যবস্থাপনার ঔষধ সমূহঃ

ট্রাইহেক্সিফেনিডিল/বেনজহেক্সোল

- ঔষধের মাত্রা
 - ০.২৫ মিলিগ্রাম ২৪ ঘন্টায় একবার থেকে শুরু করে ধীরে ধীরে একই মাত্রার ঔষধ (০.২৫ মিলিগ্রাম) ২৪ ঘন্টায় তিনবার পর্যন্ত ব্যবহার করা যায়।
 - ঔষধ শরীরে সহনীয় হলে ২৪ ঘন্টায় ২- ৪ মিলিগ্রাম হারে তিনবার পর্যন্ত বাড়ানো যেতে পারে।
 - বয়স্ক বাচ্চাদের ক্ষেত্রে উপকার পেলে উচ্চমাত্রায় এই ঔষধ দেয়া যায়।
- যে ধরনের লক্ষনের ক্ষেত্রে উপকারী
 - মূলত ডিসটোনিয়া বা যেসকল সেরেব্রাল পালসিতে নিস্তেজ পেশী দেখা যায় তার ব্যবস্থাপনায় এই ঔষধ ব্যবহার হয়। এতে করে অনেক ক্ষেত্রে বাচ্চার লালা ঝরার প্রবনতাও কমে যায়। কারন এই ঔষধের অ্যান্টিকোলেনার্জিক বৈশিষ্ট্য রয়েছে।
- পার্শ্ব প্রতিক্রিয়া
 - মুখ ও চোখ শুক্ষ হতে পারে,
 - কোষ্ঠকাঠিন্য,
 - দ্বিধান্বিত চিন্তাভাবনা,
 - একই জিনিষ দুটো করে দেখা,
 - রোগীর খিটখিটে মেজাজ দেখা যেতে পারে।
- মন্তব্য
 - পার্শ্বপ্রতিক্রিয়া কোন সমস্যা তৈরী না করলে ডিসটোনিয়া ব্যবস্থাপনার ক্ষেত্রে এই ঔষধের দক্ষতা প্রায় ২৫- ৩০%। কোন কোন শিশুর ক্ষেত্রে এই ঔষধের কর্মক্ষমতা এর চাইতেও বেশি দেখা যায়।
 - এই ঔষধের সাথে অন্য কোন অ্যান্টিকোলেনার্জিক (Anticholenergic) ঔষধ ব্যবহার করা যাবেনা এবং যদি ব্যবহারের পর কোন পার্শ্বপ্রতিক্রিয়া ক্লিনিকাল সমস্যা হিসেবে দেখা দেয় তাহলেও এই ঔষধ ব্যবহার বাদ দিতে হবে।
 - সর্বোচ্চ মাত্রায় ৩- ৪ সপ্তাহ ব্যবহারের পরেও কোন উপকার পাওয়া না গেলে ঔষধ বন্ধ করতে হবে, কারন এই ঔষধ দেরীতে কাজ করেনা।



বেক্লোফেন

- ঔষধের মাত্রা
 - ২- ৭ বছরঃ ১০- ৪০ মিলিগ্রাম প্রতিদিন তিন থেকে চার ভাগে ভাগ করে দেয়া হয়। প্রথমে ২.৫-৫ মিলিগ্রাম দিনে দুইবার থেকে শুরু করে প্রতি তিন/চার দিনে ৫- ১৫ মিলিগ্রাম পর্যন্ত দেয়া যায়। সর্বোচ্চ ৪০ মিলিগ্রাম প্রতিদিন মাপে ঔষধ দেয়া যায়।
 - ৮- ১১ বছরঃ ১০- ৬০ মিলিগ্রাম প্রতিদিন তিন থেকে চার ভাগে ভাগ করে দেয়া হয়। প্রথমে ২.৫- ৫ মিলিগ্রাম দিনে দুইবার থেকে শুরু করে প্রতি তিন/চার দিনে ৫- ১৫ মিলিগ্রাম পর্যন্ত দেয়া যায়। সর্বোচ্চ ৬০ মিলিগ্রাম প্রতিদিন মাপে ঔষধ দেয়া যায়।
 - ১২ বছর বা তদুর্ধ্বঃ ২০- ৮০ মিলিগ্রাম প্রতিদিন তিন থেকে চার ভাগে ভাগ করে মুথে থেতে দেয়া যায়। প্রথমে ২.৫- ৫ মিলিগ্রাম দিনে দুইবার থেকে শুরু করে প্রতি তিন/চার দিনে ১৫ মিলিগ্রাম পর্যন্ত দেয়া যায়। সর্বোচ্চ ৮০ মিলিগ্রাম প্রতিদিন মাপে ঔষধ দেয়া যায়।
- যে ধরনের লক্ষনের ক্ষেত্রে উপকারী
 - মূলত পেশীর অনৈচ্ছিক শক্ত হওয়া বা স্পাসটিসিটির চিকিৎসায় ব্যবহৃত হয়।
- পার্শ্ব প্রতিক্রিয়া
 - রোগীর হাইপোটোনিয়া বা নির্জীব মাংসপেশী তৈরী করতে পারে।
 - রোগীর অনবরত লালা ঝরতে পারে।
 - তন্দ্রাচ্ছন্নতা।
- মন্তব্য
 - স্পাসটিসিটির কারনে শক্ত মাংসপেশির চিকিৎসায় বেকলোফেন একটি ভালো ঔষধ।
 - মাংসপেশীর নির্জীবতা বা ডিসটোনিয়া বেশি থাকলে বা মিশ্র ধরনের ডিসটোনিয়া থাকলে এই ঔষধ ব্যবহার না করাই উত্তম।
 - কিছু কিছু রোগীর ক্ষেত্রে উচ্চ মাত্রায় বেকলোফেন দেয়া হয়, সেক্ষেত্রে খিটখিটে মেজাজ এর মতো পার্শ্ব প্রতিক্রিয়া হতে পারে। তখন ঔষধের মাত্রা ঠিক করে দিতে হয়।
 - বেনজোডায়াজেপিন বা ফেনোবারবিটোন এর মতো ঔষধ বেকলোফেন এর সাথে দেয়া হলে রোগীর তন্দ্রাচ্ছন্নতা ও লালা ঝরার মাত্রা বৃদ্ধি পেতে পারে।
 - রোগী তিন মাসের বেশি বেকলোফেন ব্যবহার করে থাকলে কয়েক সপ্তাহ জুড়ে ধীরে ধীরে ঔষধের মাত্রা কমিয়ে এক সময় তা বন্ধ করতে হবে।

লেভাডোপা/কার্বিডোপা

- ঔষধের মাত্রা
 - শরীরের প্রতি কেজি ভরে ১- ৪ মিলিগ্রাম হারে যে পরিমান ঔষধ হয় তা দুই ভাগে ভাগ করে প্রতিদিন দিতে হয়। শুরুতে প্রতি কেজি ভরে ১ মিলিগ্রাম হারে মোট ঔষধের পরিমান কে দুই ভাগে ভাগ করে ২৪ ঘন্টায় দিতে হবে। ঔষধের মাত্রার পরিমান ধীরে ধীরে বাড়িয়ে শরীরের প্রতি কেজি ভরে ৪ মিলিগ্রাম হারে মোট ঔষধকে ২৪ ঘন্টায় তিন ভাগে ভাগ করে তিন বেলা দিতে হবে।
 - খিঁচুনি স্থানিক বা শরীরে আংশিক স্থানে হলে এবং ঔষধ প্রয়োগে উপকার পেলে ঔষধের মাত্রা বাড়ানো যায়।
- যে ধরনের লক্ষনের ক্ষেত্রে উপকারী
 - ডিসটোনিয়া ব্যবস্থাপনার ক্ষেত্রে।
 - ডিসটোনিয়া বা মাংশপেশীর নির্জীবতা যদি স্বাভাবিকভাবেই দিনদিন খারাপের দিকে যায় বা কোন ব্যায়ামের কারনে অবস্থা খারাপ হতে থাকে, সেক্ষেত্রে এই ঔষধ বেশ কাজে দেয়।
- বিরূপ প্রতিক্রিয়া
 - বমি বমি ভাব।
- মন্তব্য
 - সেরেব্রাল পালসির ডিসটোনিয়া ব্যবস্থাপনায় এর সাফল্য শতকরা মাত্র ২৫ ভাগ কিন্তু এই ঔষধ ব্যবহার নিরাপদ।
 - জেনেটিক ডোপামিন রেসপনসিভ ডিসটোনিয়া অনেক সময় সেরেব্রাল পালসি রোগের মতো লক্ষন প্রকাশ করে। এক্ষেত্রে এই ঔষধ বেশ কার্যকর। যদিও এরকম রোগী বেশ কম।
 - এই ঔষধের কিছু কিছু প্রিপারেশোন লেভাডোপা/বেনসেরাজাইড হিসেবে পাওয়া যায়। এক্ষেত্রেও ঔষধের পরিমাত্রা লেভোডোপার মতোই হবে।

শিশুর অস্বাভাবিক আচরন ব্যবস্থাপনার ঔষধ সমূহঃ

ক্লোনিডিন

- ঔষধের মাত্রা
 - রাতের ঘুম ঠিক রাখার জন্যে ২৫ মাইক্রোগ্রাম করে রাতে শোবার সময় খেতে দিতে হবে।
 - ২৪ ঘন্টায় তিনবার ২৫ থেকে ১০০ মাইক্রোগ্রাম পর্যন্ত ঔষধের পরিমান বাড়িয়ে আচরন ব্যবস্থাপনার ক্ষেত্রে ব্যবহার করা হয়।
- যে ধরনের উপসর্গ দেখা গেলে শুরু করতে হবে
 - অতিক্রিয়াশীল, চঞ্চল বা অস্থির শিশু বা আক্রমনাত্মক আচরন, যেমনঃ কাউকে কামড়ানো, বিদ্যালয়ে পড়ায় অমনোযোগী হওয়া, বিদ্যালয়ে অন্য শিশুকে বিরক্ত করা, এক জায়গায় স্থির হয়ে বসে না থাকতে পারা (এই লক্ষনগুলোই মূলত বাড়িতে বা বিদ্যালয়ে চঞ্চলতা, অস্থিরতা বা আক্রমনাত্মক আচরন হিসেবে ধরা হয় এবং চিকিৎসার ব্যবস্থা করতে হয়।)
 - মারাত্মক ডিসটোনিয়াতে আক্রান্ত যেসকল রোগী কিছুদিন পরপর অস্বাভাবিক আচরন করে এবং দিন দিন আচরন বেশী খারাপের দিকে যায় তাদের ক্ষেত্রেও এই ঔষধ বেশ কার্যকর। তিন বা চারদিনের জন্যে ২৪ ঘন্টায় ৪ থেকে ৬ বার ১০০ মাইক্রোগ্রাম মাত্রার ঔষধ শর্ট বার্ষ্ট (Short Burst) এর ক্ষেত্রে উপকারী। এরপর আবার ধীরে ধীরে মাত্রা কমিয়ে স্বাভাবিক মাত্রায় নামিয়ে নিয়ে আসতে হয়।
- পার্শ্ব প্রতিক্রিয়া
 - তন্দ্রাচ্ছন্নতা
 - শারীরিক ভঙ্গিগত নিম রক্তচাপ এর কারনে কখনো কখনো তন্দ্রাচ্ছন্নতা দেখা দিতে পারে-বেশি দেখা দেয় সাধারনত ঔষধের নিম্নমাত্রার ক্ষেত্রে।
- মন্তব্য
 - তিন মাসের বেশী ক্লোনিডিন গ্রহন করলে সপ্তাহখানেক সময় নিয়ে ঔষধের মাত্রা কমিয়ে কমিয়ে ঔষধ বন্ধ করতে হয়।
 - মনোযোগে অক্ষমতার রোগীর চিকিৎসার ক্ষেত্রে ক্লোনিডিন এর সাফল্য গড়পড়তা।

রিসপেরিডন

- ঔষধের মাত্রা
 - শিশুদের ক্ষেত্রে দৈনিক ০.২৫ মিলিগ্রাম থেকে শুরু করে সর্বোচ্চ ৫ মিলিগ্রাম পর্যন্ত। সর্বনিম্ন কত মাত্রার ওউষধে কাজ করে তা ভালো করে দেখতে হবে।
- যে ধরনের উপসর্গে ব্যবহার উপযোগী
 - অতি চঞ্চল বা অতি ক্রিয়াশীল বা আক্রমনাত্মক আচরনের রোগীদের ব্যবস্থাপনায় এই ঔষধ ব্যবহার হয়।
- পার্শ্ব প্রতিক্রিয়া
 - ক্ষুধা বৃদ্ধি,
 - শরীরের ভর বৃদ্ধি,
 - বিপাক ক্রিয়ায় গোলমাল- এই ঔষধ লম্বা সময় ধরে ব্যবহার করলে হাইপারলিপিডিমিয়া দেখা দিতে পারে,
 - শরীর শক্ত হয়ে যেতে পারে।
- মন্তব্য
 - ক্লোনিডিন দিয়ে চিকিৎসা শুরু করতে হবে।
 - যেসকল বাচ্চাদের আচরনের কারনে পারিবারিক দৈনন্দিন জীবন ক্ষতিগ্রস্থ, তাদের চিকিৎসায় এই ঔষধ খুবই কার্যকর।

জরুরী অবস্থায় রোগী ব্যাবস্থাপনার ঔষধ সমূহঃ

ডায়াজিপাম

- ঔষধের মাত্রা
 - ৬ বছর বয়সের কম হলে শরীরের প্রতি কেজি ভরে ০.৫ মিলিগ্রাম করে; ৬ থেকে ১১ বছর বয়সের হলে শরীরের প্রতি কেজি ভরে ০.৩ মিলিগ্রাম করে; ১১ বছর বয়সের বেশী হলে শরীরের প্রতি কেজি ভরে ০.২ মিলিগ্রাম হিসেবে দিতে হবে।
- যে ধরনের উপসর্গে ব্যবহার উপযোগী
 - প্রলম্বিত খিঁচুনির ক্ষেত্রে(বাড়িতে ৫ মিনিটের অধিক সময় খিঁচুনি থাকলে)।
 - পার্শ্ব প্রতিক্রিয়া
 - ঘুম,
 - শ্বাস গ্রহনে অক্ষমতা এবং মারাত্মক ক্ষেত্রে ফুসফুস এর কাজ বন্ধ হয়ে যাওয়া,
 - লোকাল ইনজুরি।
- মন্তব্য
 - খিঁচুনি আক্রান্ত রোগীর পরিবার ডায়াজিপাম ব্যবহারের প্রক্রিয়া সম্পর্কে পুরোপুরি জানলে বাড়িতে ব্যবহারের জন্যে ব্যবস্থাপত্রে ডায়াজিপাম দেয়া হয়।
 - খিঁচুনির সময় যদি খুব হাত পা নড়ে বা শিশু যদি অজ্ঞান হয়ে যায়, সেক্ষেত্রে প্রাথমিক চিকিৎসার একদম প্রথম ধাপগুলো আগে অনুসরন করতে হয়, যেমনঃ বাইরের যেকোন আঘাত থেকে শিশুকে বাঁচাতে হবে, শ্বাস নেয়ার সুবিধার্থে পাশ করে শুইয়ে দিতে হবে।
 - যা যা প্রয়োজনঃ
 - নরমাল স্যালাইনে মেশানো ডায়াজিপাম ২৫ মিলিলিটার এর বোতলে থাকতে হবে যার এক মিলিলিটার তরলে ১ মিলিগ্রাম ডায়াজিপাম মিশ্রিত থাকবে (ঘনত্বের মাত্রা ভিন্ন ও হতে পারে),
 - পূনর্ব্যাবহার যোগ্য ১০ মিলিলিটার এর সিরিঞ্জ,
 - সিরিঞ্জের সাথে একটি পূনর্ব্যাবহারযোগ্য নরম প্লাস্টিক টিউব লাগানো থাকতে হবে যেনো বোতল থেকে ডায়াজিপাম তুলে তা ইনজেক্ট করা যায়,
 - এক স্যাচেট লুব্রিকেন্ট জেলি।

<u>তথ্যসূত্রঃ</u>

- Australian Medicines Handbook
- MIMS Australia
- Epocrates
- www.rch.org.au
- www.dhs.winsconsin.gov

